The synergistic effects of neuromuscular electrical stimulation combined with blood flow restriction on muscular adaptations and pain modulation

Paul Head

PhD

A Doctoral thesis submitted in partial fulfilment of the requirements of St Mary's University for the degree of Doctor of Philosophy

Faculty of Sport, Health and Applied Sciences

St Mary's University, London

June 2024

© Paul Head 2024

Contents

Abstr	Abstract12		
Decla	Declaration of originality14		
Ackno	owledgments	15	
Public	cations (Thesis)	17	
Public	Publications		
Confe	erence and professional presentations	18	
List o	f abbreviations	20	
Chapt	ter 1	22	
Introd	luction	22	
1.1	Introduction	23	
1.2	The complex interplay of pain and disuse on skeletal muscle function	23	
1.3	NMES	26	
1.4	BFR	27	
Chapt	ter 2	29	
Revie	w of the literature; NMES combined with BFR	29	
2.1	Introduction	30	
2.2	Proposed mechanisms of action	31	
2.3	Muscle strength	37	
2.4	Muscle size	41	
2.5	NMES parameters	44	
2.	.5.1 Frequency	44	
2.	.5.2 Pulse width/duration	45	
2.	.5.3 Amplitude/ Intensity	47	
2	.5.4 Electrode placement and size	49	
2.6	BFR parameters	50	
2	.6.1 BFR frequency	53	
2.7	Effect of NMES and BFR on pain	53	
2.8	2.8 Is NMES combined with BFR tolerable?		
2.9	Conclusion	57	
2.10 Thesis aims and research questions59			
2.11	Aims	59	
2.12	Null Hypotheses	59	
Chapt	Chapter 361		
General methods61		61	
3.1	3.1 Health and safety62		
3.2	3.2 Participant recruitment62		

	3.3 Ankle-brachial index	64
	3.3.1 Measurement reliability	65
	3.4 Neuromuscular electrical stimulation	66
	3.5 Determination of blood flow restriction pressure	67
	3.6 Maximal voluntary isometric contraction	69
	3.7 Isokinetic Dynamometry	70
	3.8 Quadriceps muscle thickness	71
	3.9 Quadriceps cross-sectional area	72
	3.10 Quadriceps muscle architecture	74
	3.11 Blood pressure	75
	3.12 Heart rate	76
	3.13 Wellness	77
	3.14 Rating of perceived exertion	77
	3.15 Pain	78
	3.16 End-test torque and impulse above end-test torque	78
	3.17 Voluntary activation	80
	3.18 Near-infrared spectroscopy	81
	3.19 Pressure pain thresholds	83
	3.20 Cold pain thresholds	84
	3.21 Temporal summation of pain	85
	3.22 5 Sit-to-stands	85
	3.23 Knee bend in 30 seconds	86
	3.24 Knee Injury and Osteoarthritis Outcome Score questionnaire	87
С	hapter 4	88
С	ptimisation of NMES with BFR: The effect of restriction pressure	88
J	ournal Sport Rehabilitation. 2020 July 31;30(3): 375-383	88
	4.1 Preface	89
	4.2 Abstract	89
	4.3 Introduction	90
	4.4 Method	91
	4.4.1 Participants	91
	4.4.2 Study design	92
	4.5 Procedures	93
	4.5.1 ABI	93
	4.5.2 NMES	93
	4.5.3 Determination of BFR pressure	94
	4.5.4 MVIC	94
	4.5.5 MTH	94
	4.5.6 Blood pressure	94

4.5.7 HR	94
4.5.8 RPE	94
4.5.9 Pain	94
4.6 Statistical Analysis	94
4.7 Results	95
4.7.1 MVIC	95
4.7.2 MTH	
4.7.3 Blood pressure	
4.7.4 HR	100
4.7.5 RPE	100
4.7.6 Pain	100
4.8 Discussion	
4.9 Conclusion	
Chapter 5	108
NMES combined with 40% and 80% BFR pressures, incre	eases muscular strength
and hypertrophy without movement: a 6-week Randomis	ed Control Trial108
5.1 Abstract	109
5.2 Introduction	110
5.3 Method	113
5.3.1 Ethical approval	113
5.3.2 Participants	114
5.3.3 Study design	116
5.4 Procedures	117
5.4.1 NMES	117
5.4.2 Determination of BFR pressure	118
5.5.1 Isokinetic Dynamometry	118
5.5.2 Quadriceps CSA	118
5.5.3 Quadriceps muscle architecture	118
5.5.4 RPE	118
5.5.5 Pain	119
5.7 Statistical Analysis	119
5.8 Results	120
5.8.1 MVIC at 30° knee angle	120
5.8.2 MVIC at 45 ^o knee angle	121
5.8.3 MVIC at 60° knee angle	122
5.8.4 MVIC at 75 ^o knee angle	122
5.8.5 MVIC at 90º knee angle	122
5.8.7 Concentric strength	123
5.8.8 Eccentric strength	

5.9.1 VL CSA at 50% length of femur	125
5.9.2 VL CSA at 75% length of femur	125
5.9.3 Fascicle length	127
5.9.4 Pennation angle	127
5.10.1 Maximum tolerable NMES current	128
5.10.2 Pain	129
5.11.1 Impulse above end-test torque	130
5.11.3 VA%	130
5.11.4 Slope analysis on Δ HHb values	131
5.12 Discussion	131
5.13 Conclusion	138
Chapter 6	139
Effect of NMES combined with BFR on pressure pain, cold pain and temporal	
summation of pain in healthy adults.	139
6.1 Abstract	140
6.2 Introduction	140
6.3 Method	143
6.3.1 Participants	143
6.3.2 Study design	144
6.4 Procedures	144
6.4.1 ABI	144
6.4.2 NMES	145
6.4.3 Determination of BFR pressure	145
6.4.4 PPT	145
6.4.5 CPT	145
6.4.6 TSP	145
6.4.7 RPE	145
6.4.7 Pain	145
6.5 Statistical Analysis	145
6.6 Results	146
6.6.1 PPT	146
6.6.2 CPT	147
6.6.3 TSP	147
6.6.4 Pain	147
6.6.5 RPE	148
6.7 Discussion	148
6.8 Conclusion	153
Chapter 7	154

Effect of NMES combined with BFR on pressure pain, cold pain, temporal summation of pain and function in knee OA patients	154
7.1. Abstract	155
7.2 Introduction	156
7.3 Method	159
7.3.1 Participants	159
7.3.2 Study design	160
7.4 Procedures	162
7.4.1 ABI	162
7.4.2 NMES	162
7.4.3 Determination of BFR pressure	162
7.4.4 PPT	162
7.4.5 CPT	162
7.4.6 TSP	162
7.4.7 5-repetition sit-to-stand	162
7.4.8 Knee bends in 30 seconds	163
7.4.9 KOOS questionnaire	163
7.5 Statistical Analysis	163
7.6 Results	163
7.6.1 PPT	164
7.6.2 CPT	164
7.6.3 TSP	165
7.6.4 5-repetition sit-to-stand	165
7.6.5 Knee bends in 30 seconds	166
7.6.6 KOOS	166
7.6.7 SBP	166
7.6.8 DBP	167
7.6.9 Pain	167
7.6.10 RPE	167
7.7 Discussion	167
7.8 Conclusion	171
Chapter 8	173
General discussion	173
8.1 Main findings	174
8.2 Muscle size	174
8.3 Muscle strength	176
8.4 Cardiovascular safety	177
8.5 Pain modulation	178
8.6 Clinical practice guidelines	179

8.7 Limitations	180
8.8 Future Research Considerations	180
8.9 Conclusion	181
References	182
Appendix 1	206
Appendix 2	207
Appendix 3	208
Appendix 4	209
Appendix 5	210
Appendix 5.1 Borg RPE scale	211
Appendix 5.2 Numeric pain rating scale	211
Appendix 5.3 Physical activity and readiness questionnaire	212
Appendix 5.4 Wellness questionnaire	213
Appendix 5.5 Knee injury and osteoarthritis outcome score questionnaire	214

List of figures

Chapter 2	Review of the literature; NMES combined with BFR	Page
Fig 2.1.	Diagram of NMES+BFR molecular pathways.	
Chapter 3	General methods	
Fig 3.1.	Example of vastus lateralis cross-sectional area measurement.	74
Fig 3.2.	Example of vastus lateralis muscle thickness, fascicle length and pennation angle.	75
Fig 3.3.	Example of impulse above end-test torque and end-test torque.	79
Fig 3.4.	Example of voluntary activation trace.	81
Fig 3.5.	Example of deoxyhemoglobin signal during each contraction of the 5 minute all-out test.	83
Chapter 4	Acute NMES with BFR: The effect of restriction pressure	
Fig 4.1.	Experimental protocol. All participants performed the same neuromuscular electrical stimulation (NMES) protocol under four different blood flow restriction (BFR) pressures (0, 40, 60 and 80%) Outcome measures; systolic blood pressure (SBP); diastolic blood pressure (DBP); heart rate (HR); vastus medialis (VM) and vastus lateralis (VL) muscle thickness (MTH); knee extension maximal voluntary isometric contraction (MVIC) was assessed before (pre) and after (post) each experimental condition. Outcome measures	93

assessed after every 10 NMES repetitions included; rating of perceived exertion (RPE), pain and HR. See abbreviations throughout.

- Fig 4.2. Knee extension MVIC pre-test to post-test change Δ ; values as mean \pm SEM. Significant difference was set at p < 0.05; * = significant difference between pre-test and post-test; † = significantly greater change compared to all other experimental conditions.
- Fig 4.3. Evoked NMES force (% MVIC) during every contraction; values as mean \pm SEM; * = significant decline from first repetition to the last: † = significantly different to all other conditions p < 0.05.

Chapter 5 NMES combined with varying BFR pressures, increases muscular strength and hypertrophy vs. NMES alone: a Randomised Control Trial.

- Fig 5.1. Consort flow diagram of participants through the study.
- Fig 5.2. Experimental protocol. All participants performed the same NMES 117 protocol under three different BFR pressures (0, 40 and 80%) for 6-weeks, 3 sessions a week. Outcome measures; VL CSA, isometric peak torque and concentric and eccentric peak torque (ISOK), 5 minute all-out test, NIRS, VA was assessed before and post the 6-week intervention period. Outcome measures assessed after every 5 minutes set included; RPE, pain and max tolerable current achieved.
- Fig 5.3. Isometric peak torque at a knee angle of 30° ; values as mean ± 121 SEM; [#] = significant increase compared to NMES alone.
- Fig 5.4.Isometric peak torque at a knee angle of 45° ; values as mean ±122SEM; # = significant increase compared to NMES alone.
- Fig 5.5. Isometric peak torque at a knee angle of 90° ; values as mean ± 123 SEM; [#] = significant increase compared to NMES alone.
- Fig 5.6. Quadriceps eccentric peak torque; values as mean ± SEM; [#] = 124 significant increase compared to NMES alone.
- Fig 5.7. VL CSA at 50% length of femur; values as mean ± SEM; [#] = 125 significant increase compared to NMES alone.
- Fig 5.8. VL CSA at 75% length of femur; values as mean ± SEM; # = 126 significant increase compared to NMES alone.
- Fig 5.9.Example of VL CSA at 50% femur length; pre and post127
- Fig 5.10. Maximal tolerable NMES current after every intervention session. 128 Values as mean ± SEM; [#] = significantly greater group effect than NMES alone.
- Fig 5.11. NIRS trace example zoomed into three, 3 second contractions. 131 TSI%: tissue oxygen/saturation index, O2Hb: oxyhemoglobin, HHB: deoxyhemoglobin, tHB: total hemoglobin

9

96

97

Chapter 6 Effect of NMES combined with BFR on pressure pain, cold pain and TSP in healthy adults.

Fig 6.1. Experimental protocol. All participants performed the same NMES 144 protocol under three different BFR pressures (0%, 40% and 80%). Outcome measures; PPT, CPT and TSP were assessed before (pre) and immediately post, 20 minutes and 45 minutes post each experimental condition. Outcome measures assessed after every 5 minutes included; RPE and pain.

Chapter 7 Effect of NMES combined with BFR on pressure pain, cold pain, TSP and function in knee OA patients.

Fig 7.1. Experimental protocol. All participants performed the same NMES 161 protocol under three different BFR pressures (0%, 40% and 80%) Outcome measures; KOOS, SBP, DBP, HR, PPT, CPT, TSP, 5-sit-to-stand, knee bends in 30 seconds were assessed before (pre) and immediately post, 20 minutes and 45 minutes post each experimental condition. Outcome measures assessed after every 5 minutes included; RPE and pain.

List of tables

Chapter 2	Review of the literature; NMES combined with BFR	Page
Table 2.1.	NMES parameters for NMES+BFR training studies	48
Chapter 4	Acute NMES with BFR: The effect of restrictive pressure	
Table 4.1.	NMES protocol.	93
Table 4.2.	Knee extension MVIC, muscle thickness and cardiovascular pre-test and post-test measurement values; mean (SD) [95% CI].	99
Table 4.3.	Measurement values after every set (10 contractions) of the interventions; mean (SD).	102
Chapter 5	NMES combined with varying BFR pressures, increases muscular strength and hypertrophy vs. NMES alone: a Randomised Control Trial	
Table 5.1.	Baseline characteristics of participants; mean (SD).	116
Table 5.2.	Knee extension peak torque, post-test measurement values; mean SD.	120
Table 5.3.	VL CSA at 50% and 75% length of femur, post-test measurement values; mean SD.	124

Table 5.4.	Training session pain and RPE; mean (SD).	129
Table 5.5.	5 minute all-out test: impulse above end-test torque; end-test torque; VA at the start and end of the test; Δ [HHb] signal during middle 2 s of each contraction, post-test measurement values; mean SD.	130
Chapter 6	Effect of NMES combined with BFR on pressure pain, cold pain and TSP in healthy adults	
Table 6.1.	PPT (kgf/cm ²), pre-test, post-test, 20 minutes post-test and 45 minutes post-test measurement values, mean (SD).	146
Table 6.2.	CPT, pre-test, post-test, 20 minutes post-test and 45 minutes post-test measurement values, mean (SD).	147
Table 6.3.	TSP, pre-test, post-test, 20 minutes post-test and 45 minutes post-test measurement values, mean (SD).	147
Table 6.4.	Pain and RPE values after set 1, 2, 3 and 4, mean (SD).	148
Chapter 7	Effect of NMES combined with BFR on pressure pain, cold pain and TSP in knee OA patients.	
Table 7.1.	PPT (kgf/cm ²), pre-test, post-test, 20 minutes post-test and 45 minutes post-test measurement values, mean (SD).	164
Table 7.2.	CPT, pre-test, post-test, 20 minutes post-test and 45 minutes post-test measurement values, mean (SD).	164
Table 7.3.	TSP, pre-test, post-test, 20 minutes post-test and 45 minutes post-test measurement values on the OA limb, mean (SD).	165
Table 7.4.	5 sit-to-stands time (s) and pain (0-10), pre-test, post-test, 20 minutes post-test and 45 minutes post-test measurement values, mean (SD).	166
Table 7.5.	Knee bends in 30 seconds on OA limb; amount and pain (0- 10), pre-test, post-test, 20 minutes post-test and 45 minutes post-test measurement values, mean (SD).	166
Table 7.6.	Pain and RPE values after set 1, 2, 3 and 4, mean (SD).	167

Abstract

This thesis conducts a thorough examination of the synergistic impact of neuromuscular electrical stimulation (NMES) combined with blood flow restriction (BFR) on muscular adaptations and pain modulation in healthy and knee osteoarthritis (OA) populations. The primary objective across subsequent chapters is to establish the superior efficacy of NMES+BFR over NMES alone, aiming to optimise muscular adaptations and modulate pain, potentially contributing to improved clinical outcomes and enhancing muscular adaptations without exercise. Chapter 2 commences with a scoping literature review, identifying gaps and inconsistencies in existing research on NMES+BFR and drawing upon the wider literature bases of both to develop an evidence based methodology. Notably, methodological disparities in BFR and NMES application, namely not using arterial occlusion pressure (AOP) to prescribe the BFR stimulus and not using NMES frequencies and intensities found to enhance muscular strength, with these emerging as key contributors to the varying evidence base regarding their combined effectiveness in enhancing muscle strength and size outcomes. Chapter 3 provides a comprehensive overview of the research methods employed, establishing a consistent framework for subsequent studies. Chapter 4 optimises NMES+BFR methodologies, by using NMES recommended parameters and combining it with BFR using AOP (40-80%) to standardise the restrictive pressure. This study assessed acute measures of fatigue (surrogate marker for chronic training adaptations), muscle swelling, RPE, pain perception, and cardiovascular safety. Findings revealed increased fatigue after NMES+BFR (80%) compared to NMES alone. However, acute fatigue was observed after all NMES+BFR conditions, but greater perceptual pain and RPE reported after 60% vs. 40% AOP, therefore, eliminating the NMES combined with 60% condition from future investigation. Importantly, this chapter refines intervention parameters for a subsequent training study. Chapter 5 focuses on a chronic training study to assess the effectiveness of NMES+BFR (40% and 80%) in increasing

muscle strength and size compared to NMES alone. A 6-week, 3-sessions-per-week randomised controlled trial was undertaken. Findings showed greater improvements in muscle strength (isometric and eccentric) and muscle size were observed in NMES+BFR groups, accompanied by greater NMES stimulation intensities tolerated during the training sessions compared to NMES alone. Due to the greater NMES stimulation intensity tolerated in Chapter 5 and the wider BFR evidence base reporting acute reductions in pressure pain and thermal pain thresholds, Chapter 6 investigated the acute effects of pressure, thermal and temporal summation of pain (TSP) thresholds in healthy adults, revealing an acute increase in pressure pain thresholds immediately after NMES+BFR, explaining the greater tolerated currents observed in Chapter 5. **Chapter 7** replicates the methodology in a clinical population (knee osteoarthritis patients), due to this population demonstrating altered exercise induced hypoalgesia responses to healthy adults, which has been proposed as a main mechanism for reduced pain after BFR exercise. The results demonstrated acute increases in pressure pain thresholds, improvements in sit-to-stand performance, and reduced TSP after NMES+BFR (80%), with no effects observed after NMES alone. Cardiovascular safety is confirmed. **Chapter 8** synthesises and discusses the findings, emphasising the potential of NMES+BFR in enhancing strength and hypertrophy and modulating pain in healthy and knee OA patients without requiring exercise. In summary, this thesis offers a comprehensive exploration of NMES+BFR, showcasing its potential to enhance muscular adaptations and pain modulation in both healthy and clinical populations. The research underscores the promise of this intervention in improving clinical outcomes and providing a pre exercise intervention when exercise is not possible due to pain or contraindicated to enhance muscular adaptations and modulate pain.

Declaration of originality

Students are reminded that the work that they submit for assessment must be their own. Please read the following statements and sign and date at the bottom of this form to show that you have complied:

1. This thesis and the work to which it refers are the results of your own efforts. Any ideas, data or text resulting from the work of others (whether published or unpublished) are fully identified as such within the work and attributed to the originator in the text, bibliography or footnotes.

2. This thesis has not been submitted in whole or in part for any other academic degree or professional qualification at this or any other institution.

3. Any chapters that describe the outcomes of joint research should be clearly identified as such with a statement inserted as a footnote on the first page and contributors named. Significant data, images or text resulting from the input of other researchers should be identified as such and attributed to the persons concerned by means of a footnote within the chapter.

4. It is usual to acknowledge the help and guidance of others who have assisted you during your research and preparation of your thesis. Such acknowledgements do not replace or obviate the need for individual attribution as discussed in points 1 and 3.

5. The University reserves the right to submit electronic versions of your draft documents for assessment of plagiarism using electronic detection software such as 'turnitin'. In addition, whether or not drafts have been so assessed, the University reserves the right to require an electronic version of the final document (as submitted) for assessment.

PRINT NAME: PAUL HEAD

SIGNED:

Etter

DATE: 7th September 2023

Acknowledgments

I would like to express my deepest gratitude to my supervisors, Professor Stephen Patterson, Dr Mark Waldron, Dr Nicola Theis and Dr Luke Hughes, for their invaluable guidance, unwavering support and constant encouragement throughout my PhD journey, especially when I was ready to throw in the towel a few years in! Their expertise, patience and commitment to my research have been instrumental in shaping the direction and quality of this thesis. I am truly fortunate to have had the opportunity to work under their amazing supervision and learn from their vast knowledge in the field. Additionally, I would like to acknowledge the support and advice from my previous lecturers, Karen May and Dr. Massimo Barcellona, both of whom encouraged me to further my career and undertake scientific research. Without your support, I would never have embarked on this PhD journey.

I would also like to acknowledge the indispensable contributions of Jack Lineham, the laboratory technician. His expertise, problem solving, and tireless efforts in maintaining and fixing the equipment and facilities were critical to the successful execution of the experiments. I am truly grateful for his technical assistance and support throughout my research.

Furthermore, I would like to extend my thanks and appreciation to all the volunteers who generously dedicated their time and assistance in participating this research and the students that assisted with the running of the studies. Without them this project would not have been possible. Their involvement and enthusiasm have significantly enhanced this thesis.

Finally, I am deeply indebted to my family for their support and assisting with the funds to undertake these studies and to my Uncle John who sadly passed away a few weeks before the completion date, who was extremely supportive throughout this journey and who will be greatly missed. Finally, to my partner, Rachel, for her unwavering love, understanding and endless support. Her presence in my life has made this challenging academic journey possible and meaningful. Thank you for always believing in me, encouraging me and for listening to me go on and on about this subject and the whole journey for the last 7 years!

Publications (Thesis)

Head P, Waldron M, Theis N, Patterson SD. Acute Neuromuscular Electrical Stimulation (NMES) With Blood Flow Restriction: The Effect of Restriction Pressures. *J Sport Rehab*. 2020 Jul 31;30(3):375-383.

Publications

Patterson SD, Hughes L, **Head P**, Warmington S, Brandner C. Blood flow restriction training: a novel approach to augment clinical rehabilitation: how to do it. *Br J Sports Med*. 2017 Dec;51(23):1648-1649.

Lorenz DS, Bailey L, Wilk KE, Mangine RE, **Head P**, Grindstaff TL, Morrison S. Blood Flow Restriction Training. *J Athl Train.* 2021 Sep 1;56(9):937-944.

Paton BM, Court N, Giakoumis M, **Head P**, Kayani B, Kelly S, Kerkhoffs GMMJ, Moore J, Moriarty P, Murphy S, Plastow R, Pollock N, Read P, Stirling B, Tulloch L, van Dyk N, Wilson MG, Wood D, Haddad F. London International Consensus and Delphi study on hamstring injuries part 1: classification. *Br J Sports Med*. 2023 Mar;57(5):254-265.

Plastow R, Kerkhoffs GMMJ, Wood D, Paton BM, Kayani B, Pollock N, Court N, Giakoumis M, **Head P**, Kelly S, Moore J, Moriarty P, Murphy S, Read P, Stirling B, Tulloch L, van Dyk N, Wilson M, Haddad F. London International Consensus and Delphi study on hamstring injuries part 2: operative management. *Br J Sports Med*. 2023 Mar;57(5):266-277.

Paton BM, Read P, van Dyk N, Wilson MG, Pollock N, Court N, Giakoumis M, **Head P**, Kayani B, Kelly S, Kerkhoffs GMMJ, Moore J, Moriarty P, Murphy S, Plastow R, Stirling B, Tulloch L, Wood D, Haddad F. London International Consensus and Delphi study on hamstring injuries part 3: rehabilitation, running and return to sport. *Br J Sports Med.* 2023 Mar;57(5):278-291.

Conference and professional presentations

P. Head (2018). Acute Neuromuscular Electrical Stimulation (NMES) combined with Blood Flow Restriction (BFR) on physiological, perceptual and hemodynamic variables. Annual conference for the School of Sport, Health and Applied Sciences, St. Mary's University, London. (Poster presentation).

P. Head (2019). Acute Neuromuscular Electrical Stimulation (NMES) combined with Blood Flow Restriction (BFR) on physiological, perceptual and hemodynamic variables. Chartered Society of Physiotherapy Annual Conference, London. (Poster presentation).

P. Head (2019). Efficacy of Neuromuscular Electrical Stimulation (NMES) combined with Blood Flow Restriction (BFR) on physiological, perceptual and hemodynamic variables. Annual conference for the School of Sport, Health and Applied Sciences, St. Mary's University, London. **(Oral presentation).**

P. Head (2019). Using Neuromuscular Electrical Stimulation (NMES) combined with Blood Flow Restriction (BFR) to optimise rehabilitation and return to play from rugby injuries. Annual conference for the England Rugby Football Union, Twickenham, London. **(Oral and practical presentation).**

P. Head (2019). Using Neuromuscular Electrical Stimulation (NMES) combined with Blood Flow Restriction (BFR) to optimise rehabilitation of musculoskeletal disorders. Chelsea and Westminster Hospital, London. **(Oral presentation).**

P. Head (2020). Effect of Neuromuscular Electrical Stimulation (NMES) and Blood Flow Restriction (BFR) on hamstring injury management. Annual conference for the Royal Society of Medicine, London. **(Oral presentation).**

P. Head (2020). Effect of Neuromuscular Electrical Stimulation (NMES) and Blood Flow Restriction (BFR) on acute muscle injury management. Sports MAP Muscle Injury Conference, Australia. **(Online oral presentation).**

P. Head (2021). Using Neuromuscular Electrical Stimulation (NMES) combined with Blood Flow Restriction (BFR) to optimise rehabilitation from acute injury management to return to play. Crystal Palace Football Club, London. **(Online oral presentation).**

P. Head (2022). Using Neuromuscular Electrical Stimulation (NMES) combined with Blood Flow Restriction (BFR) to optimise rehabilitation from acute injury management to return to play. Leeds United Football Club, Leeds. **(Online oral presentation).**

P. Head (2022). Using Neuromuscular Electrical Stimulation (NMES) combined with Blood Flow Restriction (BFR) to optimise rehabilitation from acute injury management to return to play. Brentford Football Club, London. **(Oral and practical presentation).**

P. Head (2023). Using Neuromuscular Electrical Stimulation (NMES) combined with Blood Flow Restriction (BFR) to optimise rehabilitation. Therapy Live 2023. **(Online oral presentation).**

List of abbreviations

ABI: Ankle brachial index	MDC: Minimum detectable change	
AMI: Arthrogenic muscle inhibition	mmHg: Millimetres of mercury	
ANCOVA: Analyses of covariance	MPB: Muscle protein breakdown	
ANOVA: Analysis of variance	MPS: Muscle protein synthesis	
AOP: Arterial occlusion pressure	MRI: Magnetic resonance image	
BFR: Blood flow restriction	MTH: Muscle thickness	
BFRT: Blood flow restriction training	mTOR: mammalian target of rapamycin	
Bpm: beats/minute	MVC: Maximal voluntary contraction	
CI: Confidence interval	MVIC: Maximum voluntary isometric contraction	
CK: Creatine kinase	mVO2: Muscle oxygen consumption	
CPT: Cold pain thresholds	NIRS: Near-infrared spectroscopy	
CSA: Cross-sectional area	Nm: Newton metres	
CV: Coefficient of variation	NMES: Neuromuscular electrical stimulation	
DBP: Diastolic blood pressure	NMES+BFR: Neuromuscular electrical stimulation combined with blood flow restriction	
EIH: Exercise-induced hypoalgesia	NMES+40% BFR: NMES combined with 40%	
EMG: Electromyography	BFR	
ERK: Extracellular signal-regulated kinase	NMES+60% BFR: NMES combined with 60%	
FMD: Flow mediated dilation	BER	
GP: General practitioner	NMES+80% BFR: NMES combined with 80% BFR	
HHb: Deoxyhemoglobin	NPRS: Numeric Pain Rating Scale	
HI-RE: High-intensity resistance exercise	OA: Osteoarthritis	
HR: Heart rate	O2: Oxvaen	
Hz: Hertz	PAR-Q : Physical activity and readiness	
ICC: Intraclass correlation coefficient	questionnaire	
KOOS: Knee Injury and Osteoarthritis Outcome Score	PPT: Pressure pain thresholds	
mA: Milliamperes	RPE: Rating of perceived exertion	
MAPK: Mitogen-activated protein kinase	SBP: Systolic blood pressure	

SCI: Spinal cord injury

SD: Standard deviation

SEM: Standard error of mean

TESTEX: Tool for the assessment of study quality and reporting in exercise

TSP: Temporal summation of pain

US: Ultrasound

- **VA:** Voluntary activation
- VL: Vastus lateralis
- VM: Vastus medialis
- **1RM:** One repetition maximum

Chapter 1

Introduction

1.1 Introduction

Skeletal muscle mass plays a crucial role in overall health and well-being, as muscle tissue quantity directly affects muscle strength, functional capacity and physical performance[1,2]. Muscle size and strength are essential for physical function and quality of life as they enable us to perform activities of daily living with ease and independence, such as walking upstairs and maintaining balance[1,2]. Furthermore, they play a crucial role in preventing chronic diseases, such as osteoporosis and improving overall health and well-being[3]. Numerous situations, such as the recovery from illness or rehabilitation after injury, can necessitate a period of physical inactivity in otherwise healthy individuals. Periods of physical inactivity can result in muscle atrophy, which exacerbates negative health consequences[4,5]. During disuse and periods of immobilisation, skeletal muscle loss occurs at a rate of approximately 0.5% of total muscle mass per day, with strength declines between 0.3% and 4.2% each day[5]. Muscle atrophy and strength declines contribute to numerous negative health consequences including a loss of functional capacity and muscle strength[6–9], the development of insulin resistance[10,11], a decline in basal metabolic rate[12,13] and reductions in physical function[14].

Large reductions in muscular strength and muscle atrophy have also been associated with older age and have been observed in knee OA patients. Functional limitations associated with knee OA include muscle weakness and atrophy, particularly of the quadriceps, which play a key role in stabilising the knee joint[15]. These reductions in muscle strength and size knee OA are due to an inability to perform resistance exercise due to pain and disuse.

1.2 The complex interplay of pain and disuse on skeletal muscle function

Individuals with a history of joint injury or pathology, such as knee OA, commonly experience muscle weakness, activation failure and muscle volume loss[16–18]. The dynamic interplay between chronic pain and disuse poses a complex challenge to

understanding the advanced physiological mechanisms influencing skeletal muscle function.

Arthrogenic muscle inhibition (AMI), is a neuromuscular condition which leads to weakness and atrophy of the surrounding muscles, designed to protect the joint from further damage by inhibiting neural activation[19]. Multiple factors, including tissue damage, pain, inflammation and reduced proprioception contribute to AMI[16-22]. Tissue damage and inflammation activate immune cells and produce cytokines that impair muscle function, while pain reflexively inhibits the muscles, affecting their ability to contract effectively[23]. Altered movement patterns, muscle imbalances and decreased activation occur due to avoidance or modification of painful movements and reduced weight bearing, leading to muscle atrophy[24]. Chronic pain induces alterations in biomechanical parameters through a sophisticated network of molecular crosstalk. Inflammatory mediators, such as cytokines and prostaglandins, orchestrate complex signalling pathways influencing joint mechanics and muscle-tendon unit interactions[20,25]. Joint injury and pathology result in fewer motor neurons available to recruit and a diminished ability to voluntarily recruit motor neurons to a normal extent, due to a central protective mechanism, leading to central activation failure and lesser motor neuron pool excitability[19,26,27]. The muscle inhibition responsible for these observations is reflexive in nature and largely mediated by presynaptic mechanisms within the spinal cord[19]. These changes disrupt the normal neural signalling between joint receptors and the central nervous system, altering the sensory information transmitted to the spinal cord and brain[26]. Mechanisms contributing to AMI include alteration in muscle resting motor thresholds, changes in the discharge of articular sensory receptors, altered spinal reflex excitability and abnormal cortical activity[19,28].

Pain can limit an individual's ability to move and exercise, particularly weight-bearing activities and resistance training in knee OA patients[29]. This limitation affects exercise tolerance and hinders efforts to maintain or improve muscle strength[30]. Pain alters movement patterns, leading to compensatory strategies and alterations in biomechanics. Individuals with knee OA may unconsciously adopt abnormal gait patterns or modify their movements to minimise pain[25]. These altered biomechanics can disrupt the normal distribution of forces across the joint and surrounding muscles, leading to imbalances and increased stress on certain muscle groups, which may contribute to muscle fatigue and weakness over time[20]. Furthermore, disuse-related pain instigates a cascade of events influencing mitochondrial dynamics, subsequently impacting oxidative capacity and muscle endurance. Perturbations in the balance between mitochondrial fusion and fission processes result in impaired energy production, contributing to declines in muscle function[31]. Exercise-induced hypoalgesia (EIH), a phenomenon characterised by a reduction in pain perception following physical activity, is altered in individuals with knee OA [25]. Rather than experiencing reductions in pain sensitivity following exercise, knee OA patients often exhibit increases in pain and symptom flare ups[32,33]. The continuous joint pathology inherent in knee OA, marked by cartilage degeneration, inflammation, and structural alterations, often results in persistent nociceptive input that may override the potential pain-reducing effects of exercise[15]. Additionally, central sensitisation, a heightened responsiveness of central nervous system neurons to peripheral stimuli, is common in knee OA, potentially limiting the effectiveness of exercise-induced analgesia[20]. Psychological factors such as fear of movement and catastrophizing, which are prevalent in knee OA patients, can magnify the perception of pain, making it challenging for exercise to induce significant hypoalgesia. Reduced exercise tolerance due to pain during physical activity may lead to avoidance of exercises that could induce hypoalgesia, hindering the potential benefits of exercise-induced pain reduction. Moreover, the inflammatory milieu in knee OA may counteract the pain-modulating effects of exercise, and the presence of neuropathic pain components can introduce additional complexities in the responsiveness to exercise-induced hypoalgesia[20,25,29,34,35]. These negative effects

of exercise lead to the previously mentioned declines in muscle strength, size and function and interventions are warranted to enhance the clinical outcomes and quality of life for these individuals. Therefore, the impact of pain and pathology on muscle health underscores the critical importance of tailored rehabilitation strategies to prevent or alleviate muscle deterioration in individuals grappling with these complex issues, when exercise is too painful or contraindicated.

The present thesis will focus on developing exercise-mimicking interventional strategies to enhance muscular strength and hypertrophy without requiring voluntary movement to enhance outcomes and recovery during these periods. Finding interventions that positively influence muscle size and strength declines, and all of the physiological declines, discussed previously, due to pain, would potentially enhance functional outcomes and improve the quality of life for these individuals.

This thesis is structured to investigate various interventions that can improve muscle adaptations and alleviate pain in clinical populations. Two primary interventions will be explored: neuromuscular electrical stimulation (NMES) and blood flow restriction (BFR).

1.3 NMES

NMES has emerged as a pivotal tool for enhancing muscle strength in painful clinical populations by directly activating muscle fibres and facilitating essential physiological adaptations. This technique involves the precise application of electrical impulses to targeted muscles, achieved through surface electrodes. These electrical stimuli initiate muscle contractions, effectively replicating the benefits of traditional resistance exercise without requiring voluntary effort. Consequently, NMES recruits a broader spectrum of muscle fibres, including those typically underused during voluntary physical activity. This heightened recruitment fosters muscle strength as muscles adapt to the increased

demand[36]. However, it is worth noting that NMES has shown mixed results in terms of its ability to improve muscle hypertrophy or alleviate pain[37–40].

1.4 BFR

In contrast, BFR presents a distinctive approach to enhancing muscle size and strength in clinical populations while also managing pain[41–43]. BFR involves the application of a pneumatic cuff or inflatable tourniquet proximally on a limb, reducing arterial blood flow into the muscle while impeding venous return. This controlled restriction creates a unique physiological environment within the muscle, characterised by reduced oxygen supply and the accumulation of metabolites, inducing heightened metabolic stress[44]. Consequently, the body responds by releasing growth-promoting factors and recruiting a greater proportion of fast-twitch muscle fibres, typically activated during high-intensity resistance training[44]. Remarkably, even with lighter resistance loads, BFR exercises induce significant muscle fatigue and stimulate muscle hypertrophy. However, BFR requires being combined with voluntary resistance exercise to produce enhanced muscle size and strength, which may not be feasible for certain clinical populations, especially knee OA.

Recently, researchers have investigated the combination of NMES with BFR[45–47]. The scientific rationale is that combining BFR with low-intensity exercise has been found to increase muscle hypertrophy and strength compared to low-intensity exercise without BFR[48,49]. NMES can be classified as low-intensity isometric exercise, as it can evoke maximum voluntary contractions between 10% to 60%, which depends on the current intensity[38]. Thus, theoretically, the combination of NMES with BFR could potentially lead to enhanced muscular adaptations compared with NMES and BFR alone.

The results of trials using NMES and BFR in humans are varied, two studies reported increased muscle strength and hypertrophy compared with NMES and BFR alone in healthy

and spinal cord-injured adults[45,46] Other studies, however, found no added benefit to muscle strength and size[47]. Despite these mixed results, the clinical application for NMES and BFR in preventing atrophy and strength declines, as well as increasing muscle strength and size when in pain, post-surgery or during periods of immobilisation when voluntary exercise is contraindicated, is promising and will be explored in greater detail throughout this thesis.

Chapter 2

Review of the literature; NMES combined with BFR

2.1 Introduction

Skeletal muscle tissue requires a constant process of building up and breaking down of proteins to maintain its structure, with a turnover rate of approximately 1-2% daily[50]. The balance between MPS and MPB determines the net muscle protein balance and consequently, the amount of muscle mass in an individual. A stable muscle mass is maintained when MPS and MPB are in equilibrium[50]. However, disuse of human skeletal muscle alters the dynamics of muscle metabolism, leading to an increased rate of MPB and reduced rates of MPS[51]. For instance, Gibson et al. [9] showed that young men who were immobilised exhibited 30% slower rates of MPS compared to the non-immobilised limb. Further studies confirmed that disuse for 10-14 days caused reductions in MPS ranging from 27-50%[52–54]. Therefore, the available evidence strongly supports the idea that muscle atrophy in humans during a period of disuse is driven by the blunting of MPS[55], emphasising the critical role of physiological processes in shaping the fate of skeletal muscle under conditions of reduced activity.

In the realm of joint injury or pathology, such as knee OA or post-surgery, individuals commonly contend with AMI[19,56]. This neuromuscular condition, an intricate interplay of physiological and neural factors, manifests as muscle weakness, activation failure and volume loss[56]. AMI serves as a protective mechanism, inhibiting neural activation to safeguard the joint from further damage, but consequently leads to declines in muscle strength. Neural contributors to AMI include alterations in muscle resting motor thresholds, changes in articular sensory receptor discharge, modified spinal reflex excitability, and abnormal cortical activity[19,56]. This inhibitory neural response, combined with immune responses and cytokine-mediated impairments, results in altered movement patterns, muscle imbalances, and decreased muscle activation. This comprehensive understanding of both physiological and neural mechanisms in disuse atrophy lays the foundation for targeted interventions aimed at preserving muscle health and function in populations facing joint-related challenges.

Recently, the combination of NMES and BFR has emerged as a novel training method that can enhance muscle strength and hypertrophy without requiring volitional effort[45,46]. This technique involves applying electrical current to create action potentials and contract targeted muscles while simultaneously restricting the blood flow to the same muscle using an inflatable or pneumatic cuff[57]. Various studies have explored the effects of NMES combined with BFR (NMES+BFR) on muscle mass and strength in diverse populations, including healthy adults, obese adults, and patients with neurological disorders[45–47,58]. Despite the growing interest in NMES+BFR, the literature on this topic remains limited and inconclusive.

Therefore, this literature review aims to provide a comprehensive summary of the current state of knowledge on the effects and mechanisms of action of NMES+BFR on muscle mass and strength in different populations. By synthesising the available evidence, this review will contribute to a better understanding of the potential benefits and limitations of NMES+BFR as a training method. Additionally, the review will identify gaps in the existing literature, which will provide research questions that will be examined by this thesis.

2.2 Proposed mechanisms of action

The muscular adaptations resulting from exercise are triggered by the combined effect of mechanical tension, muscle damage, and metabolic stress[59]. However, in cases where voluntary exercise is not possible, such as after injury or surgery, NMES can be used as an exercise mimic, by causing involuntary muscle contractions and evoking mechanical tension[60]. NMES has recently been combined with BFR, owing to its ability to magnify metabolic stress[44]. Recent research has made progress in identifying the physiological mechanisms underpinning the observed muscular adaptations resulting from NMES combined with BFR, in both human and animal models[61–65]. Currently, the evidence suggests that NMES+BFR induces muscular adaptations via the indirect effects of

metabolic accumulation on neuromuscular fatigue and localised hypoxia, resulting in enhanced net protein balance. This is corroborated by the activation of the mitogenactivated protein kinase (MAPK), extracellular signal-regulated kinase (ERK) and mammalian target of rapamycin (mTOR) signalling pathways after NMES combined with BFR in animal model training studies[61,63–66].

The process of muscle growth is a complex and tightly regulated phenomenon that relies on the dynamic interplay between protein synthesis and degradation[67]. Protein synthesis refers to the process of creating new proteins from amino acids, while protein degradation refers to the breakdown of existing proteins[67]. The balance between the rate of protein synthesis and protein degradation ultimately determines skeletal muscle mass[67]. For example, a net decrease in protein synthesis and/or a net increase in protein degradation can lead to disuse atrophy[68]. Moreover, for muscle hypertrophy to occur, the intracellular environment should favour a positive protein balance, which can be achieved by increasing MPS, decreasing muscle protein breakdown, or both[67]. This is crucial for rehabilitation professionals due to the marked muscle wastage that occurs during periods of immobilisation after injury and surgery, leading to reduced levels of MPS[67,68].

Previous research has established that various stimuli, including nutrients, growth factors and NMES, can upregulate protein synthesis in skeletal muscle, primarily at the level of translation initiation[92,94]. A single session of NMES has been shown to increase protein synthesis rates in vivo in humans by up to 27% for a period of 4 hours[94]. This regulation of translation initiation by stimuli is primarily mediated by a protein kinase called mTOR[95]. When NMES is combined with BFR there is a positive muscle hypertrophy response[45,46,69]. Slysz et al. [70] examined the efficacy of NMES+BFR during 14 days of single-leg unloading. The researchers observed that applying NMES+BFR twice a day for 5-days a week prevented atrophy and resulted in mild hypertrophy (+5% [10%]) compared to BFR alone[98]. These results suggest that NMES+BFR can enhance protein

synthesis compared to BFR alone[94,97]. However, BFR applied on its own is insufficient to enhance rates of MPS[96,97].

Accumulation of metabolites during BFR exercise has been suggested as a stimulus of muscle adaptations, as it creates a localised environment of metabolic stress, which stimulates the activation of signalling pathways involved in protein synthesis and cellular adaptation[44]. Greater accumulation of metabolites has been observed after NMES+BFR compared with NMES alone[63]. Additionally, animal models have demonstrated an increase in lactate and H+ concentrations and a decrease in pH following NMES+BFR compared to either intervention on its own[71]. The greater accumulation of metabolites during NMES+BFR appears to induce neuromuscular fatigue earlier than NMES alone[61], through the metabolic stimulation of group III and IV afferent fibres[44]. Several human and animal model studies have shown greater neuromuscular fatigue with NMES+BFR than with NMES alone[61,65,72,73].

Murthy et al. [72] investigated the effect of NMES and varying degrees of BFR on evoked force and muscle oxygenation in the wrist extensors of healthy males using a crossover design. They observed reductions in twitch force when the applied BFR was \geq 60 mmHg, while no effect was observed with lower BFR pressures of 20 and 40 mmHg[72]. In addition, Cole and Brown [73] found evoked force fatigue when NMES was combined with BFR pressures of 160 and 210 mmHg on the calf musculature of young adults, whereas BFR pressures below this did not result in evoked force fatigue[73]. These findings are supported by Wust et al. [65], who observed that evoked force fatigue increased from 33.8% to 68.2% with the addition of a thigh cuff inflated to 240 mmHg, to 5 minutes of NMES in healthy males and females.

In contrast, Satiago-Pescador et al. [74] did not find a difference in volitional fatigue between NMES and NMES+BFR groups. The authors measured volitional fatigue through knee 33

extension maximal voluntary contraction (MVC), unlike the previously mentioned studies, which assessed evoked force fatigue[65,72,73]. Nevertheless, Santiago-Pescador et al. [74] only applied one BFR pressure of 50% arterial occlusion pressure (AOP) which may explain the lack of effect observed, as fatigue only occurred in the higher BFR pressures used by others[72,73]. Animal model data also demonstrates that NMES+BFR induces evoked force fatigue compared to NMES alone[61]. Therefore, to maintain force output from the muscle when fatigued, the lower threshold motor units (Type I) may become insufficient to generate the necessary force. To compensate for this fatigue, the nervous system recruits higher threshold motor units (Type II), which have a greater force-generating capacity. The recruitment of higher threshold motor units is associated with a greater hypertrophic stimulus for muscle fibers, and their recruitment triggers a response that promotes muscle hypertrophy[75,76], resulting in a hypertrophic stimulus for a greater proportion of muscle fibers after NMES+BFR than NMES alone[45,46,61,69].

The recruitment of muscle fibres is a crucial factor in facilitating muscular adaptations as it determines the amount of tension applied to muscle fibres and triggers cellular processes necessary for muscle mass and strength gains[59,77]. Increased recruitment of muscle fibres is one of the underlying neural mechanisms that lead to muscle strength gains[59]. Motor units, consisting of a motor neuron and the muscle fibres it innervates, are the functional units of the neuromuscular system that generate muscle contractions[77]. During voluntary contractions, motor units are recruited according to the size principle, which proceeds from the smallest to the largest[78]. However, research on NMES has reported a non-selective, spatially fixed, and temporally synchronous motor unit recruitment pattern, or a complete reversal of the physiological voluntary recruitment order[36,79]. This means that type II muscle fibres are recruited even when using minimal NMES currents[80].

Li et al. [81] evaluated the effect of BFR, NMES and NMES+BFR during low-intensity squat training on muscle adaptations and muscle activation assessed through surface electromyography (EMG). They observed a greater increase in muscle activation following squats combined with NMES and BFR (NMES+BFR) and squats with NMES alone, compared to squats combined with BFR alone and the control group, who only performed the squatting intervention[81]. Their findings indicated that NMES and NMES+BFR induced greater strength and muscle activation than squats combined with BFR or done alone. Nonetheless, the fatigue mechanism that induces enhanced motor unit recruitment with NMES+BFR needs to be further assessed as a passive intervention. Factors associated with the onset of fatigue include failure of excitation of motor neurons, impairment of action potential propagation in the muscle membrane, conductivity of sarcoplasmic reticulum due to Ca2+ ion concentration, and changes in the concentration of catabolites and metabolites[82].

A mechanism for increased muscle strength and size with NMES+BFR, as opposed to NMES alone, is metabolic accumulation, which promotes a positive protein balance and indirectly induces neuromuscular fatigue[45,46,61,63,65,66]. The hypoxic environment associated with BFR may also promote anabolic signalling within the muscle and induce greater fatigue when combined with NMES[44,83]. Because the availability of oxygen is reduced during BFR, it is suggested that progressive recruitment of additional motor units may compensate for the force production deficit[44,83]. In a study by Nakajima et al. [61], NMES+BFR caused a 72.4% decline in muscle oxygenation levels. Moreover, Murthy et al. [72] demonstrated that reduced twitch force of the wrist extensors in humans was positively correlated with reduced muscle oxygenation in a dose-dependent manner.

Animal model studies indicate that protein synthesis pathways, including the phosphorylation of ribosomal protein rpS6, MAPK signalling pathways, mitochondrial biogenesis and glucose transporter, are enhanced with NMES+BFR compared to NMES

and BFR alone[61,64]. These increases are correlated with increased muscle hypertrophy after 3-weeks of training[61]. For instance, Nakajima et al. [61] provided the first animal model data demonstrating that the reduced partial pressure of oxygen (O2) saturation during NMES contractions with BFR enhanced the phosphorylation response of rpS6 in the mTOR signalling pathways, resulting in the observed muscle hypertrophy (6.2%) and increased mass (11%) after 3-weeks of training. Furthermore, NMES+BFR research observed an increase in Akt phosphorylation at Ser473, S6K1 at Thr389, and S6 at Ser235/236 following the application of NMES+BFR compared to NMES alone in rat gastrocnemius muscles[84]. Akt can activate S6K1 and S6 via mTOR phosphorylation[85]. S6K1 and S6 proteins can also regulate the translation initiation of skeletal muscle proteins[85]. S6K1 at Thr389 is a key regulator of muscle protein synthesis, which is closely associated with muscle hypertrophy after training[51,52]. Therefore, the phosphorylation of S6K1 at Thr389 is frequently used as a measure of mTOR activity[86,87]. Natsume et al. [64] observed that S6K1 and S6 were phosphorylated only when NMES was combined with BFR, suggesting that NME+BFR may enhance post-translational regulation of the mTOR pathway compared with either NMES or BFR alone.

Vascular adaptations have also been observed after NMES+BFR[45]. Arterial reactivity is typically measured as the ability of an artery to increase its size (diameter) in response to an increase in blood velocity, that is, flow mediated dilation (FMD)[88]. Gorgey et al. [45] found a 12% increase in FMD after NMES+BFR, compared to 6% after BFR alone. Previous BFR literature has shown enhancement of FMD[89]. Furthermore, FMD increased by 2.7-6.6% after NMES in the same clinical population[90]. Additionally, NMES training may improve muscle oxidative capacity and result in a fast-to-slow muscle fibre type transition and capillarisation of the stimulated muscles[91,92]. However, the application of NMES+BFR during a 2-week period of immobilisation was insufficient to mitigate artery structure impairments and FMD, suggesting the intervention-induced shear stress may not affect vascular function[93]. The evidence is currently mixed on NMES+BFR promoting
vascular adaptations and microvascular function, therefore remains inconclusive and warrants further investigation.



Fig 2.1. Diagram of NMES+BFR molecular pathways.

In conclusion, the existing research to date on NMES+BFR suggests that this combination can induce muscular adaptations through metabolic accumulation, hypoxia-induced fatigue, enhanced motor unit activation, vascular adaptations and increased protein synthesis, which is supported by the MAPK, ERK and mTOR signalling pathways[45,46,61–65]. However, these findings are not consistent across all studies, and further research is warranted.

2.3 Muscle strength

Muscle strength plays a vital role in human life, facilitating independent living by enabling us to perform essential activities[94]. To date, four human studies and two animal models have investigated the effectiveness of NMES+BFR in enhancing muscular strength compared to NMES alone, BFR alone[45–47,58,61,66,70], or a non-exercising control[47]. The current literature presents mixed findings on the superiority of NMES+BFR over NMES alone in humans for enhancing muscular strength[45–47,58]. Natsume et al. [46] where the

first to conduct a controlled trial and reported increased isometric and isotonic quadriceps strength after two weeks (20 sessions) of NMES+BFR training versus NMES alone. However, Slysz et al. [58] found differences in quadriceps strength improvements after NMES+BFR compared to a non-exercising control leg after a 6-week intervention, but no difference observed between NMES+BFR and NMES alone conditions. Gorgey et al. [45] observed similar improvements in evoked torque after 6-weeks of NMES+BFR and NMES in spinal cord injured patients, with no interaction observed. In a case study, a 15-week NMES+BFR intervention enhanced visually assessed neck stabilisation in a paediatric patient diagnosed with flaccid quadriplegia[22]. In contrast, Andrade et al. [47] reported no change in plantar flexor strength after 6-weeks of NMES+BFR training, potentially due to methodological inconsistencies to be discussed in greater detail below.

The NMES+BFR animal model data consistently shows greater force production after NMES+BFR interventions than NMES alone[61,62,64,66]. The training protocols for these animal model studies ranged from three to eight weeks[61,64,95,96]. Overall, the evidence base largely supports NMES+BFR's effectiveness at enhancing muscular strength, with one human and two animal models finding greater improvements than NMES alone[46,61,66]. However, Andrade et al. [47] reported no benefit, possibly due to the calf muscle group's training protocol and transverse placement of NMES electrodes, which has been shown to be less effective than longitudinal placement in enhancing muscular strength[38,47]. The observed diminished effectiveness of transverse placement of NMES electrodes, compared to longitudinal placement, in enhancing muscular strength is likely attributed to the suboptimal alignment with the natural direction of muscle fibers. This misalignment results in less synchronised and targeted recruitment of motor units and a dispersion of the electrical stimulus across various directions. These factors collectively reduce the effectiveness of NMES in enhancing muscular strength[38,97,98].

Slysz et al. [58] stimulated the same muscle groups as Natsume et al. [46] and Gorgey et al. [77] but obtained contrasting results. Slysz et al. [58] failed to report treatment session adherence, which was above 99.7% in the other studies[46,99], potentially compromising protocol follow-through and treatment tolerability. The BFR method used by Slysz et al. [58] was intermittent and arbitrary (220 mmHg) achieving 100% occlusion in most individuals, whereas the other studies employed a partial BFR stimulus[46,99]. Using a 100% BFR pressure is not effective for enhancing muscular strength and hypertrophy and is not supported by the broader BFR literature[41,48,100]. The different BFR protocols could have affected the results found, as Slysz et al. [58] used the same session frequency but obtained conflicting results to Gorgey et al. [99]. Additionally, Slysz et al. [58] did not report the contraction time, duty cycle used on the quadriceps, or the electrode size or placement for either their upper or lower limb NMES protocols. These elements are crucial for optimising NMES applications [38]. Another difference was that Gorgey et al. [99] used incomplete spinal cord injured patients instead of healthy subjects, which could explain the beneficial effects they found. Patients with neurological conditions tend to have greater muscular adaptations after NMES then non-neurological individuals[101].

The evidence base for NMES+BFR has limitations, as highlighted above and in various studies[46,47,99,102]. To assess the methodological quality of the NMES+BFR training studies, a previously validated assessment tool for the assessment of study quality and reporting in exercise (TESTEX) was used[103]. However, none of the trials used a separate control group, instead using different limbs to act as the control, and they failed to describe how randomisation was carried out. Using the same subject for both the intervention and control groups raises questions about the potential cross-transfer effect from the intervention or control limb, which could have affected the results[104,105]. It is well known that resistance training on one limb can cause increase in muscular strength not only in the trained limb but also in the contralateral, untrained limb due to the cross-education or cross-transfer effect[104,105]. This phenomenon is caused by both neural adaptation[104,105]

39

and muscular hypertrophy[106]. However, it has been suggested that the cross-transfer of strength increase is only due to neural factors[104,105]. This phenomenon has been suggested to occur through a different mechanism during BFR training (BFRT)[107]. Madarame et al. [107] showed that muscle cross-sectional area (CSA) and isometric peak torque of the trained arm muscles increased only when combined with resistance exercise with BFRT for the legs. They suggested that an increase in noradrenaline was involved in this remote effect of exercise on muscular size[107], but it was not different compared with no BFR and this finding has not been substantiated in further research. Furthermore, no instructions in the present NMES+BFR studies was given on what the participants should avoid during the training protocols. This confounding factor could have affected the results observed with activity undertaken away from the studies affecting the muscular adaptations observed. Using separate participants for both experimental and control groups is recommended in future research.

After BFRT, muscular strength and hypertrophy gains have been found both distal and proximal to the cuff location, suggesting systemic mechanisms may promote these adaptations[48,100]. Increased supraspinal activation has also been reported after BFR resistance exercise compared with conventional resistance exercise[108]. This phenomenon has also been found with NMES[109]. In rabbits, it was shown that repetitive unilateral muscle overuse caused by NMES overtime leads to both degenerative and regenerative tissue changes and myositis not only in the exercised muscles but also in the homologous non-exercised muscles of the contralateral leg[109]. Yong et al. [109] suggested that the nervous system is involved in the cross-transfer effects observed during NMES. This has also been supported in numerous other studies showing the spinal and supraspinal effects of muscular adaptations from NMES[39]. The cross-transfer effects observed in previous NMES and BFRT studies could have affected the NMES+BFR results observed to date[46,47,99,102].

The stimulation protocols employed in NMES+BFR training studies varied in frequency (ranging from 20-100 Hz), pulse width (400-450 µs), duty cycle (5-8 s on and 3-600 s off), and other parameters[46,47,99,102]. However only one study reported the ascent and descent phases of the stimulation cycle, pulse type and shape[47], and the number of stimulation repetitions[99]. Another limitation was the lack of reporting electrode size[46,99], which is crucial for replication, comfort, and evoked force production[38]. The methodological inconsistencies are discussed in greater detail below.

2.4 Muscle size

Muscle strength is crucial for human life and independence, while muscle size can also influence overall physical function by increasing strength and endurance[94]. The current evidence base for NMES+BFR again shows mixed and varied results regarding whether this combination is effective in producing muscular hypertrophy in humans in six human trials [46,47,58,70,99]. Gorgey et al. [77] discovered a between-group difference for extensor carpi radialis longus hypertrophy, which was 17% greater than NMES alone. The extensor carpi radialis longus is a forearm muscle positioned on the posterior (dorsal) aspect of the forearm, along the lateral (radial) surface. Originating from the lateral supracondylar ridge of the humerus to its distal attachment at the base of the second metacarpal bone in the hand, contributing to its role in extending and abducting the wrist joint[110]. Additionally, they found greater improvements in FMD changes compared to BFR alone, although no other between-group differences were observed. Notably, there was no extensor digitorum communis hypertrophy change. Since the forearm muscles affect various finger motions, previous NMES research has found it possible to selectively activate the extension/flexion of most fingers by stimulating the forearm muscles[111]. Thus, the physiological characteristics of each finger should be considered when placing the negative electrode for selective stimulation of individual fingers[111]. Therefore, electrode placement could have confined the stimulus to the extensor carpi radialis longus. Additionally, the extensor carpi radialis longus is a greater prime mover than extensor digitorum communis[111], and thus the observed middle finger extension used as a point to cease increasing NMES current intensity could have stimulated the extensor carpi radialis longus but not promoted extensor digitorum communis hypertrophy.

Skiba et al. [69] assessed the effect of NMES+BFR vs. NMES alone on the quadriceps of complete spinal cord-injured patients. They observed greater quadriceps muscle thickness increases after eight weeks of NMES+BFR compared to NMES alone on the paralysed muscles[69]. The same result was observed by Gorgey et al. [45] in spinal cord-injured patients reporting a 17% greater increase in extensor carpi radialis longus CSA in the NMES+BFR forearm compared with NMES alone. Additionally, they reported that NMES+BFR induced a 15% increase in extensor carpi radialis longus CSA. In obese, but otherwise healthy adults, Natsume et al. [46] found quadriceps size increased from pre- to post-training after NMES+BFR, with no change observed for NMES alone. However, there was no between-group difference. More recently, Bergamasco et al. [112] studied the effect of NMES+BFR compared to low-intensity resistance exercise combined with BFR on vastus lateralis CSA, after a six-week intervention. They observed greater hypertrophy after lowintensity resistance exercises combined with BFR vs. NMES+BFR. However, the pre-topost change observed after the six-week NMES+BFR intervention was (4.6%), indicating a hypertrophic response. In contrast, Slysz et al. [58] found neither NMES+BFR nor NMES alone enhanced quadriceps hypertrophy. Additionally, Andrade et al. [5] observed no change in soleus muscle thickness after six weeks of NMES+BFR training. This study is the only one to date that has found no beneficial within-subject effect of NMES+BFR on muscular adaptations. The authors state that this could have been because the soleus muscle is comprised primarily of slow-twitch fibres, which have a lower hypertrophic potential in comparison to type II fibres[47]. The previous statement is true[113]; however, BFRT has been shown to cause hypertrophy in both type I and II fibres[16–18], with some evidence showing preferential type I fibre hypertrophy[19]. Andrade et al. [5] may have seen an effect of the intervention if they utilised a larger sample size than seven, since they showed a 15.2% change in muscle thickness. Additionally, their strength results show greater muscle size improvement in their control leg than what they observed in the intervention leg. The exercise and/or physical activities that the participants were undertaking away from the experimental sessions could have contributed to the results observed by Andrade et al. [47]. Slysz et al. [70] reported a mild increase in hypertrophy (+5% [10%], p = 0.07) over a 14-day period of unloading. This suggests that hypertrophy can occur even in the absence of voluntary motion during unloading.

BFR protocols also varied in the studies, with differences in the pressure applied, cuff size, and occlusion duration. quite largely with one study causing full occlusion[45–47,70,112,114]. One study induced full occlusion using a pressure of 130% of systolic blood pressure (SBP) for the upper limb[45]. SBP represents the pressure on arterial walls during heart contraction (systole) and indicates the maximum pressure during a heartbeat[99]. One study based their BFR pressure on thigh circumference[46], while one referenced previous BFRT research but used a different cuff width than the original study[47]. Additional studies employed an arbitrary pressure, applied intermittently[70,112]. Two studies did not report the cuff size used. When applying BFR the cuff width is an important consideration because using the same pressure with a wider cuff will apply a greater restrictive stimulus[115]. Standardising the BFR stimulus according to arterial occlusion pressure has been recommended in the wider BFR literature to enhance efficacy and safety during its application[57,116,117]. Furthermore, measuring every individual's AOP in the position that the intervention will be undertaken further enhances its accuracy and reliability[116] which to date has not been utilised in NMES+BFR research.

Human trials investigating muscular hypertrophy have produced inconsistent results[45–47,58,69,112]. While three trials found a between-group difference[45,46,69], one found within-group increases[58] and one observed no effect[47]. These discrepancies may be due to methodological differences between the studies, including variations in the

parameters of NMES and BFR, intervention lengths and methodological rigour previously discussed.

Overall, the current evidence suggests that the combination of NMES+BFR may be more effective than either intervention alone in inducing muscular adaptations. However, the studies conducted thus far have varied considerably in terms of methodology. Therefore, the later sections of this review with synthesise the methods employed in these studies and provide guidance for future research requirements.

2.5 NMES parameters

The NMES parameters employed in current NMES+BFR research are largely varied, unclearly reported, and not recommended by the wider NMES evidence base[37,38,118]. This may explain the lack of consistent muscular adaptations observed compared with previous NMES research. One of the main drawbacks of NMES treatments is the fact that they can be too painful, with increasing current creating further discomfort[38,119,120]. An ideal electrical stimulation application should be able to increase muscle strength with the least possible discomfort during stimulation[121]. However, the muscular adaptation observed is determined by the current intensity applied[84]. Therefore, to maximise adaptation, the methodology should focus on applying the greatest current intensity with the minimal amount of discomfort[38,84,121,122]. Just like any other form of training, appropriate programming is required to achieve the desired result. Using appropriate NMES parameters, electrodes and locating motor points is crucial for optimising muscular adaptations from its application[38].

2.5.1 Frequency

The frequency used during NMES interventions has been identified as a critical factor in determining its effectiveness on muscular adaptations[38]. Frequency refers to the pulses

produced per second during stimulation and is measured in Hz, where 40 Hz equates to 40 pulses per second[60]. Studies have shown that an optimal frequency range of 30-50 Hz is effective in enhancing muscular adaptations[118]. However, for the quadriceps, a frequency of 50 Hz has been recommended[123], due to this frequency minimising discomfort and increasing quadriceps strength compared to frequencies below this[123]. Moreover, tetanic muscle contractions are produced at a frequency rate of approximately 50 Hz in human muscles[36].

However, the NMES+BFR training and acute studies have used frequencies that ranged from 1-100 Hz[46,47,58,63,69,70,72–74,93,99,112] or have not been reported[65,124,125]. Slysz et al. [58] have reported using frequencies between 50-100 Hz, but it is unclear whether they ever used the recommended 50 Hz for the quadriceps[38,118]. No other NMES+BFR studies to date have utilised a 50 Hz frequency on the quadriceps[46,69]. The other NMES+BFR training studies used frequencies outside of what is recommended[45,69,70,112,114]. This may help explain the mixed effects observed.

In summary, frequency plays a crucial role in determining the effectiveness of NMES interventions on muscular adaptations. An optimal frequency of 50 Hz for the quadriceps, has been recommended by the wider evidence base[38,123,126]. However, the frequencies used in the NMES+BFR literature have not used this frequency, vary widely and are often not reported, which may explain the inconsistent results observed to date. It is essential to use appropriate frequencies, 50 Hz for the quadriceps, and report them clearly in future studies to maximise the effectiveness of NMES+BFR interventions.

2.5.2 Pulse width/duration

Electrical stimulation devices produce pulses in waveform patterns that are commonly represented by geometric shapes such as square, peaked, or sine waves[38]. These

shapes characterise electrical current that either rises above a zero baseline for the duration of the stimulation paradigm (uniphasic; e.g., direct current), or alternates above and below the baseline (biphasic or alternating current)[127]. The pulse width or pulse duration refers to the time span of a single pulse. In biphasic pulses, the pulse duration considers both phases[128]. A previous study compared pulse widths of < 200 µs vs. > 200 µs delivered to the soleus muscle and found that wider pulse widths produced stronger contractions of plantarflexion and augmented overall contractile properties[129]. Longer pulse durations typically penetrate more deeply into subcutaneous tissues, so these widths should be used when trying to impact secondary tissue layers[130]. Pulse widths between 400-600 µs are recommended in the wider NMES literature as they selectively target motor fibres without negatively impacting muscle fatigue or metabolic demands[131]. Kesar et al. [131] conducted a crossover study assessing pulse widths between 150-600 µs and observed greater evoked force, without fatigue, between 400-600 µs. Additionally, pulse durations closer to 400 µs produce greater guadriceps cross-sectional activation compared with 150 µs[132]. Natsume et al. [46] failed to report the pulse width applied and the other NMES+BFR training studies reported pulse widths in the recommended range (400-450 µs)[45,47,69,70,112,114]. The fact that the majority of NMES+BFR training studies used recommended parameters, pulse width does not seem to be a contributing factor to the mixed results currently observed.

However, we propose that pulse duration could be manipulated to fit with the recommendations in the wider strength and conditioning literature to optimise time under tension and rest periods of the interventions[133,134]. The current NMES+BFR training studies use contraction durations between 4-15 s[45–47,58,69,70,112], with rest periods after every NMES contraction ranging between 3-18 s or not reported[47,58]. Previous strength and conditioning research found greater levels of protein synthetic rates when implementing a time under tension of 6 s compared with 1 s[133]. The greatest muscle adaptations after NMES+BFR used contraction times of 5-15 s and rest between contractions of 3-5 s with 8-20 minutes stimulation periods each session[46,58,69,99]. The

work to rest ratio utilised by these studies were replicated in an animal model by Natsume et al. [64] to support greater muscle hypertrophy compared to NMES alone. This work period is equal to or greater than the rest period and differs from what is recommended in the wider NMES literature[38,123,135]. When applying NMES, it is generally recommended to have longer rest periods compared to the work periods, to reduce the effects of neuromuscular fatigue[38,123,135]. However, in the strength and conditioning evidence base, acute reductions in force output (i.e., fatigue) have been shown to correlate to increased chronic muscular adaptations[136,137]. This finding of acute fatigue leading to chronic muscle hypertrophy has been shown after NMES+BFR in rats[61]. Furthermore, Slysz et al. [58] reported elsewhere that they used a 600 s rest period between NMES contractions, making their ratio 1:150. The larger rest period used by Slysz et al. [58] may have contributed to their lack of effect compared with NMES alone. Therefore, the current NMES+BFR research suggests using a work ratio that is equal or greater to the rest period between stimulated contractions[45,46,61,64]. However, this approach differs from recommendations for NMES alone and warrants further investigation to optimise parameters for enhancing muscular adaptations.

2.5.3 Amplitude/ Intensity

A parameter that plays a role in inducing muscular adaptations is the strength of the current being administered, which is usually reported in milliamperes (mA). The intensity of electrical stimulation is commonly referred to as the current amplitude and is directly related to the amount of muscle activated during electrical stimulation[38]. Magnetic resonance imaging (MRI) has been used in previous studies to map the pattern of activation after repeated NMES evoked isometric contractions[138] These studies have found that increasing the intensity of electrical stimulation resulted in the activation of more motor units, which subsequently increased force output[36,84,138]. Therefore, increasing the current amplitude may lead to greater muscle recruitment and, thus, enhance the potential for muscular adaptations[84].

Higher intensities of electrical stimulation result in a stronger depolarising effect in the structures underlying the electrodes[139], and can lead to increases in strength[140–142]. However, patient comfort and tolerance must also be considered, as higher intensities are typically less tolerated. Nevertheless, both frequency and intensity play a crucial role in determining the quality of muscle contraction produced[143].

Author (year)	Frequency	Pulse width	Duty cycle	Intensity	Motor points	Duration	Training frequency and duration
Natsumo et al	20 Цт	ND	9 c op / 2	E 10% M///C	located	20 mins 4 v F	2 wks. 2 sossions a
Natsume et al.	50 HZ	INK	8 S 01 / S	2-10% MINIC	TES	20 mins, 4 x 5	z wks, z sessions a
(2015)			s off			mins	day (20)
Gorgey et al.	20 Hz	450 µs	5 s on / 5	Wrist	No	7.5 mins	6 wks, 2 sessions a
(2016)			s off	extension			wk
Andrade et al.	35 Hz	400 µs	6 s on	20% MVIC	No	8 mins	6 wks, 3 sessions a
(2016)							wk
Slysz et al.	50-100 Hz	400 µs	NR	Max	No	12 mins, 3 x 4	6 wks, 4 sessions a
(2018)				tolerable		mins	wk
Skiba et al.	20 Hz	400 µs	15 s on /	Knee	No	12 mins, 3 x 4	8 wks, 2 sessions a
(2021)			4 s off	extension		mins	wk
Bergamasco et	75 Hz	400 µs	4 s on /	8/10 pain	Yes	20 mins, 4 x 5	6 wks, 3 sessions a
al. (2021)			18 s off			mins	wk (20)

 Table 2.1. NMES parameters for NMES+BFR training studies.

NR: not reported; wks: weeks; s: seconds

The evidence base for NMES consistently demonstrates that the greater the muscular strength and hypertrophic adaptations observed, the higher the maximal tolerable intensities used[77,92,144,145]. However, to date, only Slysz et al. [58] have utilised max tolerable intensities in the NMES+BFR literature and has multiple methodological flaws which could have contributed to their lack of observable effect[102]. The other NMES+BFR training studies used a percentage of MVC to determine the current ranged from 5-20%[46,47] and others saying it was determined when movement occurred[45,69]. While beneficial muscular adaptations have been observed after intensities that elicited 5-10% MVC by Natsume et al. [46] is much lower than what is seen to cause muscular adaptations after NMES alone using maximal tolerable intensities[77,92,144,145]. Evoked force data

from Chapter 4 (Fig 4.3) found that NMES+BFR at maximal tolerable intensities produced 8-12% MVC, which could be attributed to the cuff placement for BFR, reducing the space between the anode and cathode electrodes causing less of the muscle to be stimulated, compared to NMES alone studies reporting >50% MVC's using max tolerable currents[77,92,144,145]. Therefore, basing the NMES intensity on the MVC produced could be a limiting factor for the current NMES+BFR literature[45–47]. Future NMES+BFR research should utilise max tolerable intensities to optimise potential benefits for enhancing muscular adaptations, regardless of evoked force, as recommend by the wider NMES literature.

2.5.4 Electrode placement and size

The success of NMES current in reaching underlying tissues is highly dependent on the size and placement of electrodes[111,146–148]. Electrodes placed on muscle motor points have an impact on muscle response and should be carefully considered[60]. The motor point is the specific location on the skin over a muscle where a contraction can be electrically induced with the lowest current amplitude[147]. As skin and tissue resistance to current is lowest at this point, patient discomfort is minimised, and evoked force is maximised[147,148]. Therefore, placing electrodes over motor points is crucial in enhancing the effectiveness of NMES[148]. Current intensity is associated with greater gains in muscle strength[84], thus, it is essential to use all possible techniques to maximise current and comfort[148].

A recent study comparing electrode placement using motor points of the muscle accurately located through stimulation compared with the recommended sites of several manufacturers showed differences in muscle performance outcomes in NMES delivered to the tibialis anterior and the vastus lateralis (VL) of the lower extremity[148]. The motor point placement not only produced higher torques but also increased blood flow and oxygen use,

making it more effective[147]. Electrode size is another essential variable, with larger electrodes (> 20 cm²) promoting deeper current penetration and being more comfortable for the patient, leading to greater tolerance[60,97]. Using small electrodes (< 20 cm²) can result in inadequate motor unit recruitment and reduced effectiveness[97].

In the current NMES+BFR literature, Natsume et al. [46] and Bergamasco et al. [112] are the only groups that located muscle motor points for electrode placement, which is a major limitation as previously mentioned. All NMES+BFR training studies used electrodes larger than 20 cm² in size[46,47,58,132]. However, for quadriceps, the greater muscle adaptations occur with one large proximal electrode and two smaller electrodes over the vastus medialis and vastus lateralis motor points[123]. Although Natsume et al. [46] and Bergamasco et al. [112] used this method, Slysz et al. [58] did not, which could explain the reduced effectiveness observed in their study[58].

2.6 BFR parameters

The vast majority of studies combining NMES and BFR have prescribed an arbitrary restrictive pressure[47,58,63,124] or based their occlusion pressure on systolic blood pressure (SBP)[99]. However, recent findings suggest that neither approach is effective for controlling the magnitude of BFR[57,117]. The current recommendation is to prescribe pressure via AOP[116]. Our group previously recommended determining AOP for prescribing BFR pressures[117], as it allows for standardised pressures regardless of equipment used or participant size. This is crucial since previous research shows that the restrictive stimulus produced is influenced by a variety of factors, including cuff width and the size of the participant's limb[115,149,150].

For the lower limb, AOP is determined using a handheld Doppler probe placed over the posterior tibial artery[151,152], with an inflatable or pneumatic cuff placed around the most proximal portion of a participant's thigh, and inflated until no auscultatory pulse is detected by the Doppler probe[153]. Other methods of determining BFR pressures used in the NMES+BFR literature include 130% SBP[45], basing the pressure on participant limb size characteristics[46] or arbitrary pressures[47,58]. Using 130% SBP as a BFR cuff pressure, creates full arterial occlusion of the upper limb[115]. Loenneke et al. [149] found that wide cuff inflation at 130% SBP (130% of \sim 120 = 156 mmHg) would exceed the necessary pressure for complete arterial occlusion (144 mmHg) of the lower limb in healthy individuals. Additionally, applying the same pressure and using a different cuff width or a larger participant will affect the BFR stimulus, creating methodological inconsistency that makes it challenging to analyse and compare the effectiveness of previous NMES+BFR trials[115]. Previous research suggests that the degree of BFR may be influenced by the amount of tissue surrounding the blood vessel, which affects the pressure exerted on the vasculature[115]. Limb circumference explains most of the variance in the cuff pressure required to occlude arterial flow[149]. Wide BFR cuffs require lower pressure to restrict

arterial blood flow compared to narrow cuffs, with limb size and composition having a greater influence on the pressure[115]. The greatest muscular adaptations from NMES+BFR currently occurred when using a BFR pressure relative to the participant characteristics[46], 50% AOP[69] and 130% SBP[99]. Using AOP may help future research and may lead to greater muscular adaptations, as seen in the wider BFR literature[41,57].

Three NMES+BFR training studies[69,70,112] have used AOP to prescribe their BFR pressure. The NMES+BFR training studies showed beneficial effects when using AOP, although two of them used a 100% pressure[70,112] which is not recommended[57]. In animal models, Natsume et al. [71] reported using a cuff pressure approximately 40-60% of AOP, while Nakajima et al. [61] used a BFR pressure that considerably lowered O2 partial pressures but did not completely occlude blood flow. This may be interpreted as being

51

above 60% AOP but below 100% AOP, consistent with previous research on humans that found muscle oxygenation/deoxygenation during <60% AOP is not substantially different than during non-BFR exercise[154]. Reis et al. [154] concluded that 60% AOP appears to represent a threshold required to induce higher deoxygenation and decreased tissue oxygenation levels[154]. Skiba et al. [69] used 50% AOP combined with NMES and observed muscle growth compared with NMES alone. Therefore, current limited data supports the use of partial pressures of BFR when combined with NMES[57], although not conclusive.

During BFRT, cuff-related nerve injury is a potential complication that can be harmful[155,156]. The extent of nerve damage caused by cuff use can range from mild transient loss of function to permanent, irreversible damage[155]. Most cases of nerve damage occur beneath or near the edges of the cuff as shown by Ochoa et al. [156]. Despite the physiological and perceptual similarities to regular exercise, BFRT must be performed safely[157,158]. Rhabdomyolysis is another potential complication that has been reported after BFRT[159-162]. It involves the rapid dissolution of damaged or injured skeletal muscle, ranging from an asymptomatic illness with elevated creatine kinase (CK) levels to life-threatening conditions associated with extreme CK elevations, electrolyte imbalances, acute renal failure and disseminated intravascular coagulation[163]. All cases reporting rhabdomyolysis and acute vision loss after BFRT used arbitrary pressures and did not standardise the restrictive stimulus using AOP[160,161,164]. The protocols or cuff widths used were not reported in these cases, and this could have affected the applied restrictive stimulus, potentially contributing to these adverse events[159-162]. Furthermore, Martin-Hernandez et al. [124] observed syncopal episodes in three healthy young people during their NMES+BFR study. They also did not use AOP to determine their restrictive stimulus. Spranger et al. [165] proposed that BFRT, if performed incorrectly, may result in deleterious hemodynamic responses. Using AOP to determine BFR restrictive pressures may reduce the risk of adverse events from occurring and hemodynamic variables need to be assessed in NMES+BFR studies.

2.6.1 BFR frequency

The optimal training frequency for BFR is typically found to be between 2-3 sessions per week[48,57]. However, the current literature on NMES+BFR has employed various training frequencies, including 2 sessions per day, 5-days a week, for 2-weeks (20 sessions)[46]; 6-weeks with 3 sessions a week[47,112]; 6 weeks with 2 sessions per week[45]; 6 weeks with 4 sessions per week[58] and 8 weeks with 2 sessions per week[69]. Three beneficial and one non-effective study used the recommended training frequencies, however, the greatest effect was seen when using 2 sessions a day for a 2-week period.

In NMES research, training 2-3 times a week has also demonstrated beneficial effects on muscular adaptations[92]. Notably, Gorgey et al. [166] found muscular adaptations even with just one session a week. However, the twice-daily, 5-day-per-week protocol utilised by Natsume et al. [46] is not feasible for standard clinical practice. Therefore, future investigations should explore less frequent protocols that still yield muscular adaptations in order to determine the optimal training parameters.

2.7 Effect of NMES and BFR on pain

Pain modulation refers to the process by which the body alters a pain signal as it is transmitted along the pain pathway and involves interactions between local and central nervous system mechanisms[167]. NMES potentially alleviates pain through mechanisms such as activating large-diameter sensory fibers and triggering endorphin release. Its impact on pain modulation varies across conditions, including postoperative pain, musculoskeletal injuries, and chronic pain. While studies suggest positive effects, individual responses differ[126,168]. BFR training has been associated with higher levels of perceived muscle pain and RPE compared to traditional exercise without BFR[169,170]. The same higher perceptual ratings of pain and RPE have also been observed during NMES+BFR compared to NMES alone training sessions[46]. Furthermore, BFR, when combined with resistance and aerobic exercise, has shown reductions in pain intensity (pressure pain thresholds) in

healthy adults and patients with knee problems or lateral elbow tendinopathy[171]. As a result, it has been suggested that the BFR component may trigger a hypoalgesia response, via exercise-induced hypoalgesia (EIH), similar to high-load resistance exercise[171]. In pain-free individuals, the typical response to an acute exercise session is a phenomenon known as EIH, which is characterised by a temporary decrease in sensitivity to painful stimuli[172]. In research conducted in laboratory settings, EIH is commonly assessed by applying a painful stimulus to the body before and after a specific exercise regimen and evaluating changes in pain sensitivity, such as increased pain thresholds or reduced pain intensity in response to a standardised painful stimulus[173]. These methods have consistently demonstrated the presence of EIH in healthy individuals without pain[33]. Both aerobic and resistance exercise have shown to attenuate various aspects of pain sensitivity, including pressure, thermal and mechanical pain[33,174,175].

In contrast, the phenomenon of EIH exhibits greater variability and impairment in populations suffering from chronic pain, where pain and pain sensitivity may decrease, remain unchanged, or even increase in response to exercise[33]. Pain worsening during exercise can pose a barrier to adherence, potentially leading to a cycle of physical inactivity that can further exacerbate pain and disability in the long term. In individuals with knee OA, previous research has demonstrated evidence of nervous system sensitisation, which appears to exist on a spectrum[25]. Quantitative sensory testing in knee OA patients has identified features of pain sensitisation, such as heightened responsiveness to mechanical and thermal stimuli[176]. Moreover, direct measurements have revealed central hyperexcitability, and a lack of conditioned pain modulation, indicative of abnormal endogenous pain regulation observed in individuals with knee OA[25,32]. Regarding EIH, researchers have discovered distinct responses to EIH among knee OA patient's, differentiating those with normal and abnormal endogenous pain inhibition from pain-free controls[32]. Knee OA patients exhibit heightened pain sensitivity (decreased pressure pain

thresholds) in response to exercise, suggesting dysfunction in EIH[32]. This implies that exercise interventions may induce increased pain in knee OA patient's, which could hinder muscle strength and function.

NMES has been explored as an alternative treatment for various chronic pain conditions[126]. Both experimental and clinical studies have provided evidence supporting the use of NMES in managing neuropathic pain, including cases involving traumatic nerve damage, neuroma, postherpetic neuralgia and diabetic neuropathy[168,177]. Moreover, there is moderate to strong evidence indicating the effectiveness of NMES in addressing post-stroke issues, as well as weakness and pain following ACL-R, total knee replacement and knee OA[126]. However, no research has been conducted thus far to examine the efficacy of NMES inducing EIH in musculoskeletal disorders.

BFRT has demonstrated pain reduction throughout training programs in various clinical conditions, such as knee pain, OA and post ACL-R[178–181]. Interestingly, low-load BFR appears to yield greater pain reduction than high-load resistance exercise. Existing literature suggests the presence of a hypoalgesia effect associated with BFR[178–181].

Recent evidence indicates that EIH occurs when low-intensity exercise (e.g., \leq 30% of one repetition maximum [1RM]) is performed with BFR[182,183]. In individuals with knee pain, a single bout of low-intensity knee extension exercise with BFR was found to immediately reduce anterior knee pain compared to the same exercise without BFR[182,183]. This pain reduction lasted for at least 45 minutes and enabled the individuals to tolerate exercise with greater loads during the period[183], consistent with the findings of Giles et al. [179] in patellofemoral pain patients during a training intervention.

55

Hughes and Patterson [43] conducted a study to investigate the mechanisms underlying the hypoalgesia induced by BFR exercise. They employed a four-arm, crossover design involving low-intensity resistance exercise at 30% of 1RM with and without BFR, as well as high-intensity resistance exercise arm (HI-RE) at 70% 1RM. The duration of the HI-RE protocol matched that of the low-intensity portions of the trial, allowing for consistent pre- and post-exercise pain assessments across all arms. The authors observed that BFR resulted in greater hypoalgesia compared to free-flow low-intensity resistance exercise, both in the exercising limb and in remote muscles. The HI-RE group and both BFR groups exhibited similar reductions in pressure pain in remote muscles. In the exercising limb, BFR at 40% AOP produced inhibition comparable to HI-RE, while BFR at 80% AOP elicited the greatest increase in pressure pain threshold response[43]. Notably, pain inhibition was still present in the exercising limb 24 h after exercise in both arms. The authors observed increased plasma beta-endorphin concentration, a component of the endogenous opioid system, which contributed to the hypoalgesia effect[184].

The hypoalgesia effects of BFR and the potential impact of combining BFR with NMES have yet to be investigated. Such investigations could provide further insights and potential benefits of this combined intervention.

2.8 Is NMES combined with BFR tolerable?

In the past, both NMES and BFR have been described as painful interventions, which has been a primary limitation[122,169]. This is particularly evident in NMES interventions, where the pain induced by the electrical current has been identified as a major barrier to its use and effectiveness in enhancing muscle strength and function[38–40]. Available data shows that combining BFR with NMES results in greater pain and rating of perceived exertion (RPE) than NMES alone[185,186]. However, two studies reported adherence rates ranging

from 99.7% to 100% without any reported adverse events[46,99]. The other NMES+BFR training studies did not report adherence rates or adverse event occurrence[47,58,69,112], except for one participant withdrawal in the study by Slysz et al. [58]. The high adherence rates reported in studies showing benefits from combining NMES with BFR in both healthy and clinical populations on muscular adaptations could have contributed to their positive findings, contrasting with the lack of benefits found by others[47,58]. Natsume et al. [46] observed higher pain scores and RPE when NMES was combined with BFR compared to NMES alone. Despite the high pain and RPE ratings associated with the addition of BFR to NMES in previous reports, the extremely high adherence rate (> 99.7%) suggests that the NMES+BFR intervention was tolerable[46]. Slysz et al. [58] reported one participant withdrew due to reasons unrelated to the study. On the other hand, Andrade et al. [47] did not provide information on adherence rates, withdrawal rates, or adverse events. The lack of a mention of adverse events and the measurement of pain and RPE may have contributed to a more uncomfortable intervention in these two studies, potentially explaining the reduced effectiveness observed.

2.9 Conclusion

The evidence regarding the effect of NMES+BFR on muscular strength and hypertrophy is promising but inconsistent[46,47,58,99]. The mixed and varied responses observed could be attributed to methodological inconsistencies and the failure to apply appropriate parameters of NMES and BFR as determined in the broader literature for both interventions[47,102]. Limited mechanistic and chronic training study data suggest that the BFR stimulus should be between 40-80% AOP and the NMES parameters should include a frequency of 30-50 Hz, with the electrodes placed over the motor points of the muscle[45,61,64,66]. Future research should investigate the efficacy of NMES compared to NMES+BFR in separate subject groups, rather than using a contralateral limb, and determine the minimum AOP requirement for the BFR stimulus. The combined intervention

appears to be safe and tolerable, with no reported adverse events and high rates of intervention adherence documented in the literature to date.

BFRT has gained popularity as a rehabilitation method, especially when combined with light load resistance exercise[41,48,100]. However, its efficacy in reducing muscle atrophy and weakness without exercise is limited[187-190]. On the other hand, NMES has been effective in reducing muscle atrophy during immobilisation but has not consistently enhanced muscular size or strength in some clinical populations[191–193]. The combined use of NMES and BFR represents a new and novel rehabilitation approach. It has demonstrated greater effectiveness than NMES or BFR alone in enhancing muscular strength and hypertrophy in healthy and spinal cord injured populations[45,46]. However, the methodologies used are mixed and not well-defined, and conflicts with the broader NMES and BFRT evidence bases are evident. The hemodynamic safety of this combined intervention is uncertain, with reports of syncope episodes in the current literature, although this occurred when BFR was applied alone[124]. Further research is needed to investigate the mechanisms of action for NMES+BFR[61-66], but available data suggests similar effects as reported in the broader BFR literature[44,83]. However, the combined effect of both interventions could be synergistic or detrimental and requires further investigation to determine its potential for aiding clinical rehabilitation and muscular adaptations, especially in patients unable to perform volitional or light load resistance exercise. This thesis will help to establish the most beneficial intervention prescription, if any, for maximising muscular adaptations when voluntary exercise is not possible.

2.10 Thesis aims and research questions

2.11 Aims

- 1. To investigate and identify the most effective and safe methodologies when combining NMES and BFR.
- To evaluate whether NMES+BFR is more effective in increasing muscular strength and hypertrophy compared to NMES alone.
- 3. To analyse the underlying mechanisms contributing to muscular strength and hypertrophy adaptations resulting from the application of NMES+BFR.
- 4. To determine whether NMES+BFR produces hypoalgesia effects similar to those observed when BFR is combined with traditional resistance training.
- 5. To assess the safety and efficacy of utilising NMES+BFR in clinical populations, with a specific focus on its potential to enhance muscular strength, hypertrophy, and pain.

2.12 Null Hypotheses

- 1. There is no significant difference in efficacy and safety between various methodologies of NMES and BFR when combined.
- NMES+BFR does not lead to a greater increase in muscular strength and hypertrophy compared to NMES alone.

- 3. There are no specific mechanistic contributions to muscular strength and hypertrophy adaptations resulting from NMES+BFR.
- NMES+BFR does not produce hypoalgesia effects similar to those observed with BFR combined with traditional resistance training.
- 5. The use of NMES+BFR in clinical populations does not result in significant improvements in muscular strength, hypertrophy, or pain when compared to standard rehabilitation protocols.

Chapter 3

General methods

This chapter will provide an overview of the research methodology, data collection procedures, participant selection, analysis techniques and ethical considerations employed throughout the studies of this thesis.

3.1 Health and safety

Full ethical approval for the experimental designs and procedures was obtained from the Ethics Subcommittee of St Mary's University, London (Appendices 1-4) before the start of each experimental study. Following the onset of the COVID-19 pandemic, all previously approved ethical approvals were updated in accordance with public health and University guidelines. All procedures were conducted in compliance with the Declaration of Helsinki (2013). The experimental sessions and outcome measurements took place at St Mary's University in a well-ventilated laboratory maintained at a temperature of 18-22 °C or at a private physiotherapy clinic. Throughout the experimental testing, strict measures were undertaken to ensure the cleanliness and safety of the laboratories and equipment used by human participants. Contaminated disposables were promptly disposed of in the appropriate disposal canister after use.

3.2 Participant recruitment

Participants for the experimental studies (Chapter 4, 5 and 6) were recruited from the student and staff population at St Mary's University through word of mouth. For Chapter 7, knee OA patients were recruited from private general practitioners (GP's) and consultants, who provided interested participants with study information and contacted the lead researcher. Two case studies involving clinical populations after surgery were also recruited in Chapter 6 using similar methods. A priori power calculations were conducted using G*Power software to determine the required number of participants for each experimental study.

Before the commencement of any experimental procedures, participants received both written and verbal explanations of the procedures, associated risks and benefits and the level of commitment required for each study. They completed a health and physical activity questionnaire. Inclusion criteria included the absence of cardiovascular or metabolic diseases, non-smoker status, resting systolic blood pressure below 140 mmHg and an ankle brachial index score between 0.9-1.4 to ensure stability for NMES and BFR exercise. Participants provided informed consent by signing a consent form, and they were informed that they could withdraw from the study at any time without having to provide a reason. They were also assured that their data would be coded to maintain anonymity, protected with passwords and treated confidentially.

Prior to data collection, two familiarisation sessions were conducted to ensure participant's comfort with the laboratory environment, equipment, protocols used, optimising maximal NMES current tolerated and while minimising potential learning effects. Participants were instructed to arrive at the laboratory well-hydrated and rested, refraining from strenuous exercises and alcohol within the previous 24 h, avoiding caffeine on the same day and avoiding food three hours before each test.

Descriptive data and anthropometric measurements, such as age, height, body mass and activity level, were recorded before the start of each experimental study. Height was measured to the nearest 0.01 m using a Harpenden Stadiometer (Holtain Ltd., Crymych, UK), while body mass was measured to the nearest 0.1 kg using calibrated laboratory scales (Seca GmbH & Co., Hamburg, Germany). These measurements were taken with participants standing without shoes and wearing their experiment clothing.

3.3 Ankle-brachial index

The ankle-brachial index (ABI) is a non-invasive diagnostic test used to assess blood flow to the legs and feet, serving as a screening tool for peripheral arterial disease and other vascular conditions[194]. Every participant's ABI was measured following published guidelines[194]. The measurement involved using a standard blood pressure cuff and a handheld Doppler probe (Hi-Dop, Ana Wiz Ltd, Surbiton, London, UK) to determine the systolic blood pressure (SBP) in the brachial artery (arm) and the posterior tibial artery (leg).

To measure the ABI with the Doppler method, participants were instructed to rest in a supine position for 5 minutes with their head and heels supported, at a comfortable room temperature. Cuffs were not placed on areas with a distal bypass or ulcer, and any open lesions were covered with an impermeable dressing to prevent contamination. Participants were asked to remain still during the measurement. The cuff was wrapped around the ankle with the straight wrapping method, as with brachial measurement, and the lower edge positioned 2 cm above the superior aspect of the medial malleolus. Gel was applied to the 8 MHz Doppler probe, which was then placed at a 45-60° angle to the skin surface in the pulse area. The probe was adjusted to obtain the clearest signal.

To detect pressure, the cuff was gradually inflated until the flow signal disappeared, and then slowly deflated to identify signal reappearance. If flow was still detected at the maximum inflation level (300 mmHg), the cuff was immediately deflated to prevent discomfort. Doppler was also used to detect brachial blood flow during arm pressure measurement. The same sequence of limb pressure measurement was followed for all participants. If the first arm measurement was 10 mmHg or more different than the other arm, the measurement was repeated at the end of the sequence, and the two numbers were averaged. However, if the difference between the two measurements exceeded 10 mmHg, only the second measurement was used for data collection. The ABI for each participant was calculated by dividing the posterior tibial blood pressure by the arm systolic blood pressure[194].

3.3.1 Measurement reliability

Measurement reliability is of the utmost importance in research projects as it ensures consistent and accurate results. Reliability refers to the extent to which a measure produces consistent results over time and across different observers[195]. Unreliable measures can lead to inaccurate results, hindering the ability to draw valid conclusions and replicate studies for result validation. Test-retest reliability assesses the consistency of an outcome measure over time by administering the same measure to the same group of participants at different time points[195]. This reliability type helps determine if changes in outcome measures are due to actual participant change or measurement error. It is particularly crucial in evaluating intervention effectiveness, ensuring sensitivity changes and produce consistent and valid results[196].

The measurements in this PhD research were conducted by the same researcher (PH). Test-retest (intra-session) reliability of the measures was assessed using the coefficient of variation (CV) and minimum detectable change (MDC). The CV is a statistical measure that indicates the variability of a measurement as a percentage of its mean and is used as an indicator of the degree of variability in the data, relative to the mean. A lower percentage reflects greater reliability. CV was calculated by dividing the standard deviation by the mean and multiplying by 100:

CV (%) = (standard deviation / mean) x 100

The MDC determines the smallest change observed in a measurement with a given level of confidence. Changes exceeding the MDC indicate true change. MDC for measurements was calculated using the standard error of the mean and the 95% confidence interval (CI):

MDC = 1.96 x $\sqrt{2}$ x standard error of the mean

The test-retest reliability of measurements used in this PhD research is presented in this Chapter.

The test–retest (intra-session) reliability of ABI was assessed by conducting three sessions spaced 7 days apart on a sample of 20 adults. The results showed a CV of 0.9% and an MDC of 0.02.

3.4 Neuromuscular electrical stimulation

NMES is a therapeutic technique that utilises electrical impulses to stimulate nerves and muscles, inducing muscle contraction without voluntary effort (Chapter 2.1-2.5)[197]. In Chapter 4, participants underwent 8 minutes and 10 s of NMES at a fixed knee joint angle of 90° while seated on a custom-made strength chair equipped with a strain gauge. This duration was based on the minimal effective training dose in NMES+BFR studies conducted by Gorgey et al. [45] in spinal cord injured patients. Starting from Chapter 5, participants remained seated at a fixed knee joint angle of 90° and received 20 minutes of NMES divided into four sets of 5 minutes each. The duration was modified from Chapter 4 for the subsequent studies based on a NMES+BFR training study by Andrade et al. [47], which reported no effect on muscular strength or hypertrophy in healthy adults with NMES stimulus durations of less than 10 minutes. Subsequent studies adopted the stimulation duration of 4 x 5 minutes employed by Natsume et al. [46], who observed muscular strength and hypertrophy improvements after 2-weeks of NMES+BFR in healthy adults. The stimulation parameters used across all studies included a frequency of 50 Hz, duty cycle of 5 s of stimulation followed by a 5 s pause, pulse width of 400 µs, and intensity set at the

maximum tolerated for each participant [38, 45, 46, 198]. The parameters were chosen based on previous research indicating their optimal effectiveness in promoting quadriceps muscle adaptations[38,60,123,147,198]. During all experimental studies, the quadriceps muscles were stimulated using three self-adhesive electrodes (Axion Medical, Axion GMBH, Villengen-Schwennigen, Germany) (2 mm thick) connected to a portable battery-powered neuromuscular electrical stimulator (Mi-Theta 600; Cefar Compex; Medicompex, Ecublens, Switzerland). The negative electrode (10 x 5 cm) was positioned approximately 13.2-13.4 cm below the inquinal crease (equal to the BFR cuff width), while the two positive electrodes (5 x 5 cm) were placed over the motor points of the vastus medialis (VM) and VL muscles. Muscle motor points were identified by gradually increasing the stimulatory current using a pen electrode (Compex; Medicompex, Ecublens, Switzerland) until a clear muscle twitch was observed[147]. The pen electrode was moved slowly over the skin, with the stimulatory current gradually increased until a clear muscle twitch was observed[38]. The electrode was placed at the point that elicited the largest visible twitch, and its location was recorded. marked, and consistently applied throughout each session. Participants were instructed to relax their thigh muscles during the stimulation period.

3.5 Determination of blood flow restriction pressure

BFR is a modality that involves partially restricting blood flow to target muscle groups using an inflatable cuff[117]. Several variables, such as cuff width, participant anthropometrics and muscle size, can influence the specific BFR pressure applied[57]. Therefore, the most reliable way to standardise BFR pressure regardless of participant characteristics and cuff width is by measuring each individual's AOP[57].

In Chapter 4, AOP was determined using a handheld Doppler probe (8 Hz) placed 2 cm proximal to the end of the medial malleolus, over the posterior tibial artery[151,152]. A pneumatic cuff (PTS tourniquet system, Delfi Medical Innovations, Vancouver, Canada)

with a width of 13.4 cm and a length of 58 cm was placed around the most proximal portion of each participant's dominant thigh. The cuff pressure was increased in stepwise increments by the pneumatic system connected to the tourniquet cuff. AOP was determined when no auscultatory pulse was detected by the Doppler probe[153]. BFR pressures used during the experimental conditions were 0%, 40%, 60% and 80% of AOP in a resting condition (seated with knees flexed to 90°), matching the body position during the intervention[116].

The test–retest (intra-session) reliability of measuring AOP via handheld Doppler by conducting three sessions spaced 7 days apart on a sample of 10 adults was CV of 2.1%, indicating good reliability, and an MDC of 7.2 mmHg.

For Chapter 5, 6 and 7, an automatic AOP determining pneumatic cuff (PTS ii tourniquet system, Delfi medical innovations, Vancouver, Canada) with the same dimensions as before was placed around the most proximal portion of each participant's dominant thigh. The pneumatic system connected to the tourniquet cuff increased the cuff pressure in stepwise increments, and AOP was automatically detected without the need for handheld Doppler measurement[153]. This device has been determined to have clinically acceptable accuracy and high reliability by our research group[153]. In each experimental study, AOP was measured for each individual in the position where the interventions took place (i.e., seated with the knee at 90°)[116]. The BFR pressure was maintained throughout the protocol of Chapter 4 and during each 5 minutes NMES set, released during the 1-minute rest periods, and immediately upon completion of the fourth 5 minutes set for the protocol of Chapter 5 and onwards[46].

3.6 Maximal voluntary isometric contraction

Maximum voluntary isometric contraction (MVIC) is a standardised method of measuring muscle strength by assessing the maximum force produced during an isometric contraction[199]. In Chapter 4 and 5 (MVIC, 5 minute all-out test), knee extension MVIC was measured using a custom-made strength chair and a digital strain gauge (Interface SSM-AJ-500 Force Transducer, Interface, Scottsdale, USA). Peak torque production during both voluntary and evoked NMES contractions was evaluated. Prior to testing, the strain gauge was calibrated with a known mass to convert voltage to Newtons. Participants were seated with the backrest at an 80° angle, and straps were used to stabilise the torso and hips to prevent unwanted movement. The load cell was fixed at a 90° angle of knee flexion (goniometer), and the resistance pad was secured 2 cm above the lateral malleolus. The chair set-up was recorded and standardised for each session. However, the custom-made strength chair design did not allow for a 60-75° knee angle, which is optimal for assessing quadriceps peak torque during voluntary and evoked contractions[199,200].

The MVIC protocol consisted of a warm-up phase followed by maximal contractions. The warm-up phase included three sets of 5 s submaximal contractions at 25%, 50% and 75% of each participant's voluntary maximal effort, with 30 s of rest between repetitions. Participants were instructed to exert maximum force as quickly as possible and peak torques were defined as the highest MVIC value observed during the peak 0.5 s of each contraction, multiplied by shank length and presented in newton metres (Nm)[199,201]. Verbal encouragement was provided throughout the testing. Initially, three contractions were performed, and if two measurements differed by more than 5%, an additional contraction was performed[201]. Post-intervention MVIC measurements were conducted 60 s after the NMES intervention and cuff deflation throughout Chapter 4. All raw MVIC signals were low-pass filtered using a zero-lag fourth-order Butterworth filter with an 11 Hz cut-off frequency determined from a residual analysis. In Chapter 4, evoked force during each NMES contraction was also recorded.

The test–retest (intra-session) reliability of MVIC measurements in the strength chair was evaluated across three sessions spaced 7-days apart with 20 adult participants. The CV was 3.8% and the MDC was 9.6 Nm.

3.7 Isokinetic Dynamometry

Isokinetic dynamometers are devices that resist applied forces and control exercise speed at a predetermined rate. These dynamometers provide a comprehensive record of applied force during various muscle actions, including isometric, concentric and eccentric actions[199]. In Chapter 5, maximal isometric, eccentric and concentric strength of the quadriceps muscles were measured using an isokinetic dynamometer (Cybex, Humac Norm, Ronkonkoma, USA) to assess peak force measurements. The dynamometer, equipped with a digital strain gauge, displayed force measurements to the nearest 0.1 N. Prior to each measurement, the instrument was calibrated following the manufacturer's instructions and specifications. Participants were comfortably seated with the backrest angled at 80° and without shoes or orthotic devices.

MVIC for the quadriceps was tested at 30°, 45°, 60°, 90° and 105° of knee flexion to determine peak torque angle adaptations resulting from interventions at 90° knee flexion[199,201,202]. A shin pad positioned 2 cm above the lateral malleolus of the fibula was attached to a load cell[199]. Straps were used to secure the chest and hips in place. The setup for each participant was standardised and recorded during the familiarisation session and repeated for subsequent sessions. Participants performed three warm-up trials at 25%, 50% and 75% of their perceived maximal effort[201,202]. They were instructed to push as hard as possible against an immovable strain gauge pad attached to a force transducer for 5 s. Three contractions were performed, and the highest isometric torque

achieved was recorded for data analysis. Participants had 30 s rest periods between attempts[199,201].

Concentric and eccentric peak torque were measured at an angular velocity of 60° per s between 0° and 90° of knee flexion[199]. The lateral femoral condyle was aligned with the dynamometer's axis of rotation. Gravity compensation and limb weighing were performed according to the manufacturer's instructions. Participants performed a standardised warmup and familiarisation set of ten repetitions at approximately 50% of their maximum effort for five repetitions[201]. After 1 minute of rest, they performed five maximal concentric and eccentric knee extensor contractions at 60°/s and the highest peak torque used for data analysis.

The test–retest (intra-session) reliability of isometric measurements was evaluated across three sessions spaced 7 days apart with 10 adult participants. The CV was 3.4% and an MDC of 8.3 Nm MDC. For isotonic contractions performed on the Cybex, the CV was 4.8% with an MDC of 10.2 Nm.

3.8 Quadriceps muscle thickness

Muscle cell swelling has been proposed as a acute observation after BFR interventions, suggesting that it can inhibit and promote anabolism by altering the protein balance[203]. Previous BFR studies have assessed changes in muscle cell swelling by measuring muscle thickness[203,204]. In Chapter 4, quadriceps muscle thickness (MTH) was measured using B-mode ultrasonography (Echoblaster 128 EXT-1Z, Telemed, Lithuania; 60 mm linear scanning probe, 7 MHz transducer scanner). To determine the measurement sites for the vastus medialis (VM) and VL muscles, an anthropometric tape measure was placed along the length of the thigh from the superior tip of the patella to the anterior superior iliac spine. Using the superior pole of the patella as the reference point, the thickness of VM was

measured at 20% of this distance and VL at 50% of this distance. The VM measurements were taken from 12.5% of thigh circumference in the medial direction from the midpoint of the thigh, and the VL measurements were taken from 10% of thigh circumference in the lateral direction. These locations represent the location of the maximum cross-sectional area of the VM and VL muscles[8]. The ultrasound probe was placed separately over the VM and VL musculature for three trials. Before each scan, participants rested in a supine position for 5 minutes. The measurement sites were marked using indelible ink. With the leg fully extended, the deep and superficial aponeurosis of each muscle was identified, and the distance between the two interfaces was calculated as MTH. The mean MTH from three measurements from the centre of each image was used for data analysis[46].

The test–retest (intra-session) reliability of VM and VL MTH measurements were evaluated across three sessions spaced more than 7-days apart with 20 adult participants. The CV for VM MTH was 3.2% with an MDC of 0.6 mm, while the CV for VL MTH was 5.2% with an MDC of 0.6 mm.

3.9 Quadriceps cross-sectional area

Muscle CSA is commonly used as an indicator of changes in skeletal muscle size, such as hypertrophy and atrophy[205,206]. Recently, mode-B ultrasonography (US) has emerged as a cost-effective alternative to MRI for acquiring high-quality muscle CSA images[206]. Ultrasonography allows for accurate differentiation between skeletal muscle, connective tissue and intra- and extra-muscular fat. Lixandrao et al. [206] demonstrated good validity of US-acquired CSA measurements compared to MRI measurements (CV = 1.75%). For Chapter 5, the muscle volume of the VL was measured using B-mode ultrasound scanner (Echoblaster 128 EXT-1Z, Telemed, Lithuania; 60mm linear scanning probe, 7 MHz transducer scanner), following the protocol described by Lixandrao et al. [206]. Participants initially assumed a supine position on a plinth, and the location of the lateral epicondyle of the femur was identified. A tape measure was placed from the lateral epicondyle to the
greater trochanter of the femur, with marks made at 50% and 75% of this length[206]. Semipermanent ink was used to mark these points as references. Sequential transverse marks were made on the skin every 2 cm from the reference points, extending towards the medial and lateral aspects of the thigh, to guide the displacement of the US probe. The B-mode linear array probe was aligned with the transverse ink markers on the thigh. Sequential still images were obtained by aligning the superior edge of the probe with each mark on the skin, moving in a middle-to-lateral direction. The power and gain settings were adjusted to optimise image quality and images were recorded for each participant. The scanning head was coated with a water-soluble transmission gel to ensure acoustic contact without depressing the skin surface. Following data collection, the images of the VL muscle were reconstructed using PowerPoint (Microsoft, Redmond, WA, USA) according to the procedures described by Reeves et al. [205] and Lixandrao et al. [206]. Each image was manually rotated until the entire fascia of the VL muscle was reconstructed. The muscle CSA was measured using computerised planimetry. This involved contouring the VL muscle CSA along the muscle fascia using an 800-dpi mouse (Madena 3.2.5; EyePhysics, Los Alamitos, CA, USA) and Image J software (ImageJ, U.S. National Institutes of Health, Bethesda, Maryland, USA). The interfaces between the subcutaneous adipose tissuemuscle and the muscle-bone were identified from the US images and used for data analysis[206]. Image J was calibrated using fixed distance scales displayed in the US images. Each data point for statistical analysis was represented by the average of two pictures.

The test–retest (intra-session) reliability of VL CSA measurements was evaluated across three sessions spaced more than 7 days apart with 10 adult participants. The CV for VL CSA was 5.7% with an MDC of 1.2 cm².

73



Fig 3.1. Example of vastus lateralis cross-sectional area measurement.

3.10 Quadriceps muscle architecture

Muscle fascicle length and pennation angle play crucial roles in determining muscle force production[207]. Skeletal muscle architecture was evaluated using US imaging (Echoblaster 128 EXT-1Z, Telemed, Lithuania; 60mm linear scanning probe, 7 MHz transducer scanner). Measurements of fascicle length and pennation angle were obtained between the greater trochanter and lateral epicondyle of the femur in the sagittal plane. To ensure consistent anatomical location for subsequent measurements, the US probe was positioned in the sagittal plane, perpendicular to the mediolateral axis, and indelible ink was used as reference points. Multiple images were taken and aligned to visualise the whole fascicle. In each resting US image, the fascicular path was identified as the space between echoes originating from the perimysial tissue surrounding the fascicle[208]. Fascicle length and pennation angle were measured using Image J software (ImageJ, U.S. National Institutes of Health, Bethesda, Maryland, USA). with the participant lying supine and the knee joint fully extended. The average of two measurements for fascicle length and pennation angle was used for data analysis[208].

The test-retest (intra-session) reliability of VL fascicle length and pennation angle measurements were assessed across three sessions spread more than 7 days apart with 10 adult participants. The CV for VL fascicle length was 2.9%, with an MDC of 0.04 cm. For VL pennation angle, the CV was 4.8%, with an MDC of 1.2°.



Fig 3.2. Example of vastus lateralis muscle thickness, fascicle length and pennation angle.

3.11 Blood pressure

Blood pressure serves as an important indicator of cardiovascular health[209], and in Chapter 4 and 8, it was measured both before and after each intervention to evaluate the response under various conditions and different populations. SBP and diastolic blood pressure (DBP) were measured using an automatic blood pressure monitor (Omron M3-IT, Omron Healthcare UK Ltd, Milton Keynes, UK). Participants were seated in a chair with a backrest and their feet on the floor, in a relaxed state without speaking, for at least 5 minutes. The participant's right arm was positioned on an armrest, supported at the level of the heart, while ensuring no tight clothing constricted the arm. The cuff was placed 2 cm above the brachial artery, aligned with the 'artery mark'. The bladder of the cuff encircled at least 80% of the arm but not more than 100%[210]. The appropriate cuff size was determined following the manufacturer's recommendations. The measurement process involved automatic inflation and deflation of the cuff to determine systolic and diastolic blood pressure readings. Two measurements were taken and assessed, with a third measurement taken if variability exceeded 5 mmHg, and the mean value was recorded[210].

The test-retest (intra-session) reliability of SBP and DBP was evaluated across three sessions spaced 7 days apart, involving 20 adult participants. The CV for SBP was 3.3% CV, with an MDC of 2.5 beats/minute. For DBP, the CV was 5.1%, with an MDC of 3.2 beats/minute.

3.12 Heart rate

Measuring heart rate (HR) responses provides valuable insights into heart function[211]. In Chapter 4 HR was measured before, during and after every condition to evaluate safety[211]. In Chapter 7, HR was measured before and immediately post every intervention to assess the cardiovascular effects in knee OA patients. HR was assessed using a heart rate monitor with a coded transmitter and chest strap placed underneath each participant's xyphoid process (Polar TY1, Polar, Kempele, Finland). Participants were instructed to maintain calm breathing and avoid looking at the polar monitor. HR was recorded as the average of 5 beats and measured after 5 minutes of seated rest, both before and after the experimental conditions, as well as after each set (10 repetitions) of the NMES protocol in Chapter 4.

The test–retest (intra-session) reliability of HR was evaluated across three sessions spaced 7 days apart, involving 20 adult participants at rest. The CV for HR was 5.2%, with an MDC of 3 beats/minute.

3.13 Wellness

Prior to each training session in Chapter 5, a psychometric questionnaire was utilised to assess participant wellness (Appendix 5.4)[212]. This assessment aimed to determine if participant wellness had any influence on the tolerable currents achieved during each training session. The questionnaire consisted of five questions related to perceived fatigue, sleep quality, general muscle soreness, stress levels and mood. Participants rated each question scored on a 5-point scale, with 1 representing poor wellness and 5 representing very good wellness. The overall wellness score was calculated by summing the scores of the five questions[212]. Reliability of this questionnaire has been previously established as intraclass correlation coefficient (ICC) = 0.78[213]. The ICC is a statistic that quantifies the degree of agreement or consistency among multiple measurements, with values ranging from 0 (no agreement) to 1 (perfect agreement), interpreted as poor (ICC < 0.5), moderate ($0.5 \le ICC < 0.75$), good ($0.75 \le ICC < 0.9$), or excellent (ICC ≥ 0.9) reliability or agreement[214].

3.14 Rating of perceived exertion

The RPE scale, developed by Gunnar Borg [215], was used to assess exertion and breathlessness during activities (Appendix 5.1). Previous NMES+BFR research reported RPE scores to be higher compared to NMES alone[46]. In the context of NMES combined with varying degrees of BFR, perceptual responses were evaluated during Chapter 4, 5, 7 and 8. Participants were asked to rate their exertion on the scale, considering physical stress and fatigue rather than factors like leg pain or breathlessness. The chosen number on the scale corresponded to the intensity of the activity. familiarisation with the scale was conducted prior to data collection, and the standard Borg 6-20 scale was used[215]. RPE intra-session reliability has been previously established as ICC = 0.75-0.82[216].

The Numeric Pain Rating Scale (NPRS) is a unidimensional measure used to assess pain intensity in adults[217]. Previous research on NMES+BFR reported higher session pain scores compared to NMES alone[46]. Therefore, the perceptual responses of NMES combined with varying degrees of BFR were evaluated in Chapter 4, 5, 7 and 8. The NPRS is a segmented numeric version of the visual analogue scale, where respondents select a number from 0-10 to indicate the intensity of their pain. A rating of "0" represents no pain, while "10" represents the worst pain imaginable[218]. Participants confirmed their understanding of how to rate pain before testing. NPRS reliability has been previously established as ICC = 0.91[219].

3.16 End-test torque and impulse above end-test torque

The concept of critical power measures the highest sustainable power output that an individual can maintain without reaching exhaustion[220]. In clinical populations, such as those recovering from OA or lower limb surgery, endurance capacity is often compromised, making it challenging to engage in prolonged exercise[20,221,222]. Understanding an individual's critical power can assist healthcare professionals in designing tailored exercise programs to gradually increase endurance capacity over time[220,223]. The measurement and validation of critical power have been recently performed on the quadriceps during a 5 minute all-out test and referred to as end-test torque[224]. The test to determine end-test torque and impulse above end-test torque of the knee extensors involved 60 MVIC's of 3-5 s each, with 2 s of rest between contractions[224,225]. During the test, the muscle stimulator was triggered on the 1st, 2nd, 59th and 60th contraction to the femoral nerve to calculate voluntary activation (Section 3.17). Stimuli were delivered at the visually plateaued peak torque for each individual, and 30 s prior to the first contraction. The first and final two contractions lasted 5 s, while the remaining 56 contractions lasted 3 s, based on pilot testing that demonstrated higher reliability in voluntary activation measurements using 5 s MVIC's

compared to 3 s MVIC's. Participants were provided with feedback on their previous MVICs and were instructed to exert maximal effort during each contraction, despite the expected decline in torque by more than 50% during the test. Participants were strongly encouraged to maximise torque without being informed of the elapsed time or remaining contractions. A green light displayed on a computer screen (iPad, generation 6, Apple Inc, 1 Apple Park, Cupertino, California, USA) signalled the subjects to contract, while a red light signalled them to stop. Verbal instructions to "push" and "stop" were also provided due to participant's tendency to focus on the exercising leg rather than the screen. The test concluded upon completion of the 60th contraction cycle. The mean torque value of each contraction was used to calculate both critical torque and impulse above end-test torque. End-test torque above end-test torque values x time curve and above the end-test torque value for each trial[224,225] (Fig 3.3). The reliability of the 5 minute all-out test has been previously established as CV 4.2-8.2%[226].



250

Fig 3.3. Example of impulse above end-test torque and end-test torque.

3.17 Voluntary activation

Measuring voluntary activation provides insights into the neural drive, inhibition, and central fatigue, which directly influence an individual's ability to generate force[227,228]. The percentage of voluntary activation was estimated using the twitch interpolation protocol[229]. Singlet stimuli were applied to the femoral nerve at approximately 200–300 ms intervals during the 1st, 2nd, 59th and 60th contractions of the 5 minute all-out test, as well as 5 s before the first contraction. Rectangular pulses of 200 µs duration were delivered using a high-voltage constant-current stimulator (Digitimer DS7AH, Hertfordshire, UK) with a maximum voltage of 400 V[229]. The cathode electrode (20 mm diameter, Durastick Supreme, Chattanooga Group, Hicton, TN) was positioned over the femoral nerve in the lateral portion of the femoral triangle, while the anode electrode (20 mm, Durastick Supreme, Chattanooga Group, Hicton, TN) was placed over the gluteus medius muscle, medial to the greater trochanter of the femur[225]. To determine the optimal probe location, a series of single stimuli (30 mA) were delivered, and the probe position resulting in the greatest M-wave response was identified and marked. The intensity of supramaximal stimulations was determined by progressively increasing the current until maximal twitch was observed, indicating Mmax, the point at which further increases in stimulus intensity did not yield a higher twitch amplitude. An additional 50% was added to the highest current to ensure a supramaximal stimulus [229]. MVIC was defined as the highest torgue achieved prior to the superimposed single stimulation [230]. Superimposed torque referred to the increment in torque induced by the stimulation during voluntary contraction. The resting twitch was determined as the peak torque value induced by the single stimulation before the first voluntary contraction of the 5 minute all-out test. VA% was calculated using the following formula: VA% = [1 - (superimposed twitch amplitude/resting twitch amplitude)] ×100 (Fig 3.4). The reliability of this measure has been previously established as CV = 2.5%. ICC = 0.86[231,232].



Fig 3.4. Example of voluntary activation trace.

3.18 Near-infrared spectroscopy

Measuring muscle oxygenation using near-infrared spectroscopy (NIRS) provides valuable insight into the metabolic and physiological responses of muscle tissue[233]. Previous studies have demonstrated that BFR alone can enhance muscle oxygenation after a 7-day intervention[234]. By assessing changes in oxygen utilisation and delivery in the working muscles, NIRS can help evaluate the effectiveness of NMES+BFR[233]. Prior to the 5 minute all-out test, a NIRS optode (Oxymon, Artinis Medical Systems, Zetten, The Netherlands) emitting light at 780 and 850 nm, was regularly calibrated by Artinis Medical Systems. The NIRS optode was placed on the vastus lateralis and secured with an elastic bandage (Tiger Tear, Hampshire, United Kingdom). NIRS has increasingly been used to measure real-time levels of oxygenated haemoglobin/myoglobin in small vessels and skeletal muscle, as well as muscle oxygen consumption (mVO2)[234,235]. The optode was held within a black rubber housing that maintained a constant optode spacing of 4 cm. To minimise light loss and outside light interference, the opaque housing was taped to the skin and covered with black material. The designated area was carefully shaven and cleaned with alcohol wipes prior to the experiment. Pen marks were used to indicate the correct position of the probe holder for subsequent data collections, and additional marks were made on the skin to check for any movement of the rubber housing during testing. No movement was observed at the end of the measurements in any participant. The emitting and detecting optodes of the NIRS unit were placed distal and lateral to the greater trochanter, approximately 75% along the length of the femur. This placement on the VL was determined as the area where the greatest muscle oxygenation occurs during isometric contractions, compared to proximal locations[236,237]. The measured NIRS data analysed focused on deoxyhemoglobin (HHb) signal due to its insensitivity to changes in blood flow[238] (expressed in micromoles per litre per centimetre), as well as the derived changes in tHb (expressed in micromoles per litre per centimetre)[239,240]. The rate of change in HHb during each MVIC above 50% intensity is recognised as representative of mVO2[241,242]. With work intensity above 50% MVIC, sufficient intramuscular pressure is generated to occlude local muscle blood flow, enabling calculation of mVO2 without arterial occlusion[241,242]. The 5 minute all-out test consisted of 60, 100% MVIC's. mVO2 was determined by analysing the rate of change (slope regression) in [HHb] during the middle two seconds of each contraction period in the 5 minute all-out test when tHb was relatively constant. Specifically, the SLOPE function in excel was utilised to calculate the regression slope during the middle two seconds of every three second contraction which was manually identified. HHb resting values were obtained during the last 5 minutes of a 10 minutes rest period, during which individuals remained still and as relaxed as possible before the start of the exercise[243]. Values were determined as the difference between the HHb values obtained during the concentric phase of each repetition (as described above) and the mean HHb value during the 5 minutes resting period[244], and were used for further analysis. To minimise differences in initial values between trials, all measured parameters were normalised to the resting values acquired before each experiment, and reported as delta (Δ) values. The analysis provided a quantitative assessment of the relationships within the NIRS dataset, with a focus on changes in HHb [Δ HHb] values from baseline, during the middle 2 seconds of contraction 3-58 of the 5 minute all-out test. NIRS data were collected at a frequency of 10 Hz. Adipose tissue thickness at the site of measurement was assessed using B-mode US and within manufacturers guidelines. The first two contractions and final two contractions were not used for data analysis during the 5 minute all-out test, due to them being 5 s in length and also being influenced by muscle stimulation[239]. The reliability of HHb signal of the VL during exercise has been previously established as CV = 2.8-3.5%, ICC = 0.40-0.81[245].



Fig 3.5. Example of deoxyhemoglobin signal during each contraction of the 5 minute all-out test.

3.19 Pressure pain thresholds

Pressure pain thresholds (PPT) quantified the pressure required to elicit a painful sensation in a specific body area[246]. PPT measurements serve to evaluate pain sensitivity and assess the effectiveness of pain treatments[246,247]. Recent studies indicate that combining BFR with resistance and aerobic exercise, as well as NMES alone, positively affects PPT[135,171,178,248]. However, the impact of NMES+BFR on PPT remains unexplored.

PPT measurements were conducted with the participants seated, with both arms resting on their thighs[43,175]. A handheld pressure algometer (J Tech Medical, United States) with a 1cm² stimulation area and an approximate pressure increase rate of 1 kgf/s was used for assessment[174]. Our research group has previously validated the use of a handheld algometer for PPT assessment[43]. The PPT assessment sites were marked at the midpoint of the dominant and non-dominant quadriceps muscles (20 cm proximal to the base of the patella)[43,175]. Participants were instructed to verbally indicate "now" when they first perceived the applied pressure as painful, and the PPT was recorded as kgf/cm² at that

point. Two PPT assessments were performed at each site, with a 20 s interval between assessments, and the average value was used for analysis[43,175].

Test-retest (intra-session) reliability for PPT was assessed in 10 adults across three sessions spaced more than 7 days apart. The CV for PPT was 3.3% and the MDC was 0.4 kgf/cm².

3.20 Cold pain thresholds

Cold pain thresholds (CPT) are commonly used to evaluate an individual's perceived sensitivity to cold-induced pain[249]. Studies have reported increased CPT in knee OA patients[250], but the effects of NMES+BFR on CPT have not been assessed.

All CPT measurements were performed with participants seated, with both arms resting on their thighs[43,175]. A plastic syringe filled with frozen water and ice was used, with the ice pointing out of the syringe's end. The ice was applied to the participant's skin for 30 s, and they were asked to rate their sensation of cold pain from 0-10, with 0 representing no cold pain and 10 representing severe cold pain[249]. CPT assessment sites were marked on the inferior pole of the patella on the patellar tendons of the dominant and non-dominant quadriceps[249] Two CPT assessments were conducted at each site, with a 20 s interval between assessments, and the average value was used for analysis[43,175,249].

Test-retest (intra-session) reliability for CPT was evaluated in 10 adults across three sessions spaced more than 7 days apart. The CV for CPT was 9.8%, and the MDC was 0.4.

3.21 Temporal summation of pain

Temporal summation of pain (TSP) is a measure that assesses the intensity of pain experienced by an individual over time[247]. It is commonly used in the literature to evaluate the sensitivity of the central nervous system to pain stimuli and has shown promise in diagnosing chronic pain conditions such as knee OA[251]. However, the effects of NMES+BFR on TSP have not been investigated. TSP assessment involves applying a series of identical painful stimuli to the same area of the body while recording the intensity of the pain experienced[247]. Both TSP and PPT measurements are recommended in the broader pain literature[247,251,252]. As a measure of TSP, the pin-prick evoked wind-up ratio was utilised. The wind-up ratio is calculated by dividing the pain intensity evoked by a single pin-prick stimulus[247].

In the assessment, a single pin-prick stimulus was initially applied (Single PinPrick stimulator 256 mN: MRC Systems GmbH, Heidelberg, Germany), and the participant rated its severity. Subsequently, a series of 10 pin-pricks were administered at a frequency of 1 Hz within a 1 cm² area of skin, and the participant rated the severity of the final stimulus. Ratings were provided using the same NPRS scale used for the pain measures described previously[253]. TSP for each participant and testing site was calculated by subtracting the rating of the first pin-prick stimulus from the rating of the final pin-prick stimulus[247,252]and used for analysis.

Test-retest (intra-session) reliability for TSP was assessed in 10 adults across three sessions spaced more than 7 days apart. The CV for TSP was 6.8%, and the MDC was 0.8.

3.22 5 Sit-to-stands

Measuring sit-to-stand performance is recommended for patients with knee OA as it provides valuable insights into their lower extremity function[254]. Sit-to-stand is a common 85

task of daily living that involves the activation of multiple lower extremity muscles, including the quadriceps[254]. Assessing sit-to-stand ability helps healthcare professionals evaluate the functional capacity of knee OA patients and monitor their progress, as lower sit-to-stand ability is associated with an increased risk of falls[255]. The reliability of the 5-repetition sit-to-stand test has been previously examined, and it was reported to be adequate, with ICC's ranging from 0.67 to 0.90[256]. The validity of the sit-to-stand test is supported by its correlation with knee extension strength and gait performance[256,257]. During the sit-to-stand test, participants were instructed to perform the task five times, standing up from and sitting down on a slightly padded, armless chair with a height of 43 cm, as quickly as possible. They were instructed to fold their arms across their chests and ensure full standing and firm contact when sitting. The timing began with the command "go" and stopped when the participants completed the fifth sit-down. Time taken to complete the five repetitions and any knee pain experienced, rated on a 0-10 visual analogue scale, were recorded for further analysis[258].

3.23 Knee bend in 30 seconds

In individuals with knee OA, the ability to flex the knee can be restricted due to joint space narrowing, leading to an impact on physical function[259]. This test aims to assess the maximum number of knee bending's performed on one leg within a 30 s duration, as well as the ability to rapidly transition between eccentric and concentric muscle contractions around the knee joint[260]. Participants were instructed to bend and straighten their knee through their pre- intervention range of motion, assessed by goniometer, as many times as possible. The number of repetitions conducted was used for analysis and previous research demonstrated high intra-rater reliability (ICC = 0.92) using the same method[261].

3.24 Knee Injury and Osteoarthritis Outcome Score questionnaire

The Knee Injury and Osteoarthritis Outcome Score (KOOS) is a knee-specific instrument developed to assess individuals' perception of their knee and associated functional issues (Appendix 5.5)[262]. The KOOS physical function subscale provides a standardised and validated measure of physical function[262] It evaluates both short-term and long-term consequences of knee injury and consists of 42 items divided into five subscales; Pain, other Symptoms, Function activities in daily living, Function in Sport and Recreation, and knee-related Quality of Life. The KOOS meets the fundamental criteria for outcome measures and can be used to evaluate the progress of knee injury and treatment outcomes[261]. The questionnaire was administered before and immediately after the different conditions in Chapter 7. A previous meta-analysis has demonstrated reliability ICC = 0.85-0.90 of this questionnaire[263].

Chapter 4

Optimisation of effective parameters of NMES combined with BFR: The effect of restriction pressure

Journal Sport Rehabilitation. 2020 July 31;30(3): 375-383.

4.1 Preface

The main components of this chapter were published under the details previously stated. The introduction, results, discussion, and conclusion have been edited to elaborate on the publication and fit in with the rest of this thesis and assist with linking this Chapter to the following sections.

4.2 Abstract

Context: NMES+BFR has been shown to improve muscular strength and size better than NMES alone. However, previous studies used varied methodologies not recommended by previous NMES or BFR research. Objective: The present study investigated the acute effects of NMES combined with varying degrees of BFR using research-recommended procedures to enhance understanding and the clinical applicability of this combination. **Design:** Randomised crossover. Setting: Physiology laboratory. Participants: A total of 20 healthy adults (age: 27.1 [4.0] years; height: 177.0 [7.9] cm; body mass: 77.1 [13.2] kg). Interventions: Six sessions separated by at least 7 days. The first 2 visits served as familiarisation, with the experimental conditions performed in the final 4 sessions: NMES alone, NMES 40% BFR, NMES 60% BFR, and NMES 80% BFR. Main Outcome Measures: Maximal voluntary isometric contraction, muscle thickness, blood pressure, heart rate, rating of perceived exertion, and pain were all recorded before and after each condition. Results: The NMES 80% BFR caused greater maximal voluntary isometric contraction decline than any other condition (-38.9 [22.3] Nm, p < 0.01). VM and VL muscle thickness acutely increased after all experimental conditions (p < 0.05). Pain and RPE were higher after NMES 80% BFR compared with all other experimental conditions (p < 0.05). No cardiovascular effects were observed between conditions. Conclusion: The NMES combined with 80% BFR caused greater acute force decrement than the other conditions. However, greater perceptual ratings of pain and ratings of perceived exertion were observed with NMES 80% BFR. These acute observations must be investigated during chronic interventions to corroborate any relationship to changes in muscle strength and size in clinical populations.

4.3 Introduction

NMES has been demonstrated to prevent disuse muscle atrophy[191], however, inconsistent evidence exists regarding its efficacy in enhancing muscle adaptations[37]. Prior to the commencement of this study in 2017, trials examining NMES+BFR in humans yielded varied results. Two studies reported increased muscle strength and hypertrophy compared to NMES and BFR alone in healthy and spinal cord injured adults[45,46], while others found no added benefit to muscle strength and size[47,58].

The literature review conducted in Chapter 2 revealed methodological inconsistencies in the existing NMES+BFR studies. These inconsistencies may have contributed to conflicting findings and hindered the understanding of the underlying physiological mechanisms that induce changes in muscle strength and hypertrophy. The NMES protocols utilised thus far have exhibited considerable variability, encompassing frequencies ranging from 20 to 100 Hz. with unclear reporting of other parameters stimulation such as intensities[38,45,46,58,264]. To optimise quadriceps strength after NMES, it is recommended to employ a frequency of 50 Hz, maximal tolerable intensities and proper electrode placement over muscle motor points[147]. However, these parameters have not been utilised in previous NMES+BFR studies focusing on the quadriceps[46,58]. Moreover, a majority of studies have employed arbitrary restrictive pressures to implement BFR[38,47,58,116,124] or based their occlusion pressure on SBP[45]. Neither of these approaches are effective for controlling the magnitude of BFR, with current recommendations suggesting that pressure should be prescribed via AOP[61]. Furthermore, AOP's ranging from 40% to 80% are recommended in the wider BFR literature[57], with neither AOP or these recommended pressures being previously investigated in NMES+BFR research.

The mechanisms by which NMES combined with BFR increases muscle strength and induces hypertrophy remain unknown. A recent study observed greater evoked force decline in a rat model following NMES+BFR, which correlated with increased muscle size compared to NMES alone[38]. Furthermore, resistance training with and without BFR, leading to higher levels of fatigue as evidenced by reduced force production, results in greater improvements in muscle strength and size[265,266]. These findings suggest that acute post-exercise decrements in force production could serve a surrogate marker for optimising training programmes. However, a direct comparison of the acute muscle responses to NMES in combination with varying levels of BFR has not yet been investigated.

The present study aimed to answer question 1 of this thesis and aims to standardise methodologies to gain a better understanding of how NMES alone and in combination with varying levels of BFR (40-80% AOP) acutely affect muscular, cardiovascular and perceptual variables. Established protocols for both interventions were utilised.

It is hypothesised that NMES combined with BFR would result in higher levels of muscular fatigue, muscle swelling and perceptual variables (e.g., pain and exertion) compared to NMES alone.

4.4 Method

4.4.1 Participants

Twenty recreationally active (3.1 [1.4] h/week), healthy males (n = 15) and females (n = 5) (age: 27.1 [4.0] years; height: 177.0 [7.9] cm; body mass: 77.1 [13.2] kg and body mass index: 25.1 [3.0] kg/m²) volunteered to participate in this study. The sample size was calculated using G*Power software and the effect sizes of previous research assessing the same outcomes[267]. Inclusion criteria were: (a) absence of lower-limb injury, (b) negative

answers in the physical activity and readiness questionnaire (PAR-Q) questionnaire (Appendix 5.3), (c) no personal history of cardiovascular or metabolic disease, (d) nonsmokers, (e) resting SBP < 140 mmHg and (f) normal range on the ABI test (0.9-1.4)[194]. Participants were instructed to maintain their usual level of physical activity throughout the study. All participants provided written informed consent and the study was approved by the University ethics sub-committee (Appendix 1: SMEC_2016-17_104) and conducted in accordance with the Declaration of Helsinki (2013).

4.4.2 Study design

The study followed a randomised crossover design, generated via online software (http://www.randomization.com). All testing was undertaken at the University's temperaturecontrolled laboratory (21-22°C). Participants were required to visit the laboratory on six occasions, separated by at least 7 days and at the same time of day (±1 h), to minimise the circadian effect. All participants were tested at least 2 h postprandial and were instructed to avoid caffeine and exercise prior to testing. The first two visits served as familiarisation sessions, with the experimental conditions performed in the final four sessions. During the first visit, height, weight, ABI, knee extension MVIC, VM and VL MTH, AOP and NMES maximal tolerable intensity were measured. During the second visit, MVIC, MTH, AOP and NMES maximal tolerable intensity were repeated. After the familiarisation sessions, with the same trained researcher performing all outcome measurements (Fig 4.1):

- 1) NMES and cuff not inflated (NMES alone)
- 2) NMES combined with 40% BFR (NMES+40% BFR)
- 3) NMES combined with 60% BFR (NMES+60% BFR)
- 4) NMES combined with 80% BFR (NMES+80% BFR)



Fig 4.1. Experimental protocol. All participants performed the same neuromuscular electrical stimulation (NMES) protocol under four different blood flow restriction (BFR) pressures (0, 40, 60 and 80%) Outcome measures; systolic blood pressure (SBP); diastolic blood pressure (DBP); heart rate (HR); vastus medialis (VM) and vastus lateralis (VL) muscle thickness (MTH); knee extension maximal voluntary isometric contraction (MVIC) was assessed before (pre) and after (post) each experimental condition. Outcome measures assessed after every 10 NMES repetitions included; rating of perceived exertion (RPE), pain and HR. See abbreviations throughout.

4.5 Procedures

4.5.1 ABI

See section 3.3.

4.5.2 NMES

See section 3.4. VM and VL maximal tolerable intensities equalled 67.1 ± 44.1 mA and 70.7 ± 44.7 mA, respectively. Evoked force (% MVIC), is force generated by a muscle in response to NMES[38] and was recorded during every contraction and used for analysis.

Table 4.1. NMES protocol								
Current	Bi-phasic rectangular pulse							
Frequency (Hz)	50							
Pulse width (µ)	400							
Duty cycle (s)	5/5							
Ramp up / ramp down (s)	1.5 / 0.5							
Repetitions	40							
Intensity (mA)	Max tolerable							

4.5.3 Determination of BFR pressure

See section 3.5.

4.5.4 MVIC

See section 3.6.

4.5.5 MTH

See section 3.8.

4.5.6 Blood pressure

See section 3.11.

4.5.7 HR

See section 3.12.

4.5.8 RPE

See section 3.14.

4.5.9 Pain

See section 3.15.

4.6 Statistical Analysis

A two-way repeated-measures analysis of variance (ANOVA) was used to determine the effects of condition (0%, 40%, 60% and 80% BFR) and time; MVIC, MTH, SBP, DBP, HR across two time points (pre and post), HR, RPE, Pain across four time points (set 1, set 2,

set 3, set 4). If the assumptions of ANOVA were violated, the Greenhouse–Geisser correction factor was applied.

Interactions and main effects were followed with appropriate *post-hoc* analyses and Bonferroni adjustments. Statistical significance was set at p < 0.05.

Statistics were computed using SPSS Statistics software package version 24.0 (SPSS, Chicago, USA). Data are presented as means (standard deviation [SD]) ± standard error of mean (SEM) unless otherwise stated.

4.7 Results

No differences were observed between baseline values across the four experimental conditions (p > 0.05). No adverse events occurred.

4.7.1 MVIC

There was a main effect of time ($F_{(1,19)} = 37.2$, p < 0.001), no condition effect (p > 0.05) and a condition × time interaction ($F_{(3,57)} = 10.6$, p < 0.001) for MVIC decline (Fig 4.2). Post-hoc pairwise Bonferroni comparisons confirmed greater MVIC decline after NMES+80% BFR compared with NMES alone (p < 0.001), NMES 40% BFR (p < 0.001) and NMES+60% BFR (p = 0.001) (Fig 4.2). All differences were above the 9.9 Nm MDC, error of measurement.



Fig 4.2. Knee extension MVIC pre-test to post-test change Δ ; values as mean ± SEM. Significant difference was set at p < 0.05; * = significant difference between pre-test and post-test; † = significantly greater change compared to all other experimental conditions.



Fig 4.3. Evoked NMES force (% MVIC) during every contraction; values as mean \pm SEM; * = significant decline from first repetition to the last: \dagger = significantly different to all other conditions *p* < 0.05.

There was a main effect of time ($F_{(1,19)} = 43.1$, p < 0.001; $F_{(1,19)} = 92.1$, p < 0.001) for VM MTH and VL MTH increase, respectively (Table 4.2). However, there was no condition effect or condition × time interaction observed (p > 0.05).

4.7.3 Blood pressure

A main effect of time ($F_{(1,19)} = 12.1$, p = 0.002) was observed for SBP. There was no condition effect or condition × time interaction (p > 0.05) shown for SBP. There were no effects observed on DBP (p > 0.05) (Table 4.2).

	NMES alone			N	MES +40%	BFR	I	NMES + 60%	% BFR	NMES + 80% BFR			
	Pre	Post	C [95% CI]	Pre	Post	C [95% CI]	Pre	Post	C [95% CI]	Pre	Post	C [95% CI]	
MVIC	239.8	231.5	-8.3 [-18.5;	240.3	224.1	-16.2 [-	240.4	225.4	-15.1 [-23.8; -	242.6	203.8	-38.9 (-49.3; -	
(Nm)	(51.3)	(57.1)	1.9]	(48.3)	(46.8)*	25.0; -7.3]	(52.3)	(55.7)*	6.4]	(55.1)	(52.1)*†	28.3]	
VM MTH	25.0	25.6	0.6 [0.3;	25.2	26.0	0.8 [0.3;	25.0	25.8	0.8 [0.4; 1.3]	24.7	25.9	1.2 [0.8; 1.5]	
(mm)	(2.7)	(2.6)*	0.9]	(2.9)	(2.8)*	1.2]	(2.9)	(2.9)*		(2.7)	(2.9)*		
VL MTH	17.2	17.9	0.7 [0.5;	16.6	17.7	1.0 [0.6;	16.9	18.0	1.1 [0.7; 1.6]	17.0	18.4	1.4 [0.9; 1.9]	
(mm)	(2.8)	(2.8)*	1.0]	(2.4)	(2.9)*	1.5]	(2.5)	(3.0)*		(2.9)	(3.2)*		
SBP	122.8	125.2	2.3 [0.7;	121.9	123.9	1.9 [-1.4;	123.4	124.7	1.4 [-0.6; 3.3]	123.0	125.5	2.5 [0.8; 4.1]	
(mmHg)	(8.7)	(9.2)*	4.0]	(8.5)	(7.8)	5.2]	(9.3)	(8.1)		(8.1)	(7.8)*		
DBP	69.4	71.1	1.7 [-0.9;	70.2	71.4	1.3 [-0.6;	71.2	71.2	0.1 [-2.2; 2.4]	70.7	71.6	0.9 [-1.8;3.5]	
(mmHg)	(6.7)	(5.3)	4.4]	(6.2)	(7.6)	3.1]	(7.1)	(6.3)		(6.0)	(6.5)		
HR	61.0	60.7	-0.3 [-2.2;	60.7	61.2	0.5 [-1.1;	60.6	58.3	-2.4 [-4.6; -	62.2	59.5	-2.7 [-6.5; 1.1]	
(bpm)	(9.3)	(9.6)	1.6]	(9.3)	(8.6)	2.1]	(8.8)	(9.5)*	0.2]	(9.1)	(9.7)		

Table 4.2. Knee extension MVIC, muscle thickness and cardiovascular pre-test and post-test measurement values; mean (SD) [95% CI].

Significant differences were set at p < 0.05; * = significant difference between pre-test and post-test; † = significantly greater change compared to all other experimental conditions. C = change from pre to post; bpm = beats/minute.

There was a main effect of time ($F_{(1.4,26.7)} = 54.8$, p < 0.001), condition effect ($F_{(3,57)} = 4.1$, p = 0.01) and condition × time interaction ($F_{(6.6,125.2)} = 3.9$, p = 0.001) for HR (Table 4.2 and 4.3). Post-hoc pairwise comparisons revealed after set 1, NMES alone was lower than NMES+80% BFR (p = 0.019); after set 2, NMES+80% BFR was higher than NMES alone (p = 0.019); after set 3, <u>NMES 60 and NMES 80</u> were higher than NMES alone (p = 0.026 and p = 0.01, respectively); after set 4, NMES+80% BFR was higher than NMES alone (p = 0.019) (Table 4.2 and 4.3). However, all differences were below the 3.2 bpm MDC, showing no meaningful change.

4.7.5 RPE

There was a main effect of time ($F_{(1.1,21.3)} = 11.9$, p = 0.002), condition effect ($F_{(3,57)} = 7.7$, p < 0.001) and condition x time interaction ($F_{(3.8,72.4)} = 3.4$, p = 0.015) for RPE (Table 4.3). Post-hoc pairwise comparisons confirmed RPE to be higher; after set 1 of NMES+80% BFR compared with NMES alone (p = 0.006), after set 2 of NMES+80% BFR compared with NMES alone, NMES+40% BFR and NMES+60% BFR (p = 0.018; p = 0.027; p = 0.005, respectively), after set 3 of NMES+80% BFR compared with NMES alone, NMES+40% BFR and NMES+60% BFR compared with NMES alone, NMES+40% BFR and NMES+60% BFR compared with NMES alone, NMES+40% BFR and NMES+60% BFR compared with NMES alone, NMES+40% BFR and NMES+60% BFR (p = 0.002; p = 0.002; p = 0.038, respectively). Finally, RPE was higher after set 4 of NMES+80% BFR compared with NMES alone, NMES+40% BFR and NMES+60% BFR (p = 0.001; p = 0.001; p = 0.041, respectively).

4.7.6 Pain

There was a main effect of time ($F_{(1.6,31.2)} = 13.6$, p < 0.001), condition effect ($F_{(3,57)} = 19.6$, p < 0.001) and condition × time interaction ($F_{(5.3,100.3)} = 4.8$, p < 0.001) for pain (Table 4.3). Post-hoc pairwise comparisons revealed ratings of pain were higher; after set 1 of NMES+80% BFR compared with NMES alone, NMES+40% BFR and NMES+60% BFR (p = 0.006; p = 0.001; p = 0.027, respectively), after set 2 of NMES+80% BFR compared with NMES alone, NMES+40% BFR and NMES+60% BFR (p < 0.001; p < 0.001; p = 0.010, respectively), after set 3 of NMES+80% BFR compared with NMES alone, NMES+40% BFR and NMES+60% BFR (p < 0.001; p < 0.001; p = 0.001, respectively). Finally, pain ratings were higher after set 4 of NMES+80% BFR compared with NMES alone, NMES+40% BFR and NMES+60% BFR (p < 0.001; p < 0.001; p = 0.003, respectively) and lower after set 4 of NMES alone compared with set 4 of NMES+60% BFR (p = 0.039).

		NMES	١	NMES+40% BFR				NMES+60% BFR				NMES+80% BFR				
	Set 1	Set 2	Set 3	Set 4	Set 1	Set 2	Set 3	Set 4	Set 1	Set 2	Set 3	Set 4	Set 1	Set 2	Set 3	Set 4
HR	71.1	71.9	71.8	72.2	74.2	74.6	75.1	74.5	73.6	76.4	77.0	76.5	77.1	79.3	79.4	78.8
(bpm)	(9.1)	(9.7)	(8.4)	(8.7)	(9.8)	(9.1)	(10.4)	(9.7)	(11.6)	(10.5)	(9.6)	(11.4)	(11.8)	(11.4)	(11.3)	(12.2)
RPE	11.0	11.0	11.1	11.1	10.5	10.8	11.3	11.3	10.6	11.1	11.9	12.1*	12.1	12.9	13.4#	13.7†
(6-20)	(3.1)	(3.0)	(2.9)	(2.7)	(2.8)	(2.8)	(3.0)	(3.0)	(2.5)	(2.6)	(3.0)	(3.1)	(3.3)	(3.5)	(3.3)	(3.5)
Pain	3.6	3.5	3.6	3.5	3.4	3.7	3.8	3.9	3.6	4.2	4.6	4.8*	5.3	6.0#	6.6^{\dagger}	6.7^
(0-10)	(1.9)	(1.8)	(1.8)	(1.7)	(1.7)	(1.9)	(1.9)	(2.0)	(1.9)	(2.0)	(1.9)	(1.8)	(1.5)	(1.3)	(1.3)	(1.6)

Table 4.3. Measurement values after every set (10 contractions) of the interventions; mean (SD).

Significant differences were set at *p* < 0.05; RPE results: * = significant difference between set 1 and set 4; # = set 3 of NMES+80% BFR significantly larger than all sets of NMES alone, NMES+60% BFR and set 1 of NMES+40% BFR; † = set 4 of NMES+80% BFR significantly larger than all sets of NMES alone, NMES+60% BFR and set 1 and 2 of NMES alone; Pain results: * = significant difference between set 1 and set 4; # = set 2 of NMES+80% BFR significantly larger than all sets of NMES alone, NMES+60% BFR significantly larger than all sets of NMES alone, NMES+60% BFR and set 1 of NMES+40% BFR; † = set 3 of NMES+80% BFR significantly larger than all sets of NMES alone, NMES+60% BFR and set 1 of NMES+40% BFR; † = set 3 of NMES+80% BFR significantly larger than all sets of NMES alone, NMES+60% BFR and set 1 and 2 of NMES+40% BFR; ^ = set 4 of NMES+80% BFR significantly larger than all sets of NMES alone, NMES+60% BFR and set 1 and 2 of NMES+40% BFR; ^ = set 4 of NMES+80% BFR significantly larger than all sets of NMES alone, NMES+60% BFR and set 1 and 2 of NMES+40% BFR; ^ = set 4 of NMES+80% BFR significantly larger than all sets of NMES alone, NMES+60% BFR and set 1 and 2 of NMES+40% BFR; ^ = set 4 of NMES+80% BFR significantly larger than all sets of NMES alone, NMES+60% BFR and set 1 of NMES+40% BFR; ^ = set 4 of NMES+80% BFR significantly larger than all sets of NMES alone, NMES+60% BFR and set 1 of NMES+40% BFR.

4.8 Discussion

The purpose of this study was to standardise and determine if varying BFR pressures induce different acute effects when combined with NMES. The main findings were that the addition of BFR (40-80%) to NMES was required to acutely affect muscular fatigue, with NMES+80% BFR causing greater muscular fatigue (16.2%) than NMES alone (3.5%) (Fig 4.2 and 4.3). However, higher BFR pressures (60-80%) combined with NMES caused higher ratings of pain and RPE than NMES alone and NMES combined with 40% BFR (Table 4.3), with no deleterious cardiovascular effects (Table 4.2 and 4.3).

Our result that NMES combined with 80% BFR induced the greatest muscle fatigue is consistent with findings after BFR alone and combined with low-intensity voluntary isometric contractions[267–269], demonstrating that the addition of BFR acutely reduces force generating capacity and the level of force reduction is dependent on the pressure applied to the limb. For example, Pierce et al. [269] applied BFR (163 mmHg above SBP) passively for 5 x 5 minutes and produced equal knee extension fatigue (16%) to the present study. Our results are also in accordance with prior BFR investigations that found 80% actual and estimated AOP induced acute decrements in MVIC torque[267,270–272]. The acute decrement in MVIC shown here with the addition of BFR (18%) is also similar to that observed after a single bout of resistance exercise (20%), which has correlated with increased muscular strength and size of the VL after training protocols lasting 6 weeks[136,137]. If acute fatigue is desirable for muscular adaptations, our findings provide stronger support for combining NMES with 80% BFR, compared with 40% and 60% BFR (Fig 4.2).

Although mechanistic reasons for our findings were not investigated, muscle fatigue will have occurred due to a number of physiological processes. For example, increases in intramuscular inorganic phosphate concentration have been reported after BFR[273–275]

103

and are a known cause of peripheral fatigue[276,277]. Indeed, others have reported that a combination of submaximal exercise with arterial occlusion rapidly depletes type I and type II muscle fibres of phosphocreatine[278], leading to increases in inorganic phosphate concentration[279]. Decreases in blood flow/O2 delivery associated with BFR, exacerbate this rate of peripheral fatigue[273,277]. Muscle fatigue can be compensated for by increased motor unit activation in an effort to maintain force output[278]. Hence, during fatiguing muscle contractions there is an increased activation of motor units that innervate type II fibres, thus increasing the potential for muscle fibre hypertrophy[279]. This provides one potential reason for the reported relationships between fatiguing tasks (induced by NMES+BFR) and muscle growth[61].

The acute increases in VM and VL MTH observed (Table 4.2), were similar to previous studies that applied BFR combined with resistance exercise using pressures from 40% AOP to 150% SBP[280–283]. Our findings also support previous BFR data, showing no greater muscle swelling effect utilising higher BFR pressures > 40% AOP[282,283]. Muscle swelling has been argued to trigger the proliferation of satellite cells, thus contributing to the hypertrophic response to exercise[284]. The present study supports the use of NMES alone and combined with BFR (40-80%) to induce acute muscle swelling (Table 4.2). However, the observed increase in muscle size, ranging from 0.6 to 1.1 mm, may not be considered clinically significant. It is important to note that the MDC for this measurement was 0.6 mm (Chapter 3). Therefore, the muscle swelling observed falls within the threshold of measurement error, indicating that the changes may not be meaningful.

Pain was increased with the addition of 80% BFR to NMES compared to all of the other conditions in the present study (Table 4.3). Additionally, NMES combined with 60% BFR produced greater ratings of pain than NMES alone (Table 4.3). This indicates that the pain experienced is mostly attributable to the level of occlusive pressure (60-80%). In addition to the nociception of pressure caused by the pneumatic cuff[285], exercise-induced muscle

pain can be generated by stimulation of group III and IV muscle afferents, elicited by metabolic perturbations of the working musculature. It is generally accepted that BFR reduces metabolite clearance, thus inducing greater pain compared to non-occluded exercise[286]. Cuff inflation at higher pressures (80% AOP) has been previously characterised as moderately painful[285], which supports the lower pain ratings observed after NMES and 40% BFR (Table 4.3). The pain ratings observed may be the direct result of pressure changes or metabolic stimulation of group III and IV afferents, induced by the NMES and 80% BFR condition[287]. However, when 60-80% BFR has been applied during resistance exercise, it acutely improved pain up to 24 h after application in knee pain patients[178,179,181]. It is currently unknown if the acute analgesic effect of BFR occurs with NMES and BFR. The lower pain and RPE scores reported with the addition of 40% compared with 60% and 80% BFR to NMES in the present study, may lead to greater clinical applicability. However, NMES combined with different BFR pressures is yet to be investigated in a training study.

There were no unanticipated effects on the cardiovascular system during any of the trials (Table 4.2 and 4.3). This supports previous NMES research using maximal tolerable intensities[288,289] and BFR research using 70% BFR pressures[290,291]. In agreement with the current findings, no adverse events have occurred in healthy and spinal cord-injured adults previously[45–47]. The present findings support the use of NMES and BFR on the selected cardiovascular measures (Table 4.2 and 4.3).

The current study has some limitations, such as the sample, which was restricted to young, healthy men and women. Thus, we acknowledge that our findings may not apply to other populations. Also, the measurements were taken immediately pre and post every experimental condition. Therefore, the time-course of change in the period of time after the intervention is unknown. The investigator and participants were not blinded to experimental conditions. Blinding aims to prevent biased assessment of outcomes and ascertainment bias after randomisation[249]. Future research should, therefore, consider evaluating the

105

time-course responses to BFR and NMES interventions among a wider range of clinical populations who are likely to benefit from its application.

4.9 Conclusion

This is the first study to standardise the BFR pressure using a percentage of AOP when combining it with NMES. To determine which protocol would be best suited for training, we evaluated several factors, including muscle fatigue, muscle swelling, cardiovascular response and perceptual responses. These acute findings suggest that combining NMES with 80% BFR for the quadriceps muscle group may be the most beneficial intervention. However, NMES combined with 40% BFR cannot be excluded due to lower perceptual ratings than NMES+80% BFR and acutely reducing force production (Fig 4.2; Table 4.2), which may be a surrogate marker for muscle growth. We can only speculate that the increased metabolic stress associated with BFR has led to the increased fatigue, RPE and pain ratings observed with the addition of 40-80% BFR to NMES in the present study (Fig 4.2; Table 4.3).

This study showed that the combination of 40%, 60% and 80% BFR to NMES did not cause any deleterious effect on heart rate and blood pressure. No adverse events occurred, and the combined interventions were tolerable to the participants, reflected in the RPE and pain scores. The results also showed that the combination of 40% and 80% BFR to NMES caused acute fatigue, but there was no between group difference. Therefore, aim 1 of this thesis is partially achieved.

In subsequent studies, the duration of NMES intervention times will be modified based on previous research. This decision was influenced by a study conducted by Andrade et al. [47], that showed no effects on muscular strength or hypertrophy in healthy adults when using NMES stimulus durations of less than 10 minutes. Instead, the stimulation duration of

4 x 5 minutes employed by Natsume et al. [46] will be adopted. Furthermore, Natsume et al. [46] observed improvements in muscular strength and hypertrophy after 2-weeks of NMES+BFR in healthy adults. Therefore, subsequent studies will utilise this stimulation duration to assist in optimising the methodological parameters for NMES+BFR, aiming to promote muscular adaptations and pain responses.

Overall, this study provides valuable insights into the combination of NMES and BFR, indicating potential benefits in muscle training. However, further research is needed to explore long-term effects and elucidate the underlying mechanisms for optimal NMES+BFR protocols in enhancing muscular adaptations.

Chapter 5

NMES combined with 40% and 80% BFR pressures, increases muscular strength and hypertrophy without movement: a 6-week Randomised Control Trial
5.1 Abstract

Background: NMES+BFR compared to just NMES alone has shown mixed results with regards to improving muscular strength and size However, previous studies used varied protocols not recommended by previous NMES or BFR research. Objective: The present study investigated the effect of NMES combined with 40% and 80% of BFR using researchrecommended procedures compared to NMES alone. Design: A 6-week randomised control trial. A total of 42 healthy adults randomised into one of three groups: NMES alone; NMES+40% BFR; NMES+80% BFR. Training was 3 times a week for 6-weeks. Main Outcome Measures: Before and after 6-weeks, MVIC between 30-105°, concentric and eccentric peak torque (60°/s), quadriceps CSA's were measured and session RPE, pain and NMES current tolerated were measured in every session. **Results:** NMES+40% BFR caused greater MVIC improvements than NMES alone at 30° (11.3%, p = 0.01), 45° (13.8%, p = 0.01), and greater eccentric (12.4%, p = 0.02) peak torque. NMES+80% BFR caused greater MVIC improvements than NMES alone at 90° (13.6%, p = 0.02). NMES+40% BFR and NMES+80% BFR caused greater increases in quadriceps CSA compared to NMES alone (8.6-9.0%, p = 0.01-0.03). VM and VL NMES currents tolerated were higher with the addition of 40% and 80% BFR to NMES compared to NMES alone (89.1-96.8%, p < 0.001-0.03). Pain and ratings of perceived exertion were no different between all groups. No adverse events occurred. Conclusion: NMES combined with 40% and 80% BFR caused greater increases in quadriceps peak torque and CSA compared to NMES alone without any differences in session RPE and pain. The NMES currents tolerated may explain the differences between groups and this passive combination of interventions could assist with increasing muscle strength and size where voluntary movement is not possible.

5.2 Introduction

As revealed by Chapter 2, the current NMES+BFR literature base shows inconsistencies regarding improving muscular adaptations, potentially being caused by varied methodologies, specifically regarding what BFR pressure should be combined with NMES. The results obtained from Chapter 4 found that the addition of 80% BFR (measured by AOP) to NMES led to greater fatigue (force decrement) compared to NMES alone. Acute reductions in force production serve as indicators of potential muscular adaptations resulting from training protocols[292]. However, no differences in fatigue were observed between the other NMES+BFR conditions (40-80% AOP). Regarding perceptual ratings of pain during the intervention, the combination of NMES with 40% BFR resulted in reduced pain compared to NMES combined with 80% BFR and 60% BFR (Chapter 4) which could promote a more comfortable intervention to undertake. The results from Chapter 4 indicated that the acute fatigue response was the same when combining 40% and 60% BFR to NMES, with lower ratings of pain and RPE experienced with the addition of 40% BFR to NMES. Therefore, the NMES combined with 60% BFR condition was excluded from this study. The stimulation duration was adjusted in this and subsequent Chapters within this thesis to replicate the beneficial effects by Natsume et al. [46]. Chapter 4 found no differences between NMES+80% BFR and NMES+40% BFR regarding the primary outcomes measure, suggesting that the effects on training adaptations cannot be disregarded. Consequently, three groups were chosen to undertake a chronic training intervention, consisting of three sessions per week for a duration of six-weeks. Importantly, both NMES+BFR groups used AOP to determine the 40% and 80% BFR pressures.

The concept of critical power measures the highest sustainable power output that an individual can maintain without reaching exhaustion[220]. Critical power can be estimated in the quadriceps using intermittent MVIC's and is known as end-test torque[224]. In clinical populations, such as those recovering from OA or lower limb surgery, endurance capacity is often compromised, making it challenging for them to engage in prolonged 110

exercise[20,221,222]. Understanding an individual's end-test torque can assist healthcare professionals in designing tailored exercise programs aimed at progressively enhancing endurance capacity[220,223]. End-test torque represents a hyperbolic power-time relationship that forms the foundational basis for comprehending the physiological underpinnings of fatigue development across various exercise intensity domains[293]. When exercising at intensities below the end-test torque, skeletal muscle metabolism achieves a "steady-state" [294]. However, above end-test torque, in the "severe" intensity domain, muscle metabolism does not demonstrate steady-state behaviour, resulting in finite exercise capacity[224]. Notably, the end-test torque has been postulated to signify a 'fatigue threshold'[220], which task failure coincides with complete utilisation of a finite energy pool above the end-test torque (referred to as the impulse above end-test torque). Impulse above end-test torque represents the anaerobically biased work capacity or the additional amount of work an individual can perform above the critical power / end-test torque during highintensity efforts[220,223,295,296]. This coincides with muscle fatigue, as evidenced by decreased maximal voluntary force[223-225]. The end-test torque and impulse above endtest torque can be estimated using an all-out knee extension test for the guadriceps[223-225,297]. These two parameters accurately predict the exercise tolerance limit at severe intensities, which occur when work is done above end-test torgue[224]. Moreover, previous research suggests that impulse above end-test torque is related to a large proportion of type Il muscle fibres, whereas a high end-test torque is associated with a high proportion of type I fibres [294, 298]. BFR exercise causes increased neuromuscular fatigue and type II muscle fibre recruitment compared to free flow exercise[44]. Following 3-weeks of isometric training, improvements in impulse above end-test torque have been observed[225], therefore, we can hypothesise changes in this measure after NMES+BFR. Consequently, acquiring insights into potential muscle fibre type adaptations following NMES and NMES+BFR could assist clinical rehabilitation settings. Muscle disuse involving loss of neural influence and mechanical loading causing a slow-to-fast shift in fibre type and MyHC isoform profile, is usually accompanied by preferential atrophy of type I muscle fibres [50,51]. Therefore, assessing the aforementioned test can offer valuable insights into potential

muscle fibre type adaptations following NMES and NMES+BFR interventions, helping tailor rehabilitation approaches more effectively.

Training session discomfort and RPE have been reported to be higher during NMES+BFR compared to NMES alone in healthy adults[46]. This observation is similar to the wider literature base stating that BFR exercise causes increased ratings of session pain and RPE compared to free flow training[299–301]. However, Natsume et al. [46] did not use AOP to determine there restrictive pressure, which may have led to exaggerated perceptual ratings of discomfort and RPE[57,115]. Both NMES and BFR have been reported to be painful interventions[126,301]. One significant drawback associated with NMES is the elevation of pain levels at higher current intensities[40,60]. Previous research has demonstrated that BFR during resistance exercise and aerobic cycling actually reduces pain sensitivity immediately after training compared to unrestricted free flow training[43,302]. Therefore, session pain and RPE were recorded during this 6-week randomised control trial.

Blood deoxyhaemoglobin concentration [HHb], as measured by NIRS, allows the estimation of muscle microvascular oxygen extraction and provides insight into the dynamic equilibrium between O_2 delivery and utilisation[238,241,243,303]. Low-intensity BFR produces greater changes in blood deoxyhaemoglobin signal [HHb] compared to traditional low-intensity resistance training[304,305], demonstrating higher peripheral oxygen extraction with BFR exercise. The high metabolic demand during low- intensity BFR is strongly associated with a muscle hypertrophy response after only 2-weeks of training (r = 0.87)[306], indicating that reduced mechanical tension can induce muscle hypertrophy following exercise of high metabolic demand. Thus, assessing [HHb] responses during exhaustive tests of maximal torque production, such as a 5 minute all-out test, provides insight into both the mechanical and muscle microvascular responses to NMES and NMES+BFR training and could help to determine the temporal correspondence with measures of muscle hypertrophy. If muscular adaptations can be observed with less reported pain and perceived exertion, this could be

advantageous for reducing the risk of symptom flare-ups in clinical populations experiencing pain.

The primary aim of the present study was to determine if NMES combined with BFR of either (40% & or 80% AOP) is more effective at enhancing muscle strength and hypertrophy than NMES alone, using standardised protocols for 6-weeks. As a secondary aim, this study sought to explore whether any observed adaptations in muscle strength or hypertrophy were accompanied by any changes in impulse above end-test torque, end-test torque and/or HHb signal changes.

It was hypothesised that the combination of 80% BFR to NMES will induce greater changes in muscle strength, hypertrophy than NMES alone and NMES combined with 40% BFR.

A secondary hypothesis was anticipated that the combination of 80% BFR to NMES would result in enhanced impulse above end-test torque and HHb signal changes compared to NMES alone and NMES combined with 40% BFR.

5.3 Method

5.3.1 Ethical approval

All participants provided written informed consent and the study was approved by the University ethics sub-committee (Appendix 2: SMEC_2016-17_104) in accordance with the Declaration of Helsinki (2013).

5.3.2 Participants

Forty-two recreationally active healthy males (n = 28) and females (n = 14) volunteered and were randomised into one of three groups: NMES alone; NMES+40% BFR and NMES+80% BFR. The sample size was calculated *a-priori* using G*Power software and was based upon the effect sizes of research assessing the same outcomes (0.64) [178,182,208]. A group sample size of 14 was determined for the primary outcomes and 12 required for the secondary outcomes. Inclusion criteria were: (a) absence of lower-limb injury, (b) negative answers in the PAR-Q questionnaire (Appendix 5.3), (c) no personal history of cardiovascular or metabolic disease, (d) non-smokers, (e) resting SBP < 140 mmHg and (f) normal range on the ABI test (0.9-1.4)[194]. Participants were instructed to avoid lower limb resistance training for the duration of the study.



Fig 5.1. Consort flow diagram of participants through the study.

	NMES alone (n=14)	NMES+40% BFR	NMES+80% BFR	Р
		(n=14)	(n=14)	
Demographics				
Male : female	9:5	9:5	10:4	>0.86
Age (years)	25.1 (5.6)	26.4 (6.6)	24.6 (6.4)	>0.91
Height (m)	1.75 (0.2)	1.74 (0.1)	1.74 (0.1)	>0.92
Weight (kg)	70.8 (10.8)	72.8 (10.2)	68.4 (9.5)	>0.71
BMI (kg/m ²)	22.9 (2.7)	23.9 (2.3)	22.9 (2.0)	>0.79
Muscle characteristics				
MVIC 30 ^o (Nm)	102.2 (22.0)	105.2 (32.6)	103.4 (25.0)	>0.30
MVIC 45 ^o (Nm)	141.3 (36.2)	133.8 (35.4)	142.7 (43.6)	>0.18
MVIC 60° (Nm)	181.9 (44.4)	184.7 (38.3)	180.3 (50.6)	>0.22
MVIC 75° (Nm)	198.3 (40.1)	210.8 (54.5)	205.4 (67.2)	>0.12
MVIC 90° (Nm)	195.2 (50.6)	208.7 (54.5)	202.3 (64.1)	>0.07
MVIC 105° (Nm)	161.5 (36.1)	168.6 (53.3)	184.2 (59.0)	>0.07
Quadriceps Concentric (Nm)	146.7 (38.7)	150.1 (55.2)	152.7 (38.9)	>0.13
Quadriceps Eccentric (Nm)	179.0 (40.5)	185.0 (50.5)	182.2 (50.8)	>0.09
Vastus lateralis CSA at 50% length (cm ²)	24.7 (6.0)	26.9 (6.0)	27.5 (10.1)	>0.07
Vastus lateralis CSA at 75% length (cm²)	14.4 (4.6)	14.7 (4.5)	17.5 (11.3)	>0.16

Table 5.1. Baseline characteristics of participants; mean (SD).

5.3.3 Study design

The study followed a multi-arm parallel, randomised controlled design. Randomisation (blocks of 4) was generated via online software (<u>http://www.randomization.com</u>). All testing and training were undertaken in the university laboratory. The study underwent pre-trial registration <u>https://clinicaltrials.gov/ct2/show/NCT03662555</u>. Laboratory testing took place on 22 separate occasions, 2 familiarisations, 2 testing and 18 intervention sessions. The initial two sessions served as a familiarisation to the NMES protocol, BFR stimulus and strength testing. These took place > 7-days prior to the beginning of the intervention period and served as a standardised control period prior to the main testing. The third session (PRE) took place 1-day prior the start of the intervention period and all future sessions consisted of muscle strength and hypertrophy outcome measures. The fourth session (POST) took place 48-72 h following 6-weeks of training. All repeated measures were performed at a similar time of day (±1 h) to diminish any circadian effect. All participants were tested at least 2 h postprandial and were instructed to avoid caffeine and exercise prior to testing. After the familiarisation sessions, participants were randomly allocated by

the lead investigator to their experimental groups, with the same trained researcher performing all outcome measurements (Fig 5.2):

- 1) NMES alone
- 2) NMES+40% BFR
- 3) NMES+80% BFR



Fig 5.2. Experimental protocol. All participants performed the same NMES protocol under three different BFR pressures (0, 40 and 80%) for 6-weeks, 3 sessions a week. Outcome measures; VL CSA, isometric peak torque and concentric and eccentric peak torque (ISOK), 5 minute all-out test, NIRS, VA was assessed before and post the 6-week intervention period. Outcome measures assessed after every 5 minutes set included; RPE, pain and max tolerable current achieved.

5.4 Procedures

5.4.1 NMES

See section 3.4. Every training session involved each participant remaining seated with their knees bent to 90°, which matched the same testing position as Chapter 4 and their foot placed in front of an immovable object. The training position was standardised and

remained the same for every participant. Every training session involved a 4 x 5 minutes intervention, with maximum tolerable NMES currents verbally encouraged throughout. Each VM and VL maximal tolerable intensities were also recorded after every training session and used for analysis.

5.4.2 Determination of BFR pressure

See section 3.5. AOP was determined with each participant seated with their knees bent at 90°, matching the position that the intervention was undertaken[116].

5.5 Primary outcomes

5.5.1 Isokinetic Dynamometry

See section 3.7.

5.5.2 Quadriceps CSA

See section 3.9.

5.5.3 Quadriceps muscle architecture

See section 3.10.

5.5.4 RPE

See section 3.14.

5.5.5 Pain

See section 3.15.

5.6 Secondary outcomes

5.6.1 End-test torque and impulse above end-test torque

See section 3.16.

5.6.2 VA

See section 3.17.

5.6.3 Slope analysis on ΔHHb values

See section 3.18.

5.7 Statistical Analysis

One-way analyses of covariance (ANCOVA) were used to determine the effects of group (NMES alone; NMES+40% and NMES+80% BFR) at post-testing. The pre-test measurements were used as covariates for: MVIC30°, MVIC45°, MVIC60°, MVIC90°, MVIC105°, CON, ECC, end-test torque, impulse above end-test torque, VA%, Δ[HHb]; CSA50, CSA75, fascicle length and pennation angle.

Two-way ANCOVAs was used to determine the effects of group (NMES alone; NMES+40% and NMES+80% BFR) and time across 18 time points (intervention week 1-18) for session maximum tolerable currents for VL and VM, pain, RPE. Familiarisation 2 was used as the covariate for max tolerable current and Session 1 was used as the covariate for pain and

RPE. Interactions and main effects were followed with appropriate post-hoc analyses and Bonferroni adjustments.

Statistical significance was set at p < 0.05. Cohens d = 0.2, 0.5 and 0.8 are reported for pairwise comparisons and correspond to small, medium and large effects, respectively.

Statistics were computed using SPSS Statistics software package version 28.0 (SPSS, Chicago, USA). Data are presented as means (SD) unless otherwise stated.

5.8 Results

No adverse events occurred. All participants completed all training sessions (100% adherence).

	NMES alone (n=14)		NMES+40% BFR (n=14)		NMES+80% BFR (n=14)	
	Post	SD	Post	SD	Post	SD
MVIC 30º (Nm)	105.9	24.2	121.0	30.5#	112.4	26.3
MVIC 45º (Nm)	149.3	37.2	159.8	46.8#	161.0	34.9
MVIC 60º (Nm)	191.7	43.3	205.8	49.9	204.3	42.2
MVIC 75º (Nm)	207.1	44.6	241.4	68.1	231.2	61.3
MVIC 90º (Nm)	202.2	53.0	226.0	65.2	237.0	71.4#
MVIC 105º (Nm)	170.9	48.1	184.4	19.5	207.9	68.9
Concentric (Nm)	151.2	37.1	171.5	51.2	174.0	51.3
Eccentric (Nm)	190.2	46.8	219.6	64.9#	207.6	52.7

Table 5.2. Knee extension peak torque, post-test measurement values; mean SD.

Significant difference was set at p < 0.05; # = significantly greater group effect compared to NMES alone.

5.8 Muscle strength

5.8.1 MVIC at 30° knee angle

There was a significant main effect of group on MVIC at 30° [$F_{(2,38)} = 5.69$, p = 0.007, $\eta^2 = 0.23$]. Post-hoc analysis showed higher values in MVIC at 30° for NMES+40% BFR compared with NMES alone (p = 0.005). The mean difference between NMES+40% BFR 120

and NMES alone was 12.14 (SEM = 3.61, n = 14), resulting in a Cohen's d of 0.95. This indicates a large effect size in favour of NMES+40% BFR (d = 0.95, 95% CI [3.1, 21.185]). No differences were observed between NMES+40% BFR and NMES+80% BFR (Table 5.2).



Fig 5.3. Isometric peak torque at a knee angle of 30° ; values as mean \pm SEM; [#] = significant increase compared to NMES alone.

5.8.2 MVIC at 45° knee angle

There was a significant main effect of group on MVIC at 45° [$F_{(2,38)} = 4.53$, p = 0.017, $\eta^2 = 0.19$]. Post-hoc analysis showed higher values in MVIC at 45° after NMES+40% BFR compared with NMES alone (p = 0.014, d = 0.94, 95% CI [2.89, 32.45]). No differences were observed between NMES+40% BFR and NMES+80% BFR (Table 5.2).



Fig 5.4. Isometric peak torque at a knee angle of 45° ; values as mean \pm SEM; [#] = significant increase compared to NMES alone.

5.8.3 MVIC at 60° knee angle

There was no significant main effect of group on MVIC at 60° [$F_{(2,38)} = 2.38$, p = 0.106, $\eta^2 = 0.11$] (Table 5.2).

5.8.4 MVIC at 75° knee angle

There was no significant main effect of group on MVIC at 75° [$F_{(2,38)} = 3.1$, p = 0.057, $\eta^2 = 0.14$] (Table 5.2).

5.8.5 MVIC at 90° knee angle

There was a significant main effect of group on MVIC at 90° [$F_{(2,38)} = 4.05$, p = 0.025, $\eta^2 = 0.18$]. Post-hoc analysis showed higher values in MVIC at 90° after NMES+80% BFR compared with NMES alone (p = 0.023, d = 0.84, 95% CI [3, 51.98]). No significant differences were observed between NMES+40% BFR and NMES+80% BFR (Table 5.2).



Fig 5.5. Isometric peak torque at a knee angle of 90°; values as mean \pm SEM; [#] = significant increase compared to NMES alone.

5.8.6 MVIC at 105° knee angle

There was no significant main effect of group on MVIC at 105° [$F_{(2,38)} = 1.19$, p = 0.32, $\eta^2 = 0.06$] (Table 5.2).

5.8.7 Concentric strength

There was a significant main effect of group on concentric peak torque [$F_{(2,38)} = 3.5$, p = 0.03, $\eta^2 = 0.17$]. However, Post-hoc analysis could not identify any significant differences between groups (p > 0.064) (Table 5.2).

5.8.8 Eccentric strength

There was a significant main effect of group on eccentric peak torque [$F_{(2,38)} = 4.5$, p = 0.017, $\eta^2 = 0.19$]. Post-hoc analysis showed higher values in eccentric peak torque following NMES+40% BFR compared with NMES alone (p = 0.015, d = 0.80, 95% CI [3.7, 42.013]).

No significant differences were observed between NMES+40% BFR and NMES+80% BFR (Table 5.2).





Table 5.3. VL CSA at 50% and	5% length of femur, post-test measurement values; mean
SD.	

	NMES alone (n=14)		NMES+40% BFR (n=14)		NMES+80% BFR (n=14)	
	Post	SD	Post	SD	Post	SD
VL CSA at 50% length (cm ²)	25.2	6.0	29.8	5.6#	30.4	10.5#
VL CSA at 75% length (cm ²)	14.3	4.8	16.3	4.0#	19.5	11.9#
VL fascicle length (cm)	7.8	0.8	8.3	0.9	8.3	0.7
VL pennation angle (°)	19.6	2.2	21.3	3.3*	21.9	4.0

Significant difference was set at p < 0.05; [#] = significantly greater group effect compared to NMES alone; * = significantly greater group effects compared to NMES+80% BFR

5.9.1 VL CSA at 50% length of femur

There was a significant main effect of group on VL CSA at 50% femur length [$F_{(2,38)} = 8.8$, p < 0.001, $\eta^2 = 0.32$] (Table 5.3). Post-hoc analysis showed higher values in VL CSA at 50% femur length after NMES+40% BFR and NMES+80% BFR compared with NMES alone (p = 0.002, d = 0.97, 95% CI [0.79, 4.25] and p = 0.002, d = 0.98, 95% CI [0.8, 4.3], respectively). No significant differences were observed between NMES+40% BFR and NMES+80% BFR (Table 5.3).



Fig 5.7. VL CSA at 50% length of femur; values as mean \pm SEM; [#] = significant increase compared to NMES alone.

5.9.2 VL CSA at 75% length of femur

There was a significant main effect of group on VL CSA at 75% femur length [$F_{(2,38)} = 5.7$, p = 0.007, $\eta^2 = 0.23$]. Post-hoc analysis showed higher values in VL CSA at 75% femur length after NMES+40% BFR and NMES+80% BFR compared with NMES alone (p = 0.028, d = 0.73, 95% CI [0.11, 2.49] and p = 0.011, d = 0.82, 95% CI [0.29, 2.73], respectively).

No significant differences between NMES+40% BFR and NMES+80% BFR were observed (Table 5.3).



Fig 5.8. VL CSA at 75% length of femur; values as mean \pm SEM; [#] = significant increase compared to NMES alone.



Fig 5.9. Example of VL CSA at 50% femur length; pre and post.after NMES+40% BFR.

5.9.3 Fascicle length

There was no significant main effect of group on fascicle length [$F_{(2,38)} = 2.9$, p = 0.065, $\eta^2 = 0.13$] (Table 5.3).

5.9.4 Pennation angle

There was a main group effect on pennation angle [$F_{(2,38)} = 3.9$, p = 0.029, $\eta^2 = 0.17$] (Table 5.3). Post-hoc analysis showed higher values in pennation angle after NMES+40% BFR compared with NMES+80% BFR (p = 0.031, d = 0.72, 95% CI [0.07, 1.84]). No other differences were observed (Table 5.3).



Figure 5.10. Maximal tolerable NMES current after every intervention session. Values as mean \pm SEM; [#] = significantly greater group effect than NMES alone.

5.10 Perceptual variables

5.10.1 Maximum tolerable NMES current

There was a significant effect of time [$F_{(1,38)} = 40.3$, p < 0.001, $\eta^2 = 0.52$] and group x time interaction for greater vastus medialis [$F_{(2,38)} = 10$, p < 0.001, $\eta^2 = 0.34$] and vastus lateralis time [$F_{(1,38)} = 21.6$, p < 0.001, $\eta^2 = 0.36$] and group x time [$F_{(2,38)} = 10.1$, p < 0.001, $\eta^2 =$ 0.35], respectively for max tolerable NMES current increase. Post-hoc analysis confirmed higher values in vastus medialis max tolerable NMES current following NMES+40% BFR and NMES+80% BFR compared with NMES alone (p < 0.001, d = 1.14, 95% CI [143.59, 546.6] and p = 0.031, d = 0.72, 95% CI [18.72, 497.55], respectively). Additionally, higher values in the vastus lateralis max tolerable NMES current was observed following NMES+40% BFR compared with NMES alone (p < 0.001, d = 1.12, 95% CI [134.67, 533.77] and NMES+80% BFR compared with NMES alone p = 0.004, d = 0.94, 95% CI [79.72, 477.19]. No differences between NMES+40% BFR and NMES+80% BFR was observed (Fig 5.10).

	NMES alone		NMES+4	NMES+40% BFR		NMES+80% BFR	
Session	Pain	RPE	Pain	RPE	Pain	RPE	
1	3.3 (1.7)	11.8 (1.6)	4.5 (2.1)	12.5 (2.8)	5.0 (2.1)	13.4 (2.0)	
2	3.6 (1.9)	12.2 (2.1)	4.7 (1.9)	12.9 (3.1)	4.9 (2.6)	13.2 (2.8)	
3	4.0 (1.9)	12.7 (1.3)	4.8 (2.0)	12.8 (2.7)	4.6 (3.0)	12.7 (3.2)	
4	3.9 (1.6)	12.4 (1.6)	4.7 (1.8)	12.9 (2.0)	4.8 (2.2)	12.8 (2.5)	
5	3.6 (1.4)	12.5 (2.0)	4.1 (1.3)	12.3 (2.3)	5.0 (2.1)	12.6 (2.7)	
6	4.1 (1.3)	12.1 (1.8)	4.4 (1.4)	12.6 (2.0)	5.0 (1.8)	12.4 (2.8)	
7	4.4 (1.9)	12.4 (2.6)	4.0 (1.4)	11.9 (1.9)	4.5 (1.9)	12.4 (2.6)	
8	4.5 (2.0)	12.0 (2.8)	4.2 (1.4)	12.4 (1.9)	4.5 (2.0)	12.1 (2.7)	
9	5.4 (3.3)	12.3 (3.1)	3.9 (1.5)	11.8 (2.4)	4.8 (3.3)	3.3 (3.1)	
10	4.6 (2.0)	12.5 (2.8)	4.2 (1.6)	12.5 (1.9)	4.4 (2.0)	12.5 (2.8)	
11	4.9 (2.1)	13.2 (2.8)	4.3 (1.8)	12.1 (2.1)	4.9 (2.1)	12.9 (2.8)	
12	5.3 (3.5)	11.8 (3.0)	4.3 (1.4)	12.2 (1.8)	4.6 (3.5)	12.3 (3.0)	
13	4.2 (1.7)	12.1 (2.6)	4.1 (1.6)	12.0 (1.8)	4.2 (1.7)	12.2 (2.6)	
14	4.6 (1.7)	12.1 (2.2)	4.1 (1.6)	12.4 (2.3)	4.9 (1.7)	12.5 (2.2)	
15	4.7 (2.1)	12.4 (2.7)	4.0 (1.5)	12.2 (1.6)	4.5 (2.1)	12.7 (2.7)	
16	4.3 (1.7)	12.2 (2.6)	3.9 (1.8)	12.1 (2.3)	4.3 (1.7)	12.2 (2.6)	
17	4.1 (1.6)	11.9 (2.4)	4.3 (2.1)	12.5 (2.7)	4.6 (1.6)	12.1 (2.4)	
18	4.5 (2.0)	12.3 (2.8)	4.2 (2.0)	12.4 (2.6)	4.3 (2.0)	12.6 (2.8)	

Table 5.4. Training session pain and RPE; mean (SD)

Significant difference set at p < 0.05.

5.10.2 Pain

There was a significant effect of time [$F_{(1,38)} = 55.8$, p < 0.001, $\eta^2 = 0.60$], however, no group × time interaction observed for pain ratings ($F_{(2,38)} = 0.6$, p = 0.55, $\eta^2 = 0.031$) (Table 5.4).

5.10.3 RPE

There was a significant effect of time [$F_{(1,38)} = 28.3$, p < 0.001, $\eta^2 = 0.43$], however, no group × time interaction observed for RPE ratings ($F_{(2,38)} = 0.98$, p = 0.38, $\eta^2 = 0.049$) (Table 5.4).

	NMES alone (n=12)		NMES+40% BFR (n=12)		NMES+80% BFR (n=12)	
	Post	SD	Post	SD	Post	SD
IAETT (Nms)	7137.4	2306.2	7680.7	3170.2	7729.9	3199.5
ETT (Nms)	78.3	19.2	76.6	10.7	76.1	14.1
VA start (%)	90.8	2.7	90.3	1.6	90.6	1.9
VA end (%)	75.2	6.7	75.1	5.3	75.8	5.8
Δ HHb (mmol/s)	1.29	0.8	1.44	0.6	1.39	0.6

Table 5.5. 5 minute all-out test: impulse above end-test torque; end-test torque; VA at the start and end of the test; Δ [HHb] signal during middle 2 s of each contraction, post-test

Significant difference was set at p < 0.05.

5.11 Secondary Outcomes

5.11.1 Impulse above end-test torque

There was no significant main effect of group on impulse above end-test torque [$F_{(2,32)} = 0.8$, p = 0.46, $\eta^2 = 0.05$] (Table 5.4).

5.11.2 End-test torque

There was no significant main effect of group on end-test torque [$F_{(2,32)} = 0.3$, p = 0.74, $\eta^2 = 0.02$] (Table 5.4).

5.11.3 VA%

There was no significant main effect of group on VA% [$F_{(2,32)} = 0.3$, p = 0.76, $\eta^2 = 0.02$] (Table 5.4).



Fig 5.11. NIRS trace example zoomed into three, 3 second contractions. TSI%: tissue oxygen/saturation index, O2Hb: oxyhemoglobin, HHB: deoxyhemoglobin, tHB:

total hemoglobin

5.11.4 Slope analysis on ΔHHb values

There was no significant main effect of group on slope analysis on Δ HHb values [$F_{(2,164)} =$ 2.09, p = 1.3, $\eta^2 = 0.03$] (Table 5.4).

5.12 Discussion

The main findings of the present study revealed that combining NMES with either 40% or 80% BFR (AOP) for six-weeks, resulted in greater increases in muscular strength (isometric and eccentric) and quadriceps CSA compared to NMES alone (Table 5.2 and 5.3). Secondary findings were that NMES combined with 40-80% BFR caused the participants to tolerate higher NMES currents during the training sessions compared to NMES alone (Fig 5.9). Contrary to our hypothesis, both NMES+40% BFR and NMES+80% BFR were more effective at enhancing muscle strength and hypertrophy than NMES alone. However, greater changes in pennation angle were observed after NMES+40% BFR compared to NMES+80% BFR. Despite the observed increases in muscle strength and hypertrophy and hypertrophy after six-weeks of NMES+BFR, no effects on impulse above end-test torque, end-test torque and Δ [HHb] were observed (Table 5.5).

The present study observed differences between the NMES+BFR groups and NMES alone for increases in isometric (8.1-8.2%), concentric (10.9-11.1%) and eccentric strength (7.7-12.4%) (Table 5.2). Our findings indicate that the current tolerated by the NMES+40% BFR (749.3 [312.0] mA) and NMES+80% BFR groups (678.6 [367.3] mA) was 2.4-2.7 times greater than that tolerated by NMES alone (296.4 [129.1] mA). No differences observed between groups in current tolerated after the familiarisation sessions (36.5-40.1 mA: Fig 5.10). The greater current tolerated in the NMES+BFR groups may have contributed to the enhancement in muscle strength and hypertrophy observed in the present study compared to NMES alone. It is widely recognised that current intensity is a crucial parameter to control when using NMES to enhance isometric strength, with higher current intensities providing a greater stimulus for muscle adaptation [84]. Numerous studies have investigated the effect of NMES training on muscle strength adaptation, and based on previous findings, strength gain following NMES training depends on the magnitude of electrically evoked force during the intervention period[36-40]. Natsume et al. [84] found that using maximum tolerable currents caused greater changes in muscle strength and size after eight-weeks of training vs. using 50% maximum tolerable current. Additionally, Li et al. [81] recently evaluated the effect of BFR, NMES and NMES+BFR during low-intensity squat training on muscle adaptations and muscle activation assessed through surface EMG. They observed a greater increase in muscle activation increase in the NMES+BFR group compared with the BFR group and the control group who only performed the squatting intervention, where the increased muscle activation led to increased muscle strength and may have contributed the effects observed after NMES+BFR in the present study.

The present findings show that the greater current tolerated during NMES+BFR did not cause increased session pain and RPE during the training sessions compared to NMES alone (Table 5.4). One of the main limitations of NMES is the increase in pain with higher current levels[40,60]. Previous research has demonstrated that BFR during resistance exercise and aerobic cycling reduces pain sensitivity immediately after training compared

to unrestricted free flow training[43,302]. This reduction in pain sensitivity after BFR exercise has been partially explained by endogenous opioid and endocannabinoid system pain modulation mechanisms[43,184,302]. The endogenous opioid system utilises peptides such as endorphins, which selectively bind to mu, delta, and kappa receptors, to modulate pain perception by inhibiting pivotal neurotransmitters like substance P. Concurrently, the endocannabinoid system employs retrograde neurotransmitters, such as anandamide, creating interactions with CB1 and CB2 receptors, assisting with the regulation of pain[167,307]. Additionally, activation of a nociceptive descending inhibitory system has been proposed as an explanation of how pain perception is altered by the recruitment of high-threshold motor units[307]. Moreover, increased motor cortex activity has been observed after the recruitment of high-threshold motor units due to increased force production and this has been shown to induce analgesia in humans[308]. Therefore, increased fatigue, as reported previously after NMES+BFR[186] and the previously discussed pain inhibitory mechanisms induced by BFR exercise may explain the higher current tolerances observed in the present study and the subsequent greater muscle strength and hypertrophy after NMES+BFR. This apparent pain modulating effect of NMES+BFR compared to NMES alone needs to be explored regarding pressure, thermal and central mechanisms in both healthy and clinical populations. This will be explored in Chapter 6 and 7 of this thesis, to elucidate whether the same effects of BFR exercise on pain modulation are present after NMES+BFR.

In the present study, hypertrophy changes were 8.9% greater with the addition of BFR to NMES alone; 11% after NMES+40% BFR and NMES+80% BFR vs. 2.1% after NMES alone (Table 5.3). Increased motor unit recruitment levels during strength training, as a result of peripheral fatigue, contributes to muscle hypertrophy. When the working muscle fibres are fatigued, other muscle fibres must be activated in order to maintain the desired levels of force, increasing motor unit recruitment[309–311]. A clear positive relationship exists in the literature between baseline muscle CSA and strength, with greater CSAs correlating with

greater strength capacities[94,312–314]. Our previous study demonstrated that the combination of 40% and 80% BFR to NMES resulted in a decrease in knee extension force immediately after the session, indicating the presence of acute post-session fatigue[186]. This finding is consistent with the notion that session fatigue plays a crucial role in promoting improvements in quadriceps hypertrophy, as observed in the current study.

Regarding architectural changes, the present study observed increases in pennation angle after NMES+40% BFR, which were greater than that observed after NMES+80% BFR. Greater pennation angle allows for a larger amount of contractile tissue to attach to a given area of tendon or aponeurosis, resulting in an increased physiological CSA of the muscle[207]. This enlarged CSA optimises the length tension relationship and facilitates the recruitment of more muscle fibres, ultimately contributing to increased strength[315]. This finding helps explain the greater amount of group effects observed after NMES+40% BFR for isometric and isotonic strength compared to NMES+80% BFR in the present study. Similar observations of muscle thickness increase primarily related to an increase in pennation angle, without changes in fascicle length have been reported in previous BFR research[316]. The lack of change in fascicle length in the present study can be attributed to the specific nature of NMES+BFR contractions, which impose lower demands on the body compared to high load and high-speed isotonic contractions typically used to induce muscle fascicle lengthening[207,317]. It is possible that the lower loads and slower contraction velocities associated with NMES+BFR do not provide a sufficient stimulus for fascicle elongation.

Interestingly, despite the higher current tolerances in the NMES+BFR groups, there were no differences in session pain and RPE scores compared to NMES alone (Table 5.4). Natsume et al. [46] showed contrasting findings, with increased RPE and CR10 scores on the NMES+BFR limb compared with the NMES alone limb in their participants. The differences observed could be due to the present study utilising evidence-based protocols to improve the comfort and efficacy of NMES and BFR, namely using AOP to prescribe the BFR restrictive pressure and placing the NMES electrodes over motor points[57,147]. Placing NMES electrodes over motor points reduces the pain experienced and the evoke torque produced compared to placing the electrodes without locating the motor points[38,98,147]. Furthermore, to maximise quadriceps strength after NMES, it is recommended to use a frequency of 50 Hz[123]. These NMES parameters were not utilised in previous NMES+BFR studies on the quadriceps[46,58,69] but proved effective in increase strength and hypertrophy in the present study when combined with 40% and 80% BFR.

A methodological factor that was addressed in this study was the standardisation of BFR pressures when combined with NMES. Previous studies have implemented BFR by prescribing an arbitrary restrictive pressure[46,47,58] or based their occlusion pressure on SBP[45]. Actual pressure exerted on the vascular system may vary wildly with cuff width, limb thickness and body composition[115]. Therefore, current recommendations are that the BFR pressure should be prescribed via AOP[57]. The present study addressed this issue by prescribing the BFR pressure via AOP, providing a more standardised and accurate approach. Our findings support the beneficial effects of NMES+BFR increasing muscle strength and size reported among healthy and spinal cord injured participants[45,46] and interestingly found no difference between the NMES+40% BFR and NMES+80% BFR conditions, with both being superior to NMES alone (Table 5.2 and 5.3). However, the NMES+40% BFR group showed greater group effects for more isometric angles and eccentric contractions (Table 5.2). This aligns with the recommended BFR pressure range of 40-80% AOP in recent methodology and application guidelines[57]. The same reasoning might therefore explain the lack of effect reported by Slysz et al. [58], where 100% occlusion was applied as part of their BFR method. The authors reported no group differences for quadriceps strength and hypertrophy after NMES+BFR compared to NMES alone. Our findings currently support future NMES+BFR studies utilising the same NMES parameters and a restrictive pressure of 40% AOP, however, NMES+80% BFR cannot be discounted. In clinical practice, the NMES+40% BFR intervention is recommended based on observed increases in eccentric strength, isometric strength, and vastus lateralis hypertrophy without voluntary movement or exercise, as shown in Table 5.2 and 5.3 of the present study.

The secondary objective of this study was to investigate the effects of NMES and/or combined with BFR on impulse above end-test torque, end-test torque and Δ [HHb] during a 5 minute all-out knee extension MVIC test after six weeks of training. These measurements provide valuable insights into muscle and vascular responses[220,224,242,318]. In contrast to previous studies examining whole-body exercise[296,319] and isometric training[225], our findings demonstrated no effects of NMES or NMES+BFR on impulse above end-test torque, end-test torque and Δ [HHb] (Table 5.5).

Impulse above end-test torque serves as a measure of the overall force output generated by a muscle during an MVC by integrating force over-time[224]. However, in our study, despite the observed increases in muscle strength and hypertrophy following both six-week NMES+BFR interventions, no effects were observed on impulse above end-test torque and end-test torque. This discrepancy may be attributed to various factors. One potential explanation could be the involvement of neural adaptations that enhance activation and coordination, resulting in increased strength without a concurrent change in torque produced over time (e.g. impulse)[225]. Additionally, the observed changes in pennation angle, which result in improved recruitment patterns[315] or alterations in muscle fibre composition, may have contributed to increased strength independently of changes in force output measured by IACT[225,315]. Furthermore, increased impulse above end-test torque has been observed after 3 weeks of training. The discrepancy in results could be associated with involuntary NMES contractions compared to voluntary isometric exercises by De-Menezes et al. [225]

136

Muscle oxygen extraction, reflected by Δ [HHb], plays a crucial role in evaluating tissue perfusion and vascular health[233,234,242,320]. However, similar to impulse above endtest torque and end-test torque, no effect on Δ [HHb] was observed in our findings despite the aforementioned muscle adaptations following both NMES+BFR interventions. Improvements in muscle size and strength are intricately linked to enhanced oxygen extraction during exercise. This adaptation involves enhanced capillarization, promoting increased blood flow and efficient oxygen delivery to the muscle[233,321]. Due to an improvement in VL strength and size after NMES+BFR, a concurrent improvement in HHb extraction would be expected. This inconsistency may be explained by factors such as changes in capillary density, alterations in oxygen utilisation or adaptations in local blood flow regulation observed after BFR exercise that do not directly manifest as changes in Δ [HHb][154,242]. Furthermore, the reduced intensity of the NMES+BFR interventions along with the physiological and metabolic difference to involuntary NMES compared with voluntary resistance exercise could have contributed to the lack of effect observed[322-325]. These findings underscore the complex interplay between muscle adaptations, vascular responses and functional outcomes which cannot fully explain the muscle strength and hypertrophy improvements observed after NMES+BFR compared to NMES alone cannot be fully explained by changes in impulse above end-test torgue, end-test torgue or Δ [HHb].

Further research is needed to fully understand the underlying mechanisms and physiological responses associated with NMES+BFR interventions. Additionally, investigating the long-term effects of NMES+BFR on functional outcomes and evaluating its applicability across different clinical populations and muscle groups would contribute to a more comprehensive understanding of this intervention's potential benefits.

5.13 Conclusion

The present findings provide evidence that the addition of 40% and 80% BFR to NMES is more effective at enhancing muscular strength and hypertrophy than NMES alone in healthy adults. This is the first training study to standardise the BFR pressure using a percentage of AOP when combining it with NMES. The perceptual ratings of RPE and pain show no clinically difference between the groups and are lower than those reported by Natsume et al. [46], suggesting that the protocols used in this study may be more effective and comfortable. Importantly, no differences were observed between the NMES+40% BFR and NMES+80% BFR groups, with both showing superiority over NMES alone following the sixweek, three session a week protocol. The main factor contributing to these findings is that NMES+BFR groups tolerated higher currents during the training sessions compared to the NMES alone condition. This study assists in answering research aims 1-3 of this thesis. When voluntary exercise is not possible due to pain and pathology, or when adherence to exercise is challenging, the utilisation of NMES combined with BFR to promote muscular adaptations is a promising and possible solution. Due to the increased tolerable currents observed, subsequent studies in this thesis will explore the effects of NMES+BFR on pain modulation in healthy and clinical populations.

Chapter 6

Effect of NMES combined with BFR on pressure pain, cold pain and temporal summation of pain in healthy adults

6.1 Abstract

Context: BFR with resistance training and aerobic exercise has observed an analgesia effect that can last up to 24 h post its application. This response has not been investigated after NMES+BFR. Objective: The present study investigated the acute effects of NMES combined with varying degrees of BFR on PPT, CPT and TSP Design: Randomised crossover. Setting: Physiology laboratory. Participants: A total of 12 healthy adults (age: 28.2 [5.3] years; height: 177.4 [5.0] cm; body mass: 75.3 [8.1] kg). Interventions: five sessions separated by at least 7 days. The first 2 visits served as familiarisation, with the experimental conditions performed in the final three sessions: NMES alone, NMES+40% BFR, and NMES+80% BFR. Main Outcome Measures: PPT, CPT, TSP were recorded pre, immediately post, 20 minutes post and 45 minutes post each condition, rating of perceived exertion, and pain were recorded after every 5 minutes set of the interventions. Results: The NMES+80% BFR caused greater PPT increase immediately post compared to NMES alone (p = 0.036, d = 0.8, 95% CI [0.1, 3.9]). No other group differences observed. Pain and ratings of perceived exertion observed no group differences. Conclusion: This study showed the NMES+80% BFR acutely increased pain sensitivity as measured by PPT, but this effect had dispersed 20 minutes post the intervention. The acute effect on pain sensitivity observed helps to explain the increased current tolerated during NMES+BFR compared with NMES alone in Chapter 5.

6.2 Introduction

The findings of Chapter 4 indicated that the addition of 40% and 80% BFR to NMES was necessary to induce fatigue, as evidenced by a decrement in force production. Notably, NMES+80% demonstrated higher fatigue compared to NMES alone. The following Chapter (Chapter 5) observed greater muscle strength and size increases after 6-weeks of NMES+BFR compared to NMES alone. Notably, both NMES+BFR groups exhibited greater maximal tolerable currents achieved throughout the training period compared to NMES alone, without increasing perceptual ratings of pain (Table 5.5; Fig 5.9). This finding is contrasting, due to the wider NMES literature observing increased pain and discomfort with increasing currents[40,126,198,326]. This observation from Chapter 5 warrants further exploration to assist with the understanding of the underlying mechanisms in action and this study will assess the effects of NMES+BFR compared to NMES alone on pain sensitivity and session perceptual ratings of pain and RPE in healthy adults.

A notable limitation of both NMES and BFR interventions is that some individuals find the pain associated with these modalities intolerable[49,126], making it challenging to administer an adequate stimulus[38]. As the intensity of the NMES stimulus is increased, excitation of sensory, motor, and pain fibers occurs[97]. While the excitation of the motor neurons is the fundamental premise behind NMES enhancing muscular strength, they are located near free nerve endings and nociceptive receptors, which results in discomfort, pain, and occasionally a burning sensation[326]. BFR combined with aerobic and resistance exercise has observed improvements in pain sensitivity lasting up to 24 hours after its application[43,178,179,181,302]. Korakakis et al. [182] reported an immediate pain reduction during single-leg squat tasks with large to very large effect sizes (d = 0.56-1.32) after BFR resistance exercise, which was sustained for 45 minutes post-intervention in anterior knee pain patients. Hughes et al. [181] reported lower knee pain in anterior cruciate ligament reconstruction (ACL-R) patients following BFR resistance exercise compared to HI-RE, that lasted 24 hours post. The acute effect of NMES+BFR on pain sensitivity is currently unknown. Pain sensitivity is the degree to which an individual is responsive or susceptible to perceiving and experiencing pain, encompassing the threshold for pain initiation as well as the intensity and duration of the pain sensation, with various factors such as genetic predispositions, psychological aspects, previous pain experiences, and underlying medical conditions contributing to its multifaceted nature[33,167,327–329]. Pain pathways involve the intricate transmission and processing of signals from nociceptors to the brain, contributing to the conscious experience of pain. The endocannabinoid system,

with its CB1 receptors in the central nervous system, plays a role in modulating these pain signals[330]. In the context of BFR exercise, studies suggest potential analgesic effects linked the release of endogenous opioids modulation of to and the endocannabinoids[43,302]. Furthermore, exercise induced hypoalgesia (EIH) is a reduction in pain sensitivity following exercise has also been proposed as a heightened response inducing reductions in pressure pain following BFR exercise similar to high intensity resistance training[331]. However, EIH responses differ from healthy to clinical populations[32,332]. In knee OA, EIH can be blunted or even reversed compared to healthy adults, with exercise increasing pain or causing symptom flare ups rather than reducing it.

The greater NMES currents tolerated in Chapter 5 compared to NMES alone (Fig 5.9) suggests that the BFR stimulus has a dampening effect on the pain induced by NMES. The blunting of NMES induced pain could be via reducing sensitivity to pressure pain, cold pain or the central processing of pain[126,333]. Local pain modulation mechanisms encompass processes such as peripheral sensitisation, where injury or inflammation heightens pain sensitivity, as well as peripheral inhibition, where the release of endogenous opioids and activation of specific nerve fibres dampen pain signals[176,307,329]. Central mechanisms include descending pain modulation pathways, where brain regions like the periaqueductal gray and rostral ventromedial medulla release neurotransmitters to inhibit pain transmission[127,334]. Additionally, the gate control theory suggests that non-painful sensory input can block pain signals at the spinal cord level. The assessment of pressure, thermal and central mechanisms will lead to a greater understanding of the NMES+BFR methodologies used and whether they will be appropriate for knee OA patients, who suffer from higher central and peripheral mechanisms of pain than healthy controls [20.25.34.176]. Improved PPT and CPT have been observed after BFR exercise[331,335]. However, no BFR study to date has assessed TSP, which will help elucidate any central processing analgesic effect after NMES+BFR in the present study.

Therefore, the present study aimed to investigate whether NMES+BFR has any impact on pain sensitivity in healthy adults, prior to being tested in knee OA patients. Specifically, the study aims to investigate the acute impact of NMES+BFR on PPT, CPT and TSP compared to NMES alone. To explore whether the observed effects of NMES+BFR on pain sensitivity are consistent with the previously reported pain-reducing effects of BFR when combined with aerobic and resistance exercise. To assist with answering research aims 1 and 4 of this thesis.

Drawing upon the observations from Chapters 4 and 5 and the existing literature, it was hypothesised that NMES+BFR (40% and 80% AOP) would reduce TSP and increase PPT and CPT pain thresholds acutely after NMES+BFR (40% and 80% AOP) compared with NMES alone.

6.3 Method

6.3.1 Participants

Twelve healthy males (n = 6) and females (n = 6) (age: 27.7 [4.5] years; height: 177.0 [5.0] cm; body mass: 74.6 [7.5] kg, and body mass index: 23.9 [1.9] kg/m²) volunteered to participate in this study. The sample size was calculated a priori using G*Power software and the effect sizes of previous research assessing the same outcomes[178,182,208]. Inclusion criteria will be: (a) absence of lower-limb injury, (b) negative answers in the PAR-Q questionnaire (Appendix 5.3), (c) no personal history of cardiovascular or metabolic disease, (d) non-smokers, (e) resting SBP < 140 mmHg and (f) normal range on the ABI test (0.9-1.4)[194]. The study was approved by the University ethics sub-committee at Level 2 (Appendix 3) and in accordance with the Declaration of Helsinki (2013).

6.3.2 Study design

This study followed a randomised crossover design, generated via online software (http://www.randomization.com). All testing was undertaken at the University. Participants were required to visit the University on five occasions, separated by at least 7 days and at the same time of day (±1 h), to minimise the circadian effect. After the two familiarisation sessions, participants were randomly allocated to perform the experimental conditions, with the same trained researcher performing all outcome measurements:

- 1) NMES alone
- 2) NMES+40% BFR
- 3) NMES+80% BFR



Fig 6.1. Experimental protocol. All participants performed the same NMES protocol under three different BFR pressures (0%, 40% and 80%). Outcome measures; PPT, CPT and TSP were assessed before (pre) and immediately post, 20 minutes and 45 minutes post each experimental condition. Outcome measures assessed after every 5 minutes included; RPE and pain.

6.4 Procedures

6.4.1 ABI

See section 3.3.
6.4.2 NMES

See section 3.4.

6.4.3 Determination of BFR pressure

See section 3.5.

6.4.4 PPT

See section 3.19.

6.4.5 CPT

See section 3.20.

6.4.6 TSP

See section 3.21.

6.4.7 RPE

See section 3.14.

6.4.7 Pain

See section 3.15.

6.5 Statistical Analysis

A two-way repeated-measures ANOVA was used to determine the effects of condition (0%, 40% and 80% BFR) and time; PPT, CPT, TSP across four time points (pre, immediately post, 20 minutes post and 45 minutes post), RPE, Pain across four time points (set 1, set 2, set 3, set 4). If the assumptions of ANOVA were violated, the Greenhouse–Geisser 145

correction factor was applied. Interactions and main effects were followed with appropriate *post-hoc* analyses and Bonferroni adjustments.

Statistical significance was set at p < 0.05. Cohens d = 0.2, 0.5 and 0.8 are reported for pairwise comparisons and correspond to small, medium and large effects, respectively.

Statistics were computed using SPSS Statistics software package version 28.0 (SPSS, Chicago, USA). Data are presented as means (SD) unless otherwise stated.

6.6 Results

No adverse events occurred.

6.6.1 PPT

There was a main effect of time ($F_{(2.3,75.6)} = 26.3$, p < 0.001) and group x time interaction for PPT on the dominant leg [$F_{(4.6,75.6)} = 6.5$, p < 0.001, $\eta^2 = 0.28$]. Post-hoc tests showed greater PPT increase immediately post NMES+80% BFR compared with NMES alone (p = 0.036, d = 0.8, 95% CI [0.1, 3.9]). No effect on the non-dominant leg was observed and non-dominant leg [$F_{(4.7,77.6)} = 0.5$, p = 0.61, $\eta^2 = 0.05$] (Table 6.1).

Table 6.1. PPT (kgf/cm²), pre-test, post-test, 20 minutes post-test and 45 minutes post-testmeasurement values, mean (SD).

	NMES alone (n=12) Dominant Non dominant		NMES+40% BFR (n=12)		NMES+80% BFR (n=12)	
			Dominant	Non dominant	Dominant	Non dominant
Pre	5.6 (1.7)	5.5 (1.6)	5.7 (1.6)	5.5 (1.6)	5.6 (1.5)	5.4 (1.3)
Post	5.7 (1.8)	5.6 (1.5)	6.7 (1.9)*	5.4 (1.5)	7.7 (1.8)*#	5.2 (1.4)
Post 20 minutes	5.5 (1.7)	5.3 (1.6)	5.9 (1.8)~	5.2 (1.4)	6.0 (2.2)~	5.1 (1.3)
Post 45 minutes	5.4 (1.8)	5.5 (1.8)	5.7 (1.6)~	5.2 (1.4)	5.8 (2.0)~	5.0 (1.4)

Significant difference was set at p < 0.05; * = significant increase compared to pre time point; # = significantly greater group effect compared to NMES alone; ~ = significant decrease compared to post time point.

There was no main group x time interaction for CPT on the dominant [$F_{(5,82.4)} = 0.6$, p = 0.67, $\eta^2 = 0.04$] and non-dominant leg [$F_{(4.7,77.6)} = 0.6$, p = 0.80, $\eta^2 = 0.03$] (Table 6.2).

Table 6.2. CPT, pre-test, post-test, 20 minutes post-test and 45 minutes post-test measurement values, mean (SD).

	NMES alone (n=12) Dominant Non dominant		NMES+40	% BFR (n=12)	NMES+80% BFR (n=12)	
			Dominant	Non dominant	Dominant	Non dominant
Pre	1 (0.9)	0.9 (0.9)	1.1 (1.1)	1.1 (1.2)	1.1 (1.1)	1.1 (1.3)
Post	0.8 (0.9)	0.8 (0.9)	0.8 (0.8)	0.9 (1)	0.7 (0.8)	1 (1.1)
Post 20 minutes	0.8 (0.9)	0.8 (0.9)	0.8 (0.9)	0.8 (0.9)	0.9 (0.9)	0.9 (1.1)
Post 45 minutes	0.9 (0.9)	0.8 (0.9)	0.8 (0.9)	0.8 (1)	0.9 (1.1)	1 (1.2)

Significant difference was set at p < 0.05.

6.6.3 TSP

There was no main group x time interaction for TSP [$F_{(3.2,52)} = 0.3$, p = 0.83, $\eta^2 = 0.02$] (Table

6.3).

Table 6.3. TSP, pre-test, post-test, 20 minutes post-test and 45 minutes post-testmeasurement values, mean (SD).

	NMES alone (n=12)		NMES+40%	NMES+40% BFR (n=12)		% BFR (n=12)
	Mean	SD	Mean	SD	Mean	SD
Pre	1.67	(0.98)	1.79	(0.99)	1.83	(1.53)
Post	1.67	(1.83)	1.58	(1.24)	1.42	(1.31)
Post 20 minutes	1.50	(1.57)	1.42	(1.24)	1.50	(1.51)
Post 45 minutes	1.46	(1.59)	1.46	(1.34)	1.50	(1.62)

Significant difference was set at p < 0.05.

6.6.4 Pain

There was no main group x time interaction for session Pain [F $_{(3.7,60.4)} = 0.1$, p = 0.97, $\eta^2 = 0.1$

0.06] (Table 6.4).

6.6.5 RPE

There was no main group x time interaction for session RPE [$F_{(3.2,52.2)} = 0.2$, p = 0.88, $\eta^2 = 0.01$] (Table 6.4).

	NMES alone (n=12)		NMES+40%	BFR (n=12)	NMES+80% BFR (n=12)	
	Pain	RPE	Pain	RPE	Pain	RPE
Set 1	4.3 (1.3)	12.6 (2)	4.1 (0.8)	13.1 (1.7)	5.5 (1.3)	14.5 (1.1)
Set 2	4.2 (0.9)	12.4 (2.1)	4.2 (1)	13.0 (1.8)	5.5 (1.2)	14.6 (1.6)
Set 3	4.0 (0.9)	12.2 (2.2)	3.9 (0.8)	12.8 (2)	5.4 (0.8)	14.5 (1.5)
Set 4	3.5 (0.5)	11.8 (2.2)	3.6 (0.9)	12.5 (1.8)	5.0 (1.1)	13.7 (1.8)

Table 6.4. Pain and RPE values after set 1, 2, 3 and 4 mean (SD).

Significant difference was set at p < 0.05.

6.7 Discussion

The main finding of this study is that adding 80% BFR to NMES results in an immediate increase in PPT compared to NMES alone (Table 6.1; Fig 6.2). However, it does not affect CPT and TSP. The PPT effect was not observed at the 20 minutes post-intervention time point (Table 6.1). No effect on the contralateral thigh was observed after all conditions or measures (Table 6.1-6.3).

Our findings align with previous research demonstrating that BFR during resistance exercise and aerobic cycling reduces pain sensitivity (i.e., increased PPT) immediately after training compared to unrestricted free flow training[59,297]. However, in contrast to the 24 h duration observed after BFR aerobic exercise[302], BFR resistance exercise in ACL-R patients[181] and 45 minutes in anterior knee pain patients[182], the observed effect in the present study did not last up to 20 minutes post-intervention. Additionally, no effect on PPT was observed in the contralateral quadricep, suggesting that spinal level inhibitory mechanisms were not involved in the present findings[167]. The observed absence of an effect on TSP further aligns with the discernment that NMES coupled with BFR yields no discernible systemic or central nervous system pain-modulating effects in the context of healthy adults. These findings strongly suggest that NMES combined with an 80% BFR stimulus imparts a localised and transient impact on PPT within the specifically targeted 148

quadricep muscle. However, the apparent lack of influence on spinal level inhibitory mechanisms prompts considerations regarding its potential clinical efficacy, particularly in populations afflicted with knee OA, characterised by heightened central sensitisation relative to pain-free controls[25]. Further exploration of these implications is undertaken in Chapter 7. Previous BFR research assessing PPT locally and distally has found a distal effect and reductions in PPT in the contralateral limb[336]. Moreover, previous research in healthy adults has shown that isometric exercise produces the largest effect on reducing pain sensitivity compared to dynamic movements and the effect sizes reported are much greater than those observed in our study [173]. One key difference is the use of involuntary isometric contractions elicited by NMES in the present study, compared to voluntary actions. This disparity in contraction modes may contribute to the different findings observed in our study compared to both BFR exercise and isometric contractions on PPT. NMES actions produce involuntary contractions that bypass the central nervous system and activate muscles directly through the recruitment of both type I and type II muscle fibres with minimal current[80]. On the other hand, voluntary actions recruit muscles according to the size principle, with type I fibres recruited at low intensities and type II fibres recruited as the intensity of exercise increases[76]. Previous NMES research has also observed metabolic differences between voluntary and NMES contractions[326], suggesting that differences in contraction physiology could explain the contrasting effects observed in the present study, which warrant further exploration.

NMES applied using moderate and maximum tolerable currents in healthy adult's has observed acute increases in PPT[337]. They utilised 10 s contractions compared to the 5 s contractions used in this study, and the electrode placement was different, with electrodes placed just below the inguinal crease compared to 13.4 cm lower as in the present study due to the width of the BFR cuff. These methodological differences could have contributed to the contrasting findings[337]. Pain inhibition is complex and multifactorial, involving mechanisms including; EIH, long term depression of nociceptive synaptic connections and descending inhibition via mechanisms involved in diffuse noxious inhibitory control and conditioned pain modulation[332,334,337]. The magnitude of EIH depends on various factors, including exercise parameters such as type, dose, duration and intensity, with EIH responses increasing as the exercise intensity increases[33,332].

Regarding muscle pain during each set of the interventions, the NMES+80% BFR condition was more painful by 1.3-1.5 on a scale of 0-10 during each set compared to NMES alone and NMES+40% BFR (Table 6.4) and lower than those observed in Chapter 4. Similar findings were observed by Natsume et al. [46] who utilised a similar protocol as this study and reported higher pain and RPE values compared to NMES alone. However, 5 on the 0-10 NPRS represents moderate pain only[219], with no participants withdrawing from this study due to the discomfort experienced, indicating this observation warrants little concern in healthy adults. This finding is also similar to previous BFR research, which reported higher muscle discomfort after cycling with 80% BFR compared to 40% BFR[302]. RPE values for resistance exercise combined with 40% BFR ranged from 15 to 17, while for 80% BFR it was 18 to 19[43]. The results observed in the present study are similar or lower, which may also contribute to explaining the reduced effect observed. The phenomenon known as diffuse noxious inhibitory control, or the "pain inhibits pain" effect, is part of the descending endogenous analgesia system. Experimental studies in healthy adults have demonstrated a decrease in pain intensity after a noxious conditioning stimulus when stimulated at adjacent and remote sites [178,337]. The decreased perceptual ratings in the present study may help explain the reduced and shorter-lasting findings compared to previous BFR research.

Furthermore, previous research has suggested that exercise intensity and duration affect the magnitude of EIH response[173]. During heavy load lifting or when the muscle is fatigued, high-threshold motor units are typically recruited in addition to low-threshold motor

150

units to prevent force failure based on the size principle[76,338]. Both high-load resistance exercise and low-load resistance exercise performed to volitional failure result in similar magnitudes of hypoalgesia, which are greater compared to low-load resistance exercise performed to failure[187,302]. This suggests that the recruitment of high-threshold motor units may be one of the underlying mechanisms of EIH. The addition of 40-80% BFR to NMES causes greater neuromuscular fatigue than NMES alone (Chapter 4) and promotes increased muscle strength and size (Chapter 5), indicating recruitment of high threshold motor units. Activation of a nociceptive descending inhibitory system has been proposed as an explanation of how pain perception is altered by the recruitment of high-threshold motor units[303]. Furthermore, increased motor cortex activity has been observed after the recruitment of high-threshold motor units law to increased force production, and this has been shown to induce analgesia in humans[304]. The increased fatigue seen in Chapter 4, could have led to the increased PPT immediately after NMES+80% BFR observed in these findings. However, future research should elucidate other physiological mechanism that could have contributed to the observed results.

The previously observed hypoalgesia responses after BFR exercise have been partially explained by endogenous opioid and endocannabinoid system pain modulation mechanisms[59,297,298]. The endogenous opioid system consists of naturally occurring peptides that bind to opioid receptors in the central nervous system[299]. These receptors are involved in pain perception and activation of the opioid system can lead to a reduction in pain the sensation[167]. The endocannabinoid system consists of endogenous cannabinoids such as anandamide and 2-arachidonoylglycerol, along with their receptors CB1 and CB2. The endocannabinoid and endogenous opioid systems can interact with each other and with other neurotransmitter systems in the body to modulate pain perception[299]. The acute improvement in PPT after NMES+80% BFR is comparable to, but lower in magnitude and shorter in duration than, the improvements seen following BFR resistance and aerobic exercise[43,183,302]. These findings suggest that while NMES+80% BFR can enhance PPT, the effects are less pronounced and do not persist as

long as those achieved through BFR exercise modalities. One plausible explanation for the observed improvements in PPT with NMES+80% BFR is the activation of the endogenous opioid and endocannabinoid systems, which are known to play significant roles in pain modulation and have been implicated in the analgesic effects associated with various forms of exercise[307]. The endogenous opioid system involves the release of opioid peptides, such as endorphins, enkephalins, and dynorphins, which bind to opioid receptors in the nervous system to produce analgesic effects. Similarly, the endocannabinoid system, which includes endocannabinoids like anandamide and 2-arachidonoylglycerol (2-AG), interacts with cannabinoid receptors to modulate pain and inflammation[307]. BFR exercise has been shown to elevate PPT and levels of these endocannabinoids after aerobic and resistance exercise[302,336]. However, since blood tests were not conducted in the present study, it is not possible to conclusively determine the involvement of these endogenous systems in the pain modulation observed with NMES+80% BFR. Future research should include biochemical analyses to measure levels of endogenous opioids and endocannabinoids preand post-intervention to clarify their role in mediating the analgesic effects of NMES+80% BFR.

No effect on TSP was observed in the present study. Factors that have been found to affect TSP include: stimulus intensity, frequency and duration, psychological factors such as fear, anxiety and attentional focus, and previous pain experiences, such as individuals suffering from chronic pain, can heighten TSP[251,252]. To the authors knowledge, this study is the first to assess TSP after a BFR intervention. TSP involves central nervous system mechanisms including, peripheral sensitisation, enhanced synaptic transmission, wind-up phenomenon and central sensitisation[334]. The present study involved healthy subjects, which may explain the lack of effect on TSP observed, as the TSP ratings were relatively low compared to clinical populations[219,305].

Furthermore, no effect was observed on cold pain thresholds. A similar finding was reported after an ischemic pre-conditioning stimulus (100% BFR), which found no effect on cold pain

152

sensitivity[306]. This result could also be attributed to the relatively low CPT scores observed during pre-testing in the present study[307] (Table 6.2).

6.8 Conclusion

In conclusion, the present study demonstrated that the addition of 80% BFR to NMES resulted in a greater acute increase in PPT immediately after the intervention compared with NMES alone (Table 6.1). However, this effect was not sustained at the 20 minutes post-intervention assessment and did not affect the contralateral limb. No group differences were observed in CPT and TSP. These findings partially align with previous research demonstrating that BFR during resistance exercise and aerobic cycling reduces pain sensitivity immediately after training compared to free flow training[43,302]. Nevertheless, the reduced and shorter-lasting effect observed in the present study could be attributed to the use of involuntary NMES contractions instead of voluntary actions and methodological variations to previous NMES research. The underlying mechanisms responsible for the observed analgesic effect remain unclear, however, the greater PPT in the present study may be attributed to increased neuromuscular fatigue and subsequent recruitment of high threshold motor units to prevent force failure[188,190,255,297,298,301]. Notably, no differences in session RPE and pain were observed, and there was no effect on TSP or CPT. The perceptual ratings in this study were lower than Chapter 4 and previous BFR research, which may have contributed to the reduced effect on PPT. Overall, our findings suggest that the addition of BFR to NMES may have potential as an acute pain management strategy and help to explain the greater NMES currents tolerated during NMES+BFR in Chapter 5. Additionally, the lack of effects of TSP and CPT may be attributed to the relatively low baseline scores and the healthy population studied.

The final experimental Chapter investigated if NMES+BFR promotes an analgesic effect in knee OA patients.

153

Chapter 7

Effect of NMES combined with BFR on pressure pain, cold pain, temporal summation of pain and function in knee OA patients

7.1. Abstract

Context: Acute reductions in pain sensitivity have been observed after BFR exercise and NMES+BFR in Chapter 6 of this thesis. However, no effect lasted 20 minutes and it was conducted using healthy adults. Knee OA patients report increased pressure, thermal and conditioned pain responses to healthy adults. Objective: The present study investigated the acute effects of NMES combined with varying degrees of BFR on PPT, CPT, TSP, 5repetition sit-to-stand, and knee bends in 30 seconds in knee OA patients. Design: Randomised crossover. Setting: Private physiotherapy clinic, London. Participants: A total of 12 knee OA patients (age: 68.8 [5.8] years; height: 174.0 [8.1] cm; body mass: 80.2 [9.4] kg, and body mass index: 26.6 [3.4] kg/m²). Interventions: five sessions separated by at least 7-days. The first 2 visits served as familiarisation, with the experimental conditions performed in the final three sessions: NMES alone, NMES+40% BFR, and NMES+80% BFR. Main Outcome Measures: PPT, CPT, TSP, 5-repetition sit-to-stand and knee bends in 30 seconds were recorded pre, immediately post, 20 minutes post and 45 minutes post each condition, RPE, and pain were recorded after every 5 minutes set of the interventions. Results: The NMES+80% BFR caused greater 5-repetition sit-to-stand improvement at 45 minutes post compared to NMES alone (p = 0.04, d = 0.76, 95% CI [0.09, 4.71]). NMES+80% BFR also caused acute improvements in pain during the 5STS, PPT and TSP (p < 0.05). However, no between group differences were observed. Session pain and RPE were higher after NMES+80% BFR compared to both other groups (p > 0.05). Conclusion: This study showed that NMES+80% BFR acutely improved 5-repetition sit-to-stand function 45 minutes post its application, with reduced pain and also improved PPT and TSP. The findings present an acute pain modulating effect after NMES+80% BFR in knee OA patients.

7.2 Introduction

Chapter 6 within this thesis demonstrated that the addition of 80% BFR to NMES led to a transient reduction in pain sensitivity, as measured by PPT compared to NMES alone in healthy adults (Chapter 6). However, this effect did not last 20 minutes post the intervention and no effects on CPT or TSP were observed (Chapter 6). It is important to note that this previous study was conducted on healthy adults without any injury or pathology. This current study aims to extend these findings to a clinical population experiencing pain. Knee OA is a degenerative joint disease that affects the articular cartilage, subchondral bone and synovium of the knee joint. It is characterised by inflammation, pain, stiffness and a loss of joint function[308]. The early signs and symptoms of knee OA may include mild pain and stiffness, particularly after periods of inactivity or overuse. As the condition progresses, the pain can increase in severity and become more persistent, and may be accompanied by swelling and crepitus and associated loss of knee range of motion[308]. Because OA is considered to be a mechanically driven disease, altered joint loads are likely a requirement for its development and progression. Functional limitations associated with knee OA include muscle weakness and atrophy, particularly of the quadriceps, which play a key role in stabilising the knee joint[23,308]. These limitations can lead to individuals being unable to perform activities that require bending, lifting and supporting their body weight. However, due to the clinical signs and symptoms traditional resistance exercise to help restore these functional limitations can commonly be too painful to perform and exacerbating this condition[308]. Therefore, interventions that promote muscular adaptations with reduced mechanical stress i.e., NMES and BFR have been investigated in knee OA patients[309-311].

In knee OA, alterations in pain sensitivity compared to healthy controls are intricately linked to physiological mechanisms[20,25,32]. The degeneration of joint tissues induces peripheral sensitisation, marked by the release of inflammatory mediators and heightened responsiveness of nociceptors. Knee OA also involves central sensitisation, where the central nervous system undergoes changes amplifying pain perception. Dysfunctions in temporal summation and diffuse noxious inhibitory control contribute to an increased sensitivity to pain stimuli[20,25,32]. Moreover, pain in knee OA influences quadriceps function, creating a reciprocal relationship. Quadriceps dysfunction, often attributed to pain-related inhibition and altered neuromuscular control, can exacerbate the pain experience and contribute to the functional limitations observed in knee OA[15].

Exercise-induced hypoalgesia (EIH), the reduction of pain following exercise, manifests differently in individuals with knee OA compared to healthy controls, largely influenced by distinct physiological mechanisms[32,348]. Knee OA is characterised by heightened baseline pain sensitivity as detailed above. In contrast to healthy individuals, the magnitude of the EIH response in knee OA may cause no reduction in pain sensitivity or increased pain perception and symptom flare ups, potentially attributable to the persistent nociceptive input and altered pain processing associated with the condition[32]. The mechanisms underlying EIH in healthy controls, involving the activation of endogenous pain-inhibitory systems such as the release of endorphins and engagement of descending inhibitory pathways, may be compromised or less efficient in knee OA[32,307]. Moreover, the chronic nature of knee OA pain, rooted in structural joint changes and ongoing inflammation, may limit the duration of EIH effects in individuals with the condition. Therefore, testing NMES+BFR on pain modulation in knee OA patients may produce different results than observed in Chapter 6 on healthy adults and warrants investigation.

The current literature suggests that there may be a hypoalgesia effect after BFR exercise[178–181]. Blood flow restriction training has been shown to reduce pain across a range of training programs and in a variety of clinical conditions including knee pain and OA[178–181]. Interestingly, the reduction in pain with low load BFR appears to be greater than high load resistance exercise (HL-RE). In contrast, a network meta-analysis on electrical stimulation therapies concluded that interferential

current compared with NMES, is the most promising pain relief treatment for individuals with knee OA. However, the findings suggest that among electrical stimulation therapies, interferential current stands out as the most promising option for alleviating pain in knee OA[309]. In contrast to NMES, interferential electrical stimulation does not enhance muscular strength. Therefore, its effectiveness on assisting knee OA with their functional limitations is limited[148,349]. Due to the previously observed acute improvements in PPT in healthy controls along with enhanced muscular strength and size seen after NMES+80% BFR (Chapter 5 and 6), which could assist with the previously mentioned functional declines in knee OA, this study investigated the effect of NMES+BFR on pain and function in knee OA patients.

Due to Chapter 6 observing greater reductions in pain sensitivity with the addition of 80% BFR to NMES in a healthy population, the aim of the present study is to assess if these findings also extend to a clinical population experiencing pain. An acute pre to post change in PPT was also observed after NMES+40% BFR and therefore was not discounted from further investigation. Sit-to-stand and knee bending functional activities are reduced and reported to be painful in knee OA patients[20]. Korakakis et al. [178] observed reduced pain and improved performance during a single leg squatting task after a single session of BFR exercise in anterior knee patients. Therefore, functional measures were added to the previously used pain outcome measures in Chapter 6 to be investigated in knee OA patients. The purpose of this study is to determine if NMES+BFR has any acute effect on TSP, PPT, CPT and function in knee OA patients compared to NMES alone. Additionally, the study aims to investigate if these acute improvements in pain thresholds, if observed, translate to enhanced physical function acutely following NMES+BFR compared to NMES alone to assist in answering research aim 4 and 5. The aim of the present study was to determine if NMES+BFR has any acute lasting up to 45 minutes post its application on PPT, CPT, TSP in knee OA patients. Secondary aims include exploring whether any observed acute improvements in pain thresholds translate into enhanced physical function immediately following NMES+BFR (40% and 80% AOP) compared to NMES alone, addressing research aims 4 and 5 of this thesis. Additionally, assessing the safety of NMES+BFR in knee OA patients, a population associated with older age and potential cardiovascular comorbidities[29,34], by re-evaluating cardiovascular measures previously examined in Chapter 4.

It was hypothesised that knee OA patients subjected to NMES+BFR (40% and 80% AOP) will exhibit acutely increased PPT compared to NMES alone. Physical function assessed immediately after NMES+BFR will improve compared to NMES alone. No significant changes in CPT or TSP are expected between NMES+BFR and NMES alone. Finally, the acute cardiovascular measures assessed in Chapter 4, will demonstrate the safety of NMES+BFR in knee OA patients.

7.3 Method

7.3.1 Participants

Twelve knee OA patients were recruited and volunteered for this study; males (n = 8) and females (n = 4) (age: 68.8 [5.8] years; height: 174.1 [8.1] cm; body mass: 80.2 [9.4] kg, and body mass index: 26.6 [3.4] kg/m²). The sample size was calculated using G*Power software and the effect sizes of previous research assessing the same outcomes[178,182,208]. Inclusion criteria were: (a) diagnosis of knee osteoarthritis from GP / consultant following the NICE guidelines and X ray imaging[350], (b) one sided symptomatic knee OA, (c) no personal history of cardiovascular or metabolic disease, (d) non-smokers, (e) resting SBP < 140 mmHg and (f) normal range on the ABI test (0.9-

1.4)[194]. The study was approved by the University ethics subcommittee (Appendix 4: SMU_ETHICS_2021-22-329) in accordance with the Declaration of Helsinki (2013).

7.3.2 Study design

The study followed a randomised crossover design, generated via online software (http://www.randomization.com). All testing was undertaken at a private Physiotherapy Clinic in London. Participants were required to visit the clinic location on five occasions, separated by at least 7 days and at the same time of day (±1 h), to minimise the circadian effect. After two familiarisation sessions, participants were randomly allocated to perform the experimental conditions, with the same trained researcher performing all outcome measurements:

- 1) NMES alone
- 2) NMES+40% BFR
- 3) NMES+80% BFR



Fig 7.1. Experimental protocol. All participants performed the same NMES protocol under three different BFR pressures (0%, 40% and 80%) Outcome measures; KOOS, SBP, DBP, HR, PPT, CPT, TSP, 5-repetition sit-to-stand, knee bends in 30 seconds were assessed before (pre) and immediately post, 20 minutes and 45 minutes post each experimental condition. Outcome measures assessed after every 5 minutes included; RPE and pain.

7.4 Procedures

7.4.1 ABI

See section 3.3.

7.4.2 NMES

See section 3.4.

7.4.3 Determination of BFR pressure

See section 3.5.

7.4.4 PPT

See section 3.16.

7.4.5 CPT

See section 3.17.

7.4.6 TSP

See section 3.18.

7.4.7 5-repetition sit-to-stand

See section 3.19.

7.4.8 Knee bends in 30 seconds

See section 3.20.

7.4.9 KOOS questionnaire

See section 3.21.

7.5 Statistical Analysis

A two-way repeated-measures ANOVA was used to determine the effects of condition (0%, 40% and 80% BFR) and time; PPT, CPT, TSP, 5-repetition sit-to-stand, and knee bends in 30 s across four time points (pre, immediately post, 20 minutes post and 45 minutes post), Session pain and RPE across four time points (set 1, 2, 3 and 4) and KOOS, SBP, DBP, HR across two time points (pre, immediately post). If the assumptions of ANOVA were violated, the Greenhouse–Geisser correction factor was applied. Interactions and main effects were followed with appropriate *post-hoc* analyses and Bonferroni adjustments.

Statistical significance was set at p < 0.05. Cohens d = 0.2, 0.5 and 0.8 are reported for pairwise comparisons and correspond to small, medium and large effects, respectively.

Statistics were computed using SPSS Statistics software package version 28.0 (SPSS, Chicago, USA). Data are presented as means (SD) unless otherwise stated.

7.6 Results

No differences were observed between baseline values across the three experimental conditions (p > 0.05) and no adverse events occurred.

There was a main group x time interaction for PPT on the symptomatic leg [$F_{(5,78.5)} = 5.0$, p < 0.001, $\eta^2 = 0.23$]. However, Post-hoc analysis could not identify any significant differences between groups at any time point (p > 0.14). No effect was observed on the non-symptomatic quadricep (p < 0.05) (Table 8.1).

There was no main group x time interaction for PPT on the non-symptomatic leg [$F_{(6,99)} = 0.8$, p = 0.54, $\eta^2 = 0.05$] (Table 8.1).

Table 7.1. PPT (kgf/cm²), pre-test, post-test, 20 minutes post-test and 45 minutes post-testmeasurement values, mean (SD).

	NMES alone (n=12)		NMES+40%	NMES+40% BFR (n=12)		BFR (n=12)
	OA	Non-OA	OA	Non-OA	OA	Non-OA
Pre	4.4 (2.0)	4.3 (2.0)	4.3 (1.7)	4.3 (2.0)	4.3 (2.0)	4.4 (2.1)
Post	4.4 (1.9)	4.4 (2.0)	5.1 (2.3)*	4.3 (2.0)	5.4 (2.3)*	4.2 (2.0)
Post 20 minutes	4.5 (1.7)	4.7 (2.1)~	5.0 (2.3)*	4.4 (2.0)	5.8 (1.5)*	4.7 (1.4)
Post 45 minutes	4.3 (2.0)	4.8 (2.4)	4.7 (2.0)	4.4 (2.0)	5.9 (1.5)*	4.7 (1.5)

Significant difference was set at p < 0.05; * = significant increase compared to pre time point; ~ = significant increase compared to post time point.

7.6.2 CPT

There was no main group x time interaction for CPT on the symptomatic leg [$F_{(4.5,74.2)} = 1.7$,

p = 0.16, $\eta^2 = 0.09$] and non-symptomatic leg [$F_{(4.6,68.0)} = 0.9$, p = 0.49, $\eta^2 = 0.03$] (Table 8.2).

Table 7.2. CPT, pre-test, post-test, 20 minutes post-test and 45 minutes post-test measurement values, mean (SD).

	NMES alone (n=12) OA Non-OA		NMES+40%	NMES+40% BFR (n=12)		% BFR (n=12)
			OA	Non-OA	OA	Non-OA
Pre	2.6 (1.1)	2.3 (1.0)	2.2 (1.1)	1.9 (1.1)	2.1 (1.4)	2.0 (1.2)
Post	2.5 (1.3)	2.3 (1.2)	2.1 (1.4)	2.0 (1.3)	2.0 (1.4)	2.0 (1.2)
Post 20 minutes	2.2 (1.4)	1.9 (1.3)	2.2 (1.5)	1.9 (1.3)	1.8 (1.6)	1.5 (1.5)
Post 45 minutes	2.5 (1.2)	2.4 (1.0)	2.1 (1.1)	2.0 (1.2)	1.8 (1.4)	1.6 (1.4)

Significant difference was set at p < 0.05.

7.6.3 TSP

There was no main group x time interaction for TSP [$F_{(4.4,72.7)}$ = 0.5, p = 0.75, η^2 = 0.03] (Table 8.3).

Table 7.3. TSP, pre-test, post-test, 20 minutes post-test and 45 minutes post-test measurement values on the OA limb, mean (SD).

	NMES alone (n=12)		NMES+40%	NMES+40% BFR (n=12)		% BFR (n=12)
	Mean	SD	Mean	SD	Mean	SD
Pre	4.00	(1.48)	3.50	(1.45)	3.75	(1.14)
Post	3.58	(1.98)	3.42	(1.51)	3.17	(1.47)
Post 20 minutes	2.75	(1.96)	2.17	(1.11)	2.42	(0.79)
Post 45 minutes	3.04	(1.76)	2.04	(1.84)	1.88	(1.17)*

Significant difference was set at p < 0.05; * = significant decrease compared to pre time point.

7.6.4 5-repetition sit-to-stand

There was a main group x time interaction for 5-repetition sit-to-stand time [$F_{(4.2,69.4)} = 4.2$, p = 0.03, $\eta^2 = 0.20$]. Post-hoc analysis showed reduced time to perform 5-repetition sit-to-stand following NMES+80% BFR compared to NMES alone at the 45 minutes post time-point (p = 0.04, d = 0.76, 95% CI [0.09, 4.71]). No significant differences were observed between NMES+40% BFR and NMES+80% BFR.

There was a main group x time interaction for pain experienced during the 5-repetition sitto-stand [$F_{(3.1,50.7)} = 4.0$, p = 0.01, $\eta^2 = 0.19$] (Table 8.4). However, Post-hoc analysis could not identify any significant differences between groups at any time point (p > 0.29) (Table 7.4).

and 45 minutes post-test measurement values, mean (SD). NMES alone (n=12) NMES+40% BFR (n=12) NMES+80% BFR (n=12) Pain Pain Pain Time Time Time 4.8 (1.9) Pre 14.5 (2.2) 4.5 (1.8) 14.1 (2.6) 4.6 (1.9) 14.1 (2.6) 4.3 (2.1) Post 12.6 (2.4)* 3.8 (2.0)* 12.7 (2.4)* 14.0 (2.4) 3.8 (2.1)* 12.0 (1.9)* 3.3 (2.0)* Post 20 minutes 14.1 (2.1) 4.5 (1.5) 12.3 (2.4)* 4.0 (2.1) 11.6 (2.1)*~# Post 45 minutes 14.0 (2.3) 4.4 (1.5) 12.2 (2.4)* 3.8 (2.0) 3.2 (1.9)*

Table 7.4. 5 sit-to-stand time (s) and pain (0-10), pre-test, post-test, 20 minutes post-test and 45 minutes post-test measurement values, mean (SD).

Significant difference was set at p < 0.05. * = significant decrease compared to pre time point; ~ = significant decrease compared to pre time point. # = significant group effect compared to NMES alone.

7.6.5 Knee bends in 30 seconds

There was no main group x time interaction for knee bends performed in 30 s [$F_{(3.3,54.2)}$ = 1.8, p = 0.15, $\eta^2 = 0.10$]. Additionally, there was no main group x time interaction for pain experienced during knee bends performed in 30 s [$F_{(2.5,41.3)}$ = 1.6, p = 0.21, η^2 = 0.09] (Table 8.5).

Table 7.5. Knee bends on OA limb; amount and pain (0-10), pre-test, post-test, 20 minutespost-test and 45 minutes post-test measurement values, mean (SD).

	NMES alone (n=12)		NMES+40% E	8FR (n=12)	NMES+80% BFR (n=12)	
	Amount	Pain	Amount	Pain	Amount	Pain
Pre	18.6 (4.2)	3.7 (2.0)	19.6 (4.9)	3.6 (2.1)	19.3 (5.0)	3.7 (2.0)
Post	19.2 (4.4)	3.5 (2.1)	20.8 (4.7)	3.2 (1.8)	20.5 (5.3)	3.3 (2.0)
Post 20 minutes	19.3 (4.6)	3.6 (1.9)	21.3 (4.8)	2.2 (1.6)	21.4 (4.7)*	2.3 (1.5)
Post 45 minutes	18.9 (4.5)	3.6 (1.6)	22.2 (4.8)*	2.3 (1.6)	22.0 (4.9)*	2.3 (1.6)

Significant difference was set at p < 0.05. * = significant increase compared to pre time point.

7.6.6 KOOS

There was no main group x time interaction for session Pain [$F_{(2,33)} = 0.1$, p = 0.88, $\eta^2 = 0.08$].

7.6.7 SBP

There was no main group x time interaction for session RPE [$F_{(2,33)} = 1.4$, p = 0.26, $\eta^2 = 0.08$].

7.6.8 DBP

There was no main group x time interaction for session DBP [$F_{(2,33)} = 0.6$, p = 0.58, $\eta^2 = 0.03$].

7.6.9 Pain

There was no main group x time interaction for session Pain [$F_{(2.7,44.4)} = 0.3$, p = 0.82, $\eta^2 = 0.02$]. (Table 7.6).

7.6.10 RPE

There was no main group x time interaction for session RPE [$F_{(3.5,58.5)} = 1.7$, p = 0.16, $\eta^2 = 0.10$] (Table 7.6).

	NMES alone (n=12)		NMES+40	% BFR (n=12)	NMES+80%	BFR (n=12)			
	Pain	RPE	Pain	RPE	Pain	RPE			
Set 1	4.9 (0.8)	13.0 (1.1)	5.2 (1.5)	13.6 (1.7)	6.3 (1.0)~#	14.2 (1.6)			
Set 2	4.9 (0.8)	12.9 (1.2)	5.0 (1.0)	13.5 (1.8)	6.3 (0.9)~#	14.5 (1.7)			
Set 3	4.8 (0.7)	12.8 (1.1)	5.0 (1.0)	13.5 (1.8)	6.3 (0.9)~#	14.7 (1.6)#			
Set 4	4.8 (0.9)	12.3 (1.5)	4.8 (1.0)	13.3 (1.4)	6.3 (0.9)~#	14.6 (1.4)#			

Table 7.6. Pain and RPE values after set 1, 2, 3 and 4, mean (SD).

Significant difference was set at p < 0.05. ~ = significant group effect compared to NMES+40% BFR. [#] = significant group effect compared to NMES alone.

7.7 Discussion

The purpose of this study was to investigate the acute effects of different BFR pressures combined with NMES and NMES alone on pain thresholds and function in knee OA patients. The main findings demonstrated that 4 x 5 minutes sets of NMES+80% BFR improved time taken to perform 5 sit-to-stand repetitions by 17.7% compared to NMES alone (3.4%) at the 45 minutes post-intervention assessment (Table 7.4). Additionally, the addition of 80% BFR to NMES was necessary to acutely affect pain sensitivity, as indicated by an improvement

in PPT (Table 7.1) and reduction in TSP (p < 0.05) (Table 7.3). Both the 40% and 80% NMES+BFR groups acutely improved sit-to-stand and knee bend abilities, but only NMES+80% BFR was greater than NMES alone for 5-repetition sit-to-stand performance (p < 0.05), without observed effects on CPT or self-reported KOOS (Table 7.2). However, it is important to note that NMES+80% BFR resulted in higher ratings of session pain and RPE than NMES alone and NMES+40% BFR (Table 7.6), without any detrimental effects on cardiovascular parameters.

The findings of this study can be explained through several mechanisms. Firstly, the combination of NMES+BFR may have acutely improved muscle activation[46,81], leading to enhanced functional performance in the 5-repetition sit-to-stand test, compared to NMES alone. The addition of 80% BFR to NMES likely induced increased metabolic stress, an ischemic environment and muscle recruitment, which could promote local muscle adaptations and growth factors, ultimately acutely enhancing muscle function in knee OA patients[46,81]. Moreover, NMES+80% BFR may have reduced inhibitory signals (AMI), allowing for increased quadriceps force production[56]. AMI is a reflex inhibition of quadriceps activation resulting from abnormal afferents from a damaged joint, leading to decreased motor drive and limited force generation by the muscles[19]. Pain and quadriceps weakness are closely related in knee OA, with AMI being associated with pain rather than structural damage[19,25,56]. Previous research has shown that reducing discomfort, can decrease AMI and improve muscle function in knee OA patients[19,56]. Furthermore, reducing pain has shown to enhance muscle activation and MVC in knee OA patients[351]. The reduction in pain experienced during the 5-repetition sit-to-stand after NMES+80% BFR likely contributed to the reduced time taken to perform the 5 repetitions at the 45 minutes post-intervention assessment time point(Table 7.4). The reduction in pain and performance enhancement observed in the present study could be attributed to reductions in AMI[351].

BFR may also improve AMI through various mechanisms, including the promotion of reactive hyperaemia and increased microvascular filtration capacity induced by ischemic reperfusion[234,304,352,353]. These physiological responses may contribute to enhanced vascular reactivity, such as FMD responses as seen in previous NMES+BFR research[45]. Reactive hyperaemia refers to the increased blood flow that occurs following a period of reduced blood flow or ischemia[354]. BFR restricts blood flow to the working muscles, creating ischemia. Once the BFR is released, there is a rapid increase in blood flow to the muscles, resulting in reactive hyperaemia[44]. Furthermore, BFR has been shown to increase the production of nitric oxide within the muscle[44]. Nitric oxide has various physiological functions, including promoting blood vessel dilation and improving blood flow, thus enhancing the delivery of oxygen, nutrients and other essential factors to the muscles[19,44,56]. This may help mitigate the inhibitory effects of AMI in knee OA[56]. Additionally, ischemic reperfusion, which refers to the restoration of blood flow after a period of ischemia, can lead to improved microvascular filtration capacity[234]. Ischemia followed by reperfusion triggers a cascade of physiological responses that can enhance vascular function, including improved endothelial function, increased vasodilation, the release of hypoxia-inducible factors which play a role in tissue repair and angiogenesis potentially improving muscle function and contributing to the results observed in the present study[234,304,352,353].

NMES+BFR in Chapter 4 induced higher levels of fatigue than NMES alone. This fatigue leads to increased recruitment of muscle fibres, including fast-twitch fibres responsible for generating high force[279] which may have helped overcome the inhibitory effects of AMI and promote increased function in the present study and the previously observed chronic strength and hypertrophy gains observed in Chapter 5.

In addition to the physiological mechanisms mentioned above, the acute reduction in TSP observed from pre to post, in conjunction with improved 5-repetition sit-to-stand performance and reduced pain during this task after NMES+80% BFR, may be attributed to

169

the activation of large diameter sensory fibres through the gate control theory, inhibiting pain signals transmitted through small diameter fibres due to OA, potentially alleviating pain during functional movements observed[167,176]. Moreover, BFR-induced release of endogenous opioids and neuro-modulatory effects may have further modulated pain perception, resulting in improved performance of functional movements[43,331].

TSP refers to an increased perception of pain in response to a repeated stimulus due to amplification of pain processing mechanisms in the central nervous system[355]. In the present study, the absence of an effect observed after NMES alone suggests that the NMES+80% BFR may have modulated the central nervous system's response to pain in knee OA, leading to reduce pain perception during the 5-repetition sit-to-stand test[355]. This finding warrants further exploration to better understand the underlying mechanisms.

Higher ratings of session pain and RPE reported during the NMES+80% BFR sessions compared to NMES alone or NMES+40% BFR suggest that this combination imposes a greater subjective discomfort in knee OA patients. However, still only moderate scores on the NPRS were reported and with no drop outs due to the discomfort experienced, or flare ups in their pain or OA symptoms reported, this warrants little concern. The findings in the present study are supported by previous BFR research, which reported higher muscle discomfort and RPE after cycling and resistance exercise with 80% BFR compared to 40% BFR[302]. The phenomenon known as diffuse noxious inhibitory control, or the "pain inhibits pain" effect, is part of the descending endogenous analgesia system[356]. Experimental studies in healthy adults have demonstrated a decrease in pain intensity after a noxious conditioning stimulus [178,337]. The increased perceptual ratings of pain and RPE in the present study may help explain the acute effects of PPT and TSP observed. The difference in the results observed in the present study compared to Chapter 7 of this thesis is primarily due to the reduced PPT and increased TSP ratings observed in the OA patients at baseline compared to healthy adults, which aligns with previous findings in knee OA populations[25]. Nonetheless, it is reassuring that no adverse events occurred, or any deleterious cardiovascular responses were observed, suggesting the safety of the interventions used in knee OA patients.

In line with the findings of Chapter 6 and 7 of this thesis, previous research did not find any effect on cold pain thresholds in healthy adults or knee OA patients following NMES alone or in combination with 40% and 80% BFR (Chapter 6 and 7). Similar results were observed in BFR research that utilised 100% AOP during cold water immersion[357]. Cold pain, or the sensation of pain in response to cold stimuli, is typically transmitted through the cold thermoreceptors in the skin and travels via the spinothalamic pathway to the central nervous system[250,328]. Whereas, pressure pain travels through the spinothalamic pathway via nociceptors (pain receptors)[358]. The findings that NMES+BFR affects pressure pain but not cold pain, although speculative, suggests that it likely targets the nociceptors or pain signalling pathways specific to pressure pain rather than impacting thermal signals[328,359] and warrants further investigation.

7.8 Conclusion

In conclusion, the findings of this study demonstrated that the addition of 80% BFR to NMES resulted in an improvement in the time taken to perform 5 sit-to-stand repetitions and reduced pain during the task, indicating enhanced functional ability. Moreover, this coincided with acute improvements in PPT and TSP after NMES+80% BFR. However, both NMES+BFR groups demonstrated acute improvements in sit-to-stand and knee bending abilities from pre to post intervention assessments.

Although speculative, the observed benefits of NMES+80% BFR can be attributed to several mechanisms, including reductions in AMI, increased muscle activation, ischemic environment as well as potential modulation of pain perception, improved vascular reactivity and reduced inflammation. However, it is important to note that the NMES+80% BFR intervention was associated with higher levels of session pain and RPE. Despite this

discomfort, no adverse events or cardiovascular issues were reported, emphasising the safety of the interventions in knee OA patients.

These findings highlight the potential of NMES combined with BFR as a safe adjunctive therapy for managing knee OA. Further research is needed to explore the long-term effects, optimal pressures and individualised approaches to maximise the benefits of this intervention, due to disparities in acute effects observed in the present study compared to Chapter 6 in healthy adults in this thesis. Considering patient comfort and subjective experiences will be crucial in implementing these interventions in clinical practice. Ultimately, the integration of NMES+BFR holds promise for improving muscle function, reducing pain sensitivity and potentially leading to enhanced long-term outcomes in knee OA patients.

Chapter 8

General discussion

8.1 Main findings

This thesis set out to investigate the synergistic benefits of combining NMES with BFR to enhance muscular adaptations and pain modulation compared to NMES alone. The research encompassed acute and chronic training studies on healthy adults and those suffering from knee OA. The initial phase focused on methodological considerations, assessing safety and efficacy of NMES+BFR at different pressures. Chapter 4 led to the exclusion of NMES+60% BFR. A subsequent 6-week study confirmed NMES+BFR's safety and efficacy in enhancing muscle strength and size without movement and no differences between NMES+40% BFR and NMES+80% BFR. The final chapters evaluated NMES+BFR's impact on pain and function, revealing reduced pain sensitivity and improved physical function, particularly in knee OA patients. No adverse events were reported throughout the thesis. The collective findings warrant a comprehensive discussion on muscle adaptation, cardiovascular safety, and pain modulation.

8.2 Muscle size

Chapter 5 observed, 8.6-8.8% greater quadriceps hypertrophy at 50% length of the femur were 8.3-8.8% greater with the addition of BFR to NMES alone; 10.8% total after NMES+40% BFR (d = 0.97) and 10.6% after NMES+80% BFR (d = 0.98) compared to 2% after NMES alone. Additionally, 9.0-9.1% greater quadriceps CSA was observed at 75% length of the femur with the addition of BFR to NMES alone; 11.2% after NMES+40% BFR (d = 0.73) and NMES+80% BFR 11.3% after NMES+80% BFR (d = 0.82) compared to 2.2% after NMES alone.

Results observed during Chapter 5 observed lower hypertrophy than Gorgey et al. [45] who reported extensor carpi radialis longus CSA increased in the NMES+BFR forearm of their participants by 17%, d = 1.1, using a similar CSA US measurement for hypertrophy as used in Chapter 5, but for the forearm rather than the quadriceps. Additionally, Skiba et al. [69] 174

observed a slightly higher percentage increase of 11.7% from Chapter 5, but a lower effect size d = 0.6 from pre to post training (p < 0.05) on twenty-one men with complete spinal cord injury (SCI). They used shorter sessions lasting only 12 minutes (3 x 4 minutes sets), rather than 20 minutes (4 x 5 minutes sets) used throughout this thesis and they used unrecommended parameters of 20 Hz frequency, minimal current to elicit a contraction, rather than max tolerable and not placing the electrodes over the quadriceps motor points which could explain the lower results observed using similar restrictive pressure of 40% used in Chapter 5. It is hypothesised that the spinal cord injured muscles used in both of these may be more responsive to training than the healthy adults used in Chapter 5 [90,166,360,361].

The other NMES+BFR studies using healthy adults, rather than clinical populations all observed lower effect sizes for improvements in muscle size than Chapter 5 of this thesis. Natsume et al. [46] observed muscle thickness of the quadriceps increased after 2-weeks of training (+3.9%, d = 0.18) and decreased after 2-weeks of detraining (-3.0%), whereas no notable change was observed under NMES alone. Although they used muscle thickness, the biggest difference is their intervention only lasted 2-weeks. Previous BFR reviews suggest that training durations of 6-weeks and over are optimal for enhancing muscular adaptations[48,49]. This is supported by the NMES+BFR research showing greater hypertrophy than NMES alone when performing 6-weeks training interventions[45,69] and Chapter 5.

Slysz et al. [58] only observed a 1% increase (d = 0.04) in the quadriceps size after their 6week intervention. Bergamasco et al. [112] observed increased CSA of the quadricep, using the same CSA measurement as used in Chapter 5, NMES+BFR = 23.0 (2.7) cm², 4.6%, d = 0.4. 20 training sessions were performed for both protocols over 6-weeks. Both used 100% BFR and found the lowest effect on muscle hypertrophy, using 6-week interventions, Natsume et al. [46] was slightly lower using only a 2-week intervention. The high 100% BFR pressure is not recommend in the wider BFR literature[57] and could have contributed to the reduced results observed compared to Chapter 5. Furthermore, a partial occlusion method (40-80%) is also supported by animal model data observing muscle growth after NMES+BFR[64,362] and by this thesis.

8.3 Muscle strength

Chapter 5 observed isometric strength increases from 30-105° were on average 8.1-8.2% greater with the addition of BFR to NMES alone; 13% total after NMES+40% BFR (d = 0.43) and NMES+80% BFR 12.8% (d = 0.44) compared to 4.8%, d = 0.2 after NMES alone for isometric contractions. Isokinetic strength, concentric and eccentric strength increases 9.2-11.7% greater with the addition of BFR to NMES alone; 16.5% total after NMES+40% BFR (d = 0.5) and NMES+80% BFR 14% (d = 0.48) compared to 4.8%, d = 0.17 after NMES alone for isokinetic contractions. These results observed are similar, but also more or less pronounced to previous NMES+BFR studies conducted prior to Chapter 5 of this thesis[45–47,58] and two studies published after it began[69,112].

Slysz et al. [58] observed a moderate effect size d = 0.69, 23.6%, for increased isometric knee extension using max tolerable currents. Natsume et al. [46] reported NMES-BFR isometric strength improved greater than Chapter 5 (+14.2%), but lower isokinetic improvements (+7.0% at 90°/s and +8.3% at 180°/s) conditions after their 2-weeks of training, (d = 0.64 isometrically, 0.31 at 90°/s and 0.35 at 180°/s). The reduced training duration and BFR stimulus could have equated to the reduced isokinetic strength observed. Furthermore, greater pennation angle increases were observed after NMES+40% BFR in Chapter 5 of this thesis, which could have contributed to the greater increases isokinetic strength observed. Increasing the pennation angle has differing effects on isokinetic and isometric strength. This observed effect on pennation in Chapter 5 may also help to explain the greater isokinetic strength increases.

Regarding strength increases after NMES, the current tolerated is the main factor[38,126,363]. The greatest improvement in isometric observed previously used max tolerable currents[58], then Chapter 5 results using max tolerable currents, followed by Natsume et al. [46] using a current intensity that produced a 5%–10% MVIC and the smallest effect observed by Andrade et al. [47] using a current amplitude to achieve a 20% MVIC. The wider NMES literature, the NMES+BFR research to date along with this thesis, supports that max tolerable currents, rather than aiming for low level MVIC's, is recommended for optimising both isometric and isokinetic strength using NMES combined with BFR.

Other parameters of note are the NMES frequency, with the most effective for enhancing strength for the quadriceps is 50 Hz[58] observed in Chapter 5. However, the improvements observed by Natsume et al. [46] using only 30 Hz need to be explored in a longer duration study to determine its efficacy compared to 50 Hz. The work-to-rest ratio of 1:1 for NMES contractions also seems optimal regarding previous NMES+BFR research, which differs from the wider NMES literature recommending large rest periods to limiting fatigue[38] and leads to enhancement of muscle strength and hypertrophy after NMES+BFR (40-80%).

8.4 Cardiovascular safety

The findings of Chapter 4 revealed no meaningful differences in HR responses between the different groups; NMES alone, NMES+40% BFR, NMES+60% BFR and NMES+80% BFR. For blood pressure, no interaction effect was observed. The BP responses after training for SBP ranged from 1.4-2.5 mmHg and DBP 0.1-1.7 mmHg, with no between group differences observed. The findings from Chapter 6, using NMES alone, NMES+40% BFR and NMES+80% BFR, observed no effects on HR, SBP, or DBP in knee OA patients. No effect of NMES+BFR on blood pressure was also observed in spinal cord injured patients also reporting no adverse events[45]. The lack of deleterious cardiovascular effect observed

in Chapter 4 and 6 of this thesis, using different population groups is promising for determining the safety of this approach along with no adverse events occurring throughout all of the studies in this thesis.

8.5 Pain modulation

The findings of Chapter 4, showed higher session ratings of pain and RPE during there NMES alone and NMES combined with 40, 60% and 80% BFR interventions compared to Chapter 5 onwards. The main difference in the parameters used was the set durations being 1.4 minutes in Chapter 4 and 5 minutes from Chapter 5 onwards. The longer set duration of 5 minutes from Chapter 5 onwards resulted in in no interaction and lower pain and RPE scores. These results are supported by research on conditioned pain modulation, which suggests that session duration and frequency play a role in reducing pain modulation experienced. The findings suggest that the 5 minutes set duration used from Chapter 5 onwards is optimal for perceptual comfort.

During Chapter 5, pain values, RPE, and maximum tolerable currents were reported. The pain experienced in each session decreased from the first session to the last for both NMES+BFR groups, while the NMES alone group experienced an increase in pain. This coincided with higher NMES currents used and tolerated under both NMES+BFR conditions compared to NMES alone. The subjects in the NMES alone group did exhibit conditioning to the stimulus, as evidenced by the greater currents tolerated from the first to the last session. However, the conditioning effect was greater with the addition of BFR. This finding aligns with the observations made by Natsume et al. [46] who reported greater reductions in session discomfort with NMES+BFR compared to NMES alone and the acute reductions in PPT observed after NMES+BFR in Chapter 6. After NMES+40% BFR, the effect size at the immediate post time point was moderate (d = 0.6) and after NMES+80% BFR was large (d = 0.8). However, the effect did not last for 20 minutes, which differs from previous BFR

research showing that the PPT increase lasts up to 24 hours[331]. In Chapter 7, the effect size observed for pain reduction during the 5-repetition sit-to-stand task at the 45 minutes post-intervention time point was d = 0.5 compared to NMES alone. We observed similar improvements in pain during functional squatting tasks as reported by Korakakis et al. [182]. This improvement in pain and 5-repetition sit-to-stand performance also coincided with a reduction in TSP after NMES+80% BFR and lasted 45 minutes as observed previously[182]. Although speculative, the reduced pain sensitivity and TSP could have led to improvements in AMI, leading to enhanced muscle activation[81] and therefore 5-repetition sit-to-stand performance in knee OA patients in Chapter 7[355].

8.6 Clinical practice guidelines

The results of this thesis suggest the optimal NMES+BFR protocol for enhancing muscular strength (isometric and eccentric) and VL size, particularly in situations where exercise is contraindicated or challenging due to pain, is NMES combined with 40% BFR for 4 sets of 5 minutes, three times a week for six weeks.

In the context of knee OA patients, the use of NMES combined with 80% BFR is recommended to contribute to short-term improvements in pain modulation, characterized by more than a 19% reduction in the time taken to perform 5 sit-to-stand movements with decreased pain after a single session.

In line with results from this thesis, clinicians should expect to observe >10% improvements in muscle size and >12% improvements in muscle strength following a 6 week, 3 session a week protocol. The NMES electrodes should be placed over muscle motor points and AOP should be determined in the position that the treatment is undertaken. The NMES parameters reported in Chapter 5 onwards should be utilised to enhance muscular strength,

size and assist in modulating pain when combined with 40% and 80% BFR. Max tolerable currents should be encouraged throughout all treatment sessions.

8.7 Limitations

Throughout this thesis, only the quadriceps muscle group has been tested due to its importance to physical function and to help provide an optimal methodology. The same protocol may have different results on varying muscles due to their physiological characteristics and their functional demands.

The cohort used for three experimental studies of this thesis was based around university students and staff members. Therefore, a further limitation will be the results are limited to the age groups tested and also socioeconomic population associated with the students and staff at St Marys University, Twickenham.

8.8 Future Research Considerations

One consistent theme throughout the thesis is the need for standardised protocols and methodologies for NMES and BFR interventions. Further research should focus on establishing consensus guidelines for the application of NMES+BFR, including parameters such as stimulation frequency, intensity, duration and BFR pressure.

While this thesis investigated three different BFR pressures (40%, 60% and 80%), there is still a need to explore a wider range of pressures. Future research should examine the acute and long-term effects of NMES combined with lower and higher BFR pressures to determine the optimal pressure for specific populations and outcomes.
Further research is warranted to elicit to exact mechanisms behind the muscle adaptations and acute improvements in pain sensitivity and function observed throughout this thesis. Further research should explore the neural, vascular, and molecular mechanisms involved in the NMES+BFR response. NMES+BFR potentially reducing inhibition and enhancing muscle function at an accelerated rate is really promising and holds a lot of value for the rehabilitation of knee OA patients.

8.9 Conclusion

In conclusion, this thesis investigated the effects of combining NMES with BFR on muscle strength, size, function and pain in both healthy and knee OA patients. Overall, the findings of this thesis support the use of NMES+40% BFR and NMES+80% BFR in increasing muscle strength, size and pain modulation compared to NMES alone in both healthy adults and knee OA patients. In knee OA, the greater reductions in pain sensitivity and functional improvements after NMES+80% BFR currently make it a preferable choice over NMES+40% BFR. However, further controlled trials are needed to determine the long-term effectiveness of NMES+80% BFR compared to NMES+40% BFR.

When voluntary exercise is not possible or contraindicated due to pain or pathology, NMES+BFR provides a safe and efficacious intervention to enhance muscular adaptations and improve pain modulation, which can assist with the rehabilitation of clinical populations.

References

- 1 Seguin R, Nelson ME. The benefits of strength training for older adults. *Am J Prev Med*. 2003;25:141–9. doi: 10.1016/S0749-3797(03)00177-6
- Jones SW, Hill RJ, Krasney PA, et al. Disuse atrophy and exercise rehabilitation in humans profoundly affects the expression of genes associated with the regulation of skeletal muscle mass. The FASEB Journal. 2004;18:1025–7. doi: 10.1096/fj.03-1228fje
- Jang SY, Park J, Ryu SY, et al. Low muscle mass is associated with osteoporosis: A nationwide population-based study. *Maturitas*. 2020;133:54–9. doi: 10.1016/j.maturitas.2020.01.003
- 4 Ingemann-Hansen T, Halkjaer-Kristensen J. Computerized tomographic determination of human thigh components. The effects of immobilization in plaster and subsequent physical training. *Scand J Rehabil Med.* 1980;12:27–31.
- 5 Wall BT, Dirks ML, Snijders T, *et al.* Substantial skeletal muscle loss occurs during only 5 days of disuse. *Acta Physiologica*. 2014;210:600–11.
- 6 Deitrick JE. The effect of immobilization on metabolic and physiological functions of normal men. *Bull N Y Acad Med.* 1948;24:364.
- 7 White MJ, Davies CTM, Brooksby P. The effects of short-term voluntary immobilization on the contractile properties of the human triceps surae. *Quarterly Journal of Experimental Physiology: Translation and Integration*. 1984;69:685–91.
- 8 Kawakami Y, Akima H, Kubo K, *et al.* Changes in muscle size, architecture, and neural activation after 20 days of bed rest with and without resistance exercise. *Eur J Appl Physiol.* 2001;84:7–12.
- 9 Gibson JNA, Halliday D, Morrison WL, *et al.* Decrease in human quadriceps muscle protein turnover consequent upon leg immobilization. *Clin Sci.* 1987;72:503–9.
- 10 Stuart CA, Shangraw RE, Prince MJ, *et al.* Bed-rest-induced insulin resistance occurs primarily in muscle. *Metabolism.* 1988;37:802–6.
- 11 Bergouignan A, Rudwill F, Simon C, *et al.* Physical inactivity as the culprit of metabolic inflexibility: evidence from bed-rest studies. *J Appl Physiol.* 2011.
- 12 Haruna Y, Suzuki Y, Kawakubo K, *et al.* Decremental reset in basal metabolism during 20-days bed rest. *Acta Physiol Scand Suppl.* 1994;616:43–9.
- 13 Tzankoff SP, Norris AH. Effect of muscle mass decrease on age-related BMR changes. *J Appl Physiol*. 1977;43:1001–6.
- 14 Reid KF, Naumova EN, Carabello RJ, *et al.* Lower extremity muscle mass predicts functional performance in mobility-limited elders. *J Nutr Health Aging.* 2008;12:493.
- 15 Rice DA, McNair PJ, Lewis GN. Mechanisms of quadriceps muscle weakness in knee joint osteoarthritis: The effects of prolonged vibration on torque and muscle activation in osteoarthritic and healthy control subjects. *Arthritis Res Ther.* 2011;13. doi: 10.1186/ar3467
- 16 Liikavainio T, Lyytinen T, Tyrväinen E, *et al.* Physical Function and Properties of Quadriceps Femoris Muscle in Men With Knee Osteoarthritis. *Arch Phys Med Rehabil.* 2008;89:2185–94. doi: 10.1016/j.apmr.2008.04.012

- 17 Birchmeier T, Lisee C, Kane K, *et al.* Quadriceps Muscle Size Following ACL Injury and Reconstruction: A Systematic Review. *Journal of Orthopaedic Research*. 2020;38:598–608. doi: 10.1002/jor.24489
- 18 Väistö O, Toivanen J, Kannus P, *et al.* Anterior knee pain and thigh muscle strength after intramedullary nailing of a tibial shaft fracture: an 8-year follow-up of 28 consecutive cases. *J Orthop Trauma*. 2007;21:165–71.
- 19 Norte G, Rush J, Sherman D. Arthrogenic Muscle Inhibition: Best Evidence, Mechanisms, and Theory for Treating the Unseen in Clinical Rehabilitation. J Sport Rehabil. 2022;31:717–35. doi: 10.1123/jsr.2021-0139
- 20 Neogi T. The Epidermiology and Impact of Pain in Osteoarthritis. *Osteoarthritis Research Society*. 2013;21:1145–53. doi: 10.1016/j.joca.2013.03.018.The
- 21 Hurley M V., Scott DL, Rees J, *et al.* Sensorimotor changes and functional performance in patients with knee osteoarthritis. *Ann Rheum Dis.* 1997;56:641–8. doi: 10.1136/ard.56.11.641
- 22 Lepley LK. Deficits in quadriceps strength and patient-oriented outcomes at return to activity after ACL reconstruction: a review of the current literature. *Sports Health*. 2015;7:231–8.
- 23 Saferding V, Blüml S. Innate immunity as the trigger of systemic autoimmune diseases. J Autoimmun. 2020;110.
- 24 Mallinson JE, Murton AJ. Mechanisms responsible for disuse muscle atrophy: potential role of protein provision and exercise as countermeasures. *Nutrition*. 2013;29:22–8.
- 25 Fingleton C, Smart K, Moloney N, *et al.* Pain sensitization in people with knee osteoarthritis: A systematic review and meta-analysis. Osteoarthritis Cartilage. 2015;23:1043–56.
- 26 Nataraj R, Audu ML, Kirsch RF, *et al.* Comprehensive Joint Feedback Control for Standing by Functional Neuromuscular Stimulation-a Simulation Study. Published Online First: 2010. doi: 10.1109/TNSRE
- 27 Lynch AD, Logerstedt DS, Axe MJ, et al. Quadriceps activation failure after anterior cruciate ligament rupture is not mediated by knee joint effusion. Journal of Orthopaedic and Sports Physical Therapy. 2012;42:502–10. doi: 10.2519/jospt.2012.3793
- 28 Monda M, Vicidomini C, Viggiano A, *et al.* Inhibition of prostaglandin synthesis reduces the induction of MyoD expression in rat soleus muscle. *J Muscle Res Cell Motil.* 2009;30:139–44.
- 29 Zhang W, Moskowitz RW, Nuki G, *et al.* OARSI recommendations for the management of hip and knee osteoarthritis, Part I: Critical appraisal of existing treatment guidelines and systematic review of current research evidence. Osteoarthritis Cartilage. 2007;15:981–1000. doi: 10.1016/j.joca.2007.06.014
- 30 Coletti C, Acosta GF, Keslacy S, *et al.* Exercise-mediated reinnervation of skeletal muscle in elderly people: An update. *Eur J Transl Myol.* 2022;32. doi: 10.4081/ejtm.2022.10416
- 31 Powers SK, Smuder AJ, Judge AR. Oxidative stress and disuse muscle atrophy: Cause or consequence? *Curr Opin Clin Nutr Metab Care*. 2012;15:240–5. doi: 10.1097/MCO.0b013e328352b4c2

- 32 Fingleton C, Smart KM, Doody CM. Exercise-induced Hypoalgesia in People with Knee Osteoarthritis with Normal and Abnormal Conditioned Pain Modulation. *Clinical Journal of Pain*. 2017;33:395–404. doi: 10.1097/AJP.0000000000000418
- Rice D, Nijs J, Kosek E, *et al.* Exercise-induced hypoalgesia in pain-free and chronic pain populations: state of the art and future directions. *J Pain*. 2019;20:1249–66.
- 34 Zhang W, Moskowitz RW, Nuki G, et al. OARSI recommendations for the management of hip and knee osteoarthritis, Part II: OARSI evidence-based, expert consensus guidelines. Osteoarthritis Cartilage. 2008;16:137–62. doi: 10.1016/j.joca.2007.12.013
- 35 Zhang W, Nuki G, Moskowitz RW, *et al.* OARSI recommendations for the management of hip and knee osteoarthritis. Part III: Changes in evidence following systematic cumulative update of research published through January 2009. Osteoarthritis Cartilage. 2010;18:476–99. doi: 10.1016/j.joca.2010.01.013
- 36 Bickel CS, Gregory CM, Dean JC. Motor unit recruitment during neuromuscular electrical stimulation: a critical appraisal. *Eur J Appl Physiol*. 2011;111:2399.
- 37 Jones S, Man WD-C, Gao W, *et al.* Neuromuscular electrical stimulation for muscle weakness in adults with advanced disease. *Cochrane Database of Systematic Reviews.* 2016;10:CD009419. doi: 10.1002/14651858.CD009419.pub3
- 38 Maffiuletti NA. Physiological and methodological considerations for the use of neuromuscular electrical stimulation. *Eur J Appl Physiol*. 2010;110:223–34. doi: 10.1007/s00421-010-1502-y
- 39 Hortobágyi T, Maffiuletti NA. Neural adaptations to electrical stimulation strength training. *Eur J Appl Physiol.* 2011;111:2439–49. doi: 10.1007/s00421-011-2012-2
- 40 Herzig D, Maffiuletti NA, Eser P. The Application of Neuromuscular Electrical Stimulation Training in Various Non-neurologic Patient Populations: A Narrative Review. *PM and R*. 2015;7:1167–78. doi: 10.1016/j.pmrj.2015.03.022
- 41 Hughes L, Paton B, Rosenblatt B, *et al.* Blood flow restriction training in clinical musculoskeletal rehabilitation: a systematic review and meta-analysis. *Br J Sports Med.* 2017;51:1003–11. doi: 10.1136/bjsports-2016-097071
- 42 Loenneke JP, Wilson JM, Marín PJ, *et al.* Low intensity blood flow restriction training: A meta-analysis. *Eur J Appl Physiol.* 2012;112:1849–59. doi: 10.1007/s00421-011-2167-x
- 43 Hughes L, Patterson SD. The effect of blood flow restriction exercise on exerciseinduced hypoalgesia and endogenous opioid and endocannabinoid mechanisms of pain modulation. *J Appl Physiol.* 2020;128:914–24.
- 44 Pearson SJ, Hussain SR. A Review on the Mechanisms of Blood-Flow Restriction Resistance Training-Induced Muscle Hypertrophy. *Sports Medicine*. 2015;45:187– 200. doi: 10.1007/s40279-014-0264-9
- 45 Gorgey AS, Timmons MK, Dolbow DR, *et al.* Electrical stimulation and blood flow restriction increase wrist extensor cross-sectional area and flow meditated dilatation following spinal cord injury. *Eur J Appl Physiol.* 2016;116:1231–44. doi: 10.1007/s00421-016-3385-z

- 46 Natsume T, Ozaki H, Saito AI, *et al.* Effects of Electrostimulation with Blood Flow Restriction on Muscle Size and Strength. *Med Sci Sports Exerc.* 2015;47:2621–7. doi: 10.1249/MSS.00000000000722
- 47 Andrade SF, Skiba GH, Krueger E, *et al.* Journal of Exercise Physiology online Effects of Electrostimulation with Blood Flow Restriction on Muscle Thickness and Strength of the Soleus. 2016;19:59–69.
- 48 Loenneke JP, Wilson JM, Bemben MG. Low intensity blood flow restriction training : a meta-analysis. 2012;1849–59. doi: 10.1007/s00421-011-2167-x
- 49 Slysz J, Stultz J, Burr JF. The efficacy of blood flow restricted exercise: A systematic review & meta-analysis. J Sci Med Sport. 2016;19:669–75.
- 50 Rudrappa SS, Wilkinson DJ, Greenhaff PL, *et al.* Human skeletal muscle disuse atrophy: Effects on muscle protein synthesis, breakdown, and insulin resistance-A qualitative review. *Front Physiol.* 2016;7:1–10. doi: 10.3389/fphys.2016.00361
- 51 Mallinson JE, Murton AJ. Mechanisms responsible for disuse muscle atrophy: Potential role of protein provision and exercise as countermeasures. *Nutrition*. 2013;29:22–8. doi: 10.1016/j.nut.2012.04.012
- 52 Ferrando AA, Lane HW, Stuart CA, *et al.* Prolonged bed rest decreases skeletal muscle and whole body protein synthesis. *American Journal of Physiology-Endocrinology And Metabolism.* 1996;270:E627–33.
- 53 Paddon-Jones D, Sheffield-Moore M, Urban RJ, *et al.* Essential amino acid and carbohydrate supplementation ameliorates muscle protein loss in humans during 28 days bedrest. *J Clin Endocrinol Metab.* 2004;89:4351–8.
- 54 Glover EI, Phillips SM, Oates BR, *et al.* Immobilization induces anabolic resistance in human myofibrillar protein synthesis with low and high dose amino acid infusion. *Journal of Physiology*. 2008;586:6049–61. doi: 10.1113/jphysiol.2008.160333
- 55 Wall BT, Loon LJC van. Nutritional strategies to attenuate muscle disuse atrophy. ;71:195–208. doi: 10.1111/nure.12019
- 56 Rice DA, McNair PJ. Quadriceps Arthrogenic Muscle Inhibition: Neural Mechanisms and Treatment Perspectives. Semin Arthritis Rheum. 2010;40:250–66.
- 57 Patterson SD, Hughes L, Warmington S, *et al.* Blood Flow Restriction Exercise Position Stand : Considerations of Methodology , Application , and Safety. 2019;10:1–15. doi: 10.3389/fphys.2019.00533
- 58 Slysz JT, Burr JF. The effects of blood flow restricted electrostimulation on strength and hypertrophy. *J Sport Rehabil*. 2018;27:257–62.
- 59 Schoenfeld BJ. The Mechanisms of Muscle Hypertrophy and Their Application to Resistance Training. *J Strength Cond Res.* 2010;24:2857–72. doi: 10.1519/JSC.0b013e3181e840f3
- 60 Doucet BM, Lam A, Griffin L. Neuromuscular electrical stimulation for skeletal muscle function. *Yale J Biol Med.* 2012;85:201–15. doi: 10.1007/BF02371250
- 61 Nakajima T, Koide S, Yasuda T, *et al.* Muscle hypertrophy following blood flowrestricted low force isometric electrical stimulation in rat tibialis anterior: Role for muscle hypoxia. *J Appl Physiol.* 2018;japplphysiol.00972.2017. doi: 10.1152/japplphysiol.00972.2017

- 62 Tanaka M, Morifuji T, Yoshikawa M, *et al.* Effects of combined treatment with blood flow restriction and low-intensity electrical stimulation on diabetes mellitusassociated muscle atrophy in rats. *J Diabetes*. 2019;11:326–34. doi: 10.1111/1753-0407.12857
- 63 Inagaki Y, Madarame H, Neya M, et al. Increase in serum growth hormone induced by electrical stimulation of muscle combined with blood flow restriction. Eur J Appl Physiol. 2011;111:2715–21. doi: 10.1007/s00421-011-1899-y
- 64 Natsume T, Yoshihara T, Naito H. Electromyostimulation with Blood Flow Restriction Enhances Activation of mTOR and MAPK Signalling Pathways in Rat Gastrocnemius Muscles. *The FASEB Journal*. 2018;32:Ib46–Ib46.
- 65 Wüst RCI, Morse CI, De Haan A, *et al.* Sex differences in contractile properties and fatigue resistance of human skeletal muscle. *Exp Physiol.* 2008;93:843–50. doi: 10.1113/expphysiol.2007.041764
- 66 Yoshikawa M, Morifuji T, Matsumoto T, *et al.* Effects of combined treatment with blood flow restriction and low current electrical stimulation on muscle hypertrophy in rats. *J Appl Physiol.* 2019.
- 67 Anthony TG. Mechanisms of protein balance in skeletal muscle. *Domest Anim Endocrinol.* 2016;56:S23–32.
- 68 Urso ML. Disuse atrophy of human skeletal muscle: cell signaling and potential interventions. *Med Sci Sports Exerc*. 2009;41:1860–8.
- 69 Skiba GH, Andrade SF, Rodacki AF. Effects of functional electro-stimulation combined with blood flow restriction in affected muscles by spinal cord injury. *Neurological Sciences*. 2022;43:603–13. doi: 10.1007/s10072-021-05307-x
- 70 Slysz JT, Boston M, King R, *et al.* Blood Flow Restriction Combined with Electrical Stimulation Attenuates Thigh Muscle Disuse Atrophy. *Med Sci Sports Exerc.* 2021;53:1033–40. doi: 10.1249/MSS.00000000002544
- 71 Natsume T, Yoshihara T, Naito H. Electromyostimulation with blood flow restriction enhances activation of mTOR and MAPK signaling pathways in rat gastrocnemius muscles. *Applied Physiology, Nutrition, and Metabolism.* 2018;44:637–44.
- 72 Murthy G, Hargens AR, Lehman S, *et al.* Ischemia causes muscle fatigue. *Journal of Orthopaedic Research*. 2001;19:436–40. doi: 10.1016/S0736-0266(00)90019-6
- 73 Cole MA, Brown MD. Response of the human triceps surae muscle to electrical stimulation during varying levels of blood flow restriction. *Eur J Appl Physiol.* 2000;82:39–44. doi: 10.1007/s004210050649
- 74 Santiago-Pescador S, Fajardo-Blanco D, López-Ortiz S, *et al.* Acute effects of electrostimulation and blood flow restriction on muscle thickness and fatigue in the lower body. *Eur J Sport Sci.* Published Online First: 2022. doi: 10.1080/17461391.2022.2113145
- 75 Hodson-Tole EF, Wakeling JM. Motor unit recruitment for dynamic tasks: Current understanding and future directions. J Comp Physiol B. 2009;179:57–66.
- 76 Henneman E. THE SIZE-PRINCIPLE: A DETERMINISTIC OUTPUT EMERGES FROM A SET OF PROBABILISTIC CONNECTIONS. 1985.
- 77 Ruther CL, Golden CL, Harris RT, *et al.* Hypertrophy, resistance training, and the nature of skeletal muscle activation. *The Journal of Strength & Conditioning Research.* 1995;9:155–9.

- 78 Hodson-Tole EF, Wakeling JM. Motor unit recruitment for dynamic tasks: current understanding and future directions. *Journal of Comparative Physiology B*. 2009;179:57–66.
- Gregory CM, Bickel CS. Recruitment Patterns in Human. *Phys Ther.* 2005;85:358–64.
- 80 Jubeau M, Zory R, Gondin J, *et al.* Effect of electrostimulation training-detraining on neuromuscular fatigue mechanisms. *Neurosci Lett.* 2007;424:41–6. doi: 10.1016/j.neulet.2007.07.018
- Li N, Yang J, Jiang N, *et al.* The Effect of Blood Flow Restriction Training Combined with Electrical Muscle Stimulation on Neuromuscular Adaptation: A Randomized Controlled Trial. Published Online First: 2022. doi: 10.21203/rs.3.rs-1690744/v1
- Asmussen E. Muscle fatigue. June 1979. *Med Sci Sports Exerc*. 1993;25:411.
- 83 Jessee MB, Mattocks KT, Buckner SL, *et al.* Mechanisms of Blood Flow Restriction : The New Testament. 2018;00:1–8.
- 84 Natsume T, Ozaki H, Kakigi R, *et al.* Effects of training intensity in electromyostimulation on human skeletal muscle. *Eur J Appl Physiol.* 2018;118:1339–47.
- 85 Drummond MJ, Fujita S, Takashi A, *et al.* Human muscle gene expression following resistance exercise and blood flow restriction. *Med Sci Sports Exerc.* 2008;40:691.
- 86 Jacinto E, Hall MN. Tor signalling in bugs, brain and brawn. *Nat Rev Mol Cell Biol.* 2003;4:117–26.
- 87 Sarbassov DD, Ali SM, Sabatini DM. Growing roles for the mTOR pathway. *Curr Opin Cell Biol.* 2005;17:596–603.
- 88 Thijssen DHJ, Black MA, Pyke KE, *et al.* Assessment of flow-mediated dilation in humans: a methodological and physiological guideline. *American Journal of Physiology-Heart and Circulatory Physiology.* 2011;300:H2–12.
- 89 Hunt JEA, Galea D, Tufft G, et al. Time course of regional vascular adaptations to low load resistance training with blood flow restriction. J Appl Physiol. 2013;115:403–11.
- 90 Stoner L, Sabatier MJ, Mahoney ET, *et al.* Electrical stimulation-evoked resistance exercise therapy improves arterial health after chronic spinal cord injury. *Spinal Cord.* 2007;45:49–56.
- 91 Pérez M, Lucia A, Rivero J-L, *et al.* Effects of transcutaneous short-term electrical stimulation on M. vastus lateralis characteristics of healthy young men. *Pflügers Archiv.* 2002;443:866–74.
- 92 Gondin J, Brocca L, Bellinzona E, *et al.* Neuromuscular electrical stimulation training induces atypical adaptations of the human skeletal muscle phenotype: a functional and proteomic analysis. *J Appl Physiol.* 2011;110:433–50.
- 93 Cohen JN, Slysz JT, King TJ, *et al.* Blood flow restriction in the presence or absence of muscle contractions does not preserve vasculature structure and function following 14–days of limb immobilization. *Eur J Appl Physiol.* 2021;121:2437–47. doi: 10.1007/s00421-021-04715-3
- 94 Rantanen T. Muscle strength, disability and mortality. *Scand J Med Sci Sports*. 2003;13:3–8.

- 95 Yoshikawa M, Morifuji T, Matsumoto T, *et al.* Effects of combined treatment with blood flow restriction and low-current electrical stimulation on muscle hypertrophy in rats. *J Appl Physiol.* 2019;127:1288–96. doi: 10.1152/japplphysiol.00070.2019.-This
- 96 Tanaka M, Morifuji T, Sugimoto K, et al. Effects of combined treatment with blood flow restriction and low-current electrical stimulation on capillary regression in the soleus muscle of diabetic rats. J Appl Physiol. 2021;131:1219–29. doi: 10.1152/japplphysiol.00366.2021
- 97 Lyons GM, Leane GE, Clarke-Moloney M, *et al.* An investigation of the effect of electrode size and electrode location on comfort during stimulation of the gastrocnemius muscle. *Med Eng Phys.* 2004;26:873–8.
- 98 Botter A, Oprandi G, Lanfranco F, et al. Atlas of the muscle motor points for the lower limb: Implications for electrical stimulation procedures and electrode positioning. Eur J Appl Physiol. 2011;111:2461–71. doi: 10.1007/s00421-011-2093y
- 99 Gorgey AS, Timmons MK, Dolbow DR, et al. Electrical stimulation and blood flow restriction increase wrist extensor cross-sectional area and flow meditated dilatation following spinal cord injury. Eur J Appl Physiol. 2016;116:1231–44. doi: 10.1007/s00421-016-3385-z
- Slysz J, Stultz J, Burr JF. Journal of Science and Medicine in Sport The efficacy of blood flow restricted exercise : A systematic review &. J Sci Med Sport. 2016;19:669–75. doi: 10.1016/j.jsams.2015.09.005
- 101 Maffiuletti NA, Roig M, Karatzanos E, *et al.* Neuromuscular electrical stimulation for preventing skeletal-muscle weakness and wasting in critically ill patients: a systematic review. *BMC Med.* 2013;11:137. doi: 10.1186/1741-7015-11-137
- 102 Slysz JT, Island PE. the Physiological Effects of Blood Flow Restricted Muscle. 2015.
- 103 Smart NA, Waldron M, Ismail H, *et al.* Validation of a new tool for the assessment of study quality and reporting in exercise training studies: TESTEX. *Int J Evid Based Healthc.* 2015;13.
- 104 Oakman A, Zhou S, Davie A. Cross-education effect observed in voluntary and electromyostimulation strength training. *17 International Symposium on Biomechanics in Sports*. 1999;401–4.
- 105 Kamen G. Neural issues in the control of muscular strength. *Res Q Exerc Sport*. 2004;75:3–8. doi: 10.1080/02701367.2004.10609127
- 106 Moritani T, DeVries HA. Potential for gross muscle hypertrophy in older men. *Journals of Gerontology*. 1980;35:672–82. doi: 10.1093/geronj/35.5.672
- 107 Madarame H, Neya M, Ochi E, et al. Cross-transfer effects of resistance training with blood flow restriction. Med Sci Sports Exerc. 2008;40:258–63. doi: 10.1249/mss.0b013e31815c6d7e
- 108 Brandner CR, Warmington SA, Kidgell DJ. Corticomotor Excitability is Increased Following an Acute Bout of Blood Flow Restriction Resistance Exercise. *Front Hum Neurosci.* 2015;9:1–10. doi: 10.3389/fnhum.2015.00652
- 109 Song Y, Forsgren S, Yu J, et al. Effects on Contralateral Muscles after Unilateral Electrical Muscle Stimulation and Exercise. PLoS One. 2012;7. doi: 10.1371/journal.pone.0052230

- 110 Yoshida Y. Anatomical Studies on the Extensor Carpi Radialis Longus and Brevis Muscles in Japanese. 1994.
- 111 Bao X, Zhou Y, Wang Y, *et al.* Electrode placement on the forearm for selective stimulation of finger extension/flexion. *PLoS One*. 2018;13.
- Bergamasco JGA, Alvarez IF, de Campos Biazon TMP, *et al.* Effects of Blood Flow Restriction Combined With Resistance Training or Neuromuscular Electrostimulation on Muscle Cross-Sectional Area. *J Sport Rehabil.* 2022;31:319– 24. doi: 10.1123/jsr.2021-0101
- 113 Lim C, Nunes EA, Currier BS, *et al.* An Evidence-Based Narrative Review of Mechanisms of Resistance Exercise-Induced Human Skeletal Muscle Hypertrophy. *Med Sci Sports Exerc.* 2022;54:1546–59. doi: 10.1249/MSS.00000000002929
- 114 Slysz J, Burr JF. The Effects of Blood Flow Restricted Electrostimulation on Strength & Hypertrophy. *J Sport Rehabil.* 2017;1–17.
- 115 Mouser JG, Dankel SJ, Jessee MB, *et al.* A tale of three cuffs: the hemodynamics of blood flow restriction. *Eur J Appl Physiol.* 2017;117:1493–9. doi: 10.1007/s00421-017-3644-7
- 116 Hughes L, Jeffries O, Waldron M, et al. Influence and reliability of lower-limb arterial occlusion pressure at different body positions. *PeerJ*. 2018;6:e4697. doi: 10.7717/peerj.4697
- 117 Patterson SD, Hughes L, Head P, *et al.* Blood flow restriction training : a novel approach to augment clinical rehabilitation : how to do it. 2017;2016–8. doi: 10.1136/bjsports-2017-097738
- 118 Glaviano NR, Saliba S. Can the Use of Neuromuscular Electrical Stimulation Be Improved to Optimize Quadriceps Strengthening? *Sports Health: A Multidisciplinary Approach.* 2016;8:79–85. doi: 10.1177/1941738115618174
- 119 Maffiuletti NA, Herrero AJ, Jubeau M, *et al.* Differences in electrical stimulation thresholds between men and women. *Ann Neurol.* 2008;63:507–12.
- 120 Laufer Y, Shtraker H, Gabyzon ME. The effects of exercise and neuromuscular electrical stimulation in subjects with knee osteoarthritis: A 3-month follow-up study. *Clin Interv Aging*. 2014;9:1153–61. doi: 10.2147/CIA.S64104
- 121 Delitto A, Rose SJ. Comparative comfort of three waveforms used in electrically eliciting quadriceps femoris muscle contractions. *Phys Ther.* 1986;66:1704–7.
- 122 Alon G, Smith G V. Tolerance and conditioning to neuro-muscular electrical stimulation within and between sessions and gender. *J Sports Sci Med.* 2005;4:395.
- 123 Spector P, Laufer Y, Elboim Gabyzon M, et al. Neuromuscular Electrical Stimulation Therapy to Restore Quadriceps Muscle Function in Patients After Orthopaedic Surgery. J Bone Joint Surg. 2016;98:2017–24. doi: 10.2106/JBJS.16.00192
- 124 Martín-Hernández J, Santos-Lozano A, Foster C, et al. Syncope Episodes and Blood Flow Restriction Training. *Clin J Sport Med.* 2017;0:1. doi: 10.1097/JSM.00000000000496
- 125 Iwashita H. A pediatric case with an unstabilized neck treated with skeletal muscle electrical stimulation and KAATSU training®. *International Journal of KAATSU Training Research*. 2015;11:7–12.

- 126 Nussbaum EL, Houghton P, Anthony J, *et al.* Neuromuscular electrical stimulation for treatment of muscle impairment: Critical review and recommendations for clinical practice. *Physiotherapy Canada*. 2017;69:1–76. doi: 10.3138/ptc.2015-88
- 127 Gracanin F, Trnkoczy A. Optimal stimulus parameters for minimum pain in the chronic stimulation of innervated muscle. *Arch Phys Med Rehabil.* 1975;56:243–9.
- 128 McLoda TA, Carmack JA. Optimal burst duration during a facilitated quadriceps femoris contraction. *J Athl Train*. 2000;35:145.
- 129 Lagerquist O, Collins DF. Influence of stimulus pulse width on M-waves, H-reflexes, and torque during tetanic low-intensity neuromuscular stimulation. *Muscle Nerve*. 2010;42:886–93.
- 130 Bracciano AG. *Physical agent modalities: Theory and application for the occupational therapist.* Slack Incorporated 2008.
- 131 Kesar T, Binder-Macleod S. Effect of frequency and pulse duration on human muscle fatigue during repetitive electrical stimulation. *Exp Physiol.* 2006;91:967–76.
- 132 Gorgey AS, Mahoney E, Kendall T, *et al.* Effects of neuromuscular electrical stimulation parameters on specific tension. *Eur J Appl Physiol.* 2006;97:737–44.
- 133 Burd NA, Andrews RJ, West DWD, *et al.* Muscle time under tension during resistance exercise stimulates differential muscle protein sub-fractional synthetic responses in men. *J Physiol.* 2012;590:351–62.
- 134 Ratamess N, Alvar B, Evetoch T, *et al.* Progression models in resistance training for healthy adults [ACSM position stand]. *Med Sci Sports Exerc.* 2009;41:687–708.
- 135 Glaviano NR, Huntsman S, Dembeck A, *et al.* Improvements in kinematics, muscle activity and pain during functional tasks in females with patellofemoral pain following a single patterned electrical stimulation treatment. *Clinical Biomechanics*. 2016;32:20–7. doi: 10.1016/j.clinbiomech.2015.12.007
- 136 Peltonen H, Walker S, Lähitie A, *et al.* Isometric parameters in the monitoring of maximal strength, power, and hypertrophic resistance-training. *Applied Physiology, Nutrition, and Metabolism.* 2017;43:145–53.
- 137 Fisher JP, Blossom D, Steele J. A comparison of volume-equated knee extensions to failure, or not to failure, upon rating of perceived exertion and strength adaptations. *Applied Physiology, Nutrition, and Metabolism.* 2015;41:168–74.
- 138 Adams GR, Harris RT, Woodard D, *et al.* Mapping of electrical muscle stimulation using MRI. *J Appl Physiol.* 1993;74:532–7.
- 139 Mesin L, Merlo E, Merletti R, *et al.* Investigation of motor unit recruitment during stimulated contractions of tibialis anterior muscle. *Journal of Electromyography and Kinesiology*. 2010;20:580–9.
- 140 Maffiuletti NA, Pensini M, Martin A. Activation of human plantar flexor muscles increases after electromyostimulation training. *J Appl Physiol*. 2002;92:1383–92.
- 141 Stevens-Lapsley JE, Balter JE, Wolfe P, et al. Early Neuromuscular Electrical Stimulation to Improve Quadriceps Muscle Strength After Total Knee Arthroplasty: A Randomized Controlled Trial. Phys Ther. 2012;92:210–26. doi: 10.2522/ptj.20110124

- 142 Piva SR, Goodnite EA, Azuma K, *et al.* Neuromuscular electrical stimulation and volitional exercise for individuals with rheumatoid arthritis: a multiple-patient case report. *Phys Ther.* 2007;87:1064–77.
- 143 Baker LL, Wederich C, McNeal DR, *et al. Neuro muscular electrical stimulation: a practical guide*. Los Amigos Research & Education Institute 2000.
- 144 Gondin J, GUETTE M, BALLAY Y, *et al.* Electromyostimulation training effects on neural drive and muscle architecture. *Med Sci Sports Exerc.* 2005;37:1291–9.
- 145 STEVENSON SW, DUDLEY GA. Dietary creatine supplementation and muscular adaptation to resistive overload. *Med Sci Sports Exerc*. 2001;33:1304–10.
- 146 Vieira TM, Potenza P, Gastaldi L, *et al.* Electrode position markedly affects knee torque in tetanic, stimulated contractions. *Eur J Appl Physiol.* 2016;116:335–42.
- 147 Gobbo M, Maffiuletti NA, Orizio C, *et al.* Muscle motor point identification is essential for optimizing neuromuscular electrical stimulation use. J Neuroeng Rehabil. 2014;11.
- 148 Gobbo M, Gaffurini P, Bissolotti L, et al. Transcutaneous neuromuscular electrical stimulation: influence of electrode positioning and stimulus amplitude settings on muscle response. Eur J Appl Physiol. 2011;111:2451–9.
- 149 Loenneke JP, Fahs CA, Rossow LM, *et al.* Effects of cuff width on arterial occlusion: Implications for blood flow restricted exercise. *Eur J Appl Physiol.* 2012;112:2903–12. doi: 10.1007/s00421-011-2266-8
- 150 Hunt JEA, Stodart C, Ferguson RA. The influence of participant characteristics on the relationship between cuff pressure and level of blood flow restriction. *Eur J Appl Physiol.* 2016;116:1421–32. doi: 10.1007/s00421-016-3399-6
- 151 Laurentino GC, Loenneke JP, Mouser JG, *et al.* Validity of the handheld Doppler to determine lower-limb blood flow restriction pressure for exercise protocols. *The Journal of Strength & Conditioning Research.* 2020;34:2693–6.
- 152 Lima-Soares F, Pessoa KA, Cabido CET, *et al.* Determining the Arterial Occlusion Pressure for Blood Flow Restriction: Pulse Oximeter as a New Method Compared With a Handheld Doppler. *The Journal of Strength & Conditioning Research.* 2020.
- 153 Hughes L, Rosenblatt B, Gissane C, *et al.* Interface pressure, perceptual, and mean arterial pressure responses to different blood flow restriction systems. *Scand J Med Sci Sports.* 2018;28:1757–65.
- Reis JF, Fatela P, Mendonca G V., *et al.* Tissue oxygenation in response to different relative levels of blood-flow restricted exercise. *Front Physiol.* 2019;10:1–9. doi: 10.3389/fphys.2019.00407
- 155 Noordin S, McEwen JA, Kragh CJF, et al. Surgical Tourniquets in Orthopaedics. The Journal of Bone and Joint Surgery-American Volume. 2009;91:2958–67. doi: 10.2106/JBJS.I.00634
- 156 Ochoa J, Fowler TJ, Gilliatt RW. Anatomical changes in peripheral nerves compressed by a pneumatic tourniquet. *J Anat*. 1972;113:433–55.
- 157 Loenneke JP, Wilson JM, Wilson GJ, et al. Potential safety issues with blood flow restriction training. Scand J Med Sci Sports. 2011;21:510–8. doi: 10.1111/j.1600-0838.2010.01290.x

- 158 Loenneke JP, Thiebaud RS, Abe T. Does blood flow restriction result in skeletal muscle damage? A critical review of available evidence. *Scand J Med Sci Sports*. 2014;24:e415-422. doi: 10.1111/sms.12210
- Nakajima T, Kurano M, Iida H, et al. Use and safety of KAATSU training:Results of a national survey. International Journal of KAATSU Training Research. 2006;2:5– 13. doi: 10.3806/ijktr.2.5
- 160 TABATA S. Rhabdomyolysis After Performing Blood Flow Restriction Training:Acase Report. 2016.
- 161 Iversen E, Rstad V. Low-load ischemic exercise-Induced rhabdomyolysis. *Clinical Journal of Sport Medicine*. 2010;20:218–9. doi: 10.1097/JSM.0b013e3181df8d10
- 162 Clark BC, Manini TM. Can KAATSU Exercise Cause Rhabdomyolysis? *Clinical journal of sport medicine*. 2016;0:10–1. doi: 10.1097/JSM.000000000000309
- 163 Huerta-Alardín AL, Varon J, Marik PE. Bench-to-bedside review: Rhabdomyolysis -an overview for clinicians. *Crit Care*. 2005;9:158–69. doi: 10.1186/cc2978
- 164 Clark BC, Manini TM. Can KAATSU exercise cause rhabdomyolysis? *Clinical Journal of Sport Medicine*. 2017;27:e1–2. doi: 10.1097/JSM.0000000000000309
- 165 Spranger MD, Krishnan AC, Levy PD, et al. Blood flow restriction training and the exercise pressor reflex: a call for concern. Am J Physiol Heart Circ Physiol. 2015;309:H1440–52. doi: 10.1152/ajpheart.00208.2015
- 166 Gorgey AS, Dolbow DR, Cifu DX, *et al.* Neuromuscular electrical stimulation attenuates thigh skeletal muscles atrophy but not trunk muscles after spinal cord injury. *Journal of Electromyography and Kinesiology*. 2013;23:977–84. doi: 10.1016/j.jelekin.2013.04.007
- 167 Yam MF, Loh YC, Tan CS, *et al.* General pathways of pain sensation and the major neurotransmitters involved in pain regulation. Int J Mol Sci. 2018;19.
- 168 Deogaonkar M, Slavin K V. Peripheral nerve/field stimulation for neuropathic pain. *Neurosurgery Clinics*. 2014;25:1–10.
- 169 Loenneke JP, Balapur A, Thrower AD, *et al.* The perceptual responses to occluded exercise. *Int J Sports Med.* 2011;32:181–4. doi: 10.1055/s-0030-1268472
- 170 Loenneke JP, Kim D, Mouser JG, et al. Are there perceptual differences to varying levels of blood flow restriction? *Physiol Behav.* 2016;157:277–80. doi: 10.1016/j.physbeh.2016.02.022
- 171 Karanasios S, Lignos I, Kouvaras K, *et al.* Low-Intensity Blood Flow Restriction Exercises Modulate Pain Sensitivity in Healthy Adults: A Systematic Review. *Healthcare (Basel).* 2023;11. doi: 10.3390/healthcare11050726
- 172 Koltyn KF. Analgesia following exercise. Sports medicine. 2000;29:85–98.
- 173 Naugle KM, Fillingim RB, Riley III JL. A meta-analytic review of the hypoalgesic effects of exercise. *J Pain*. 2012;13:1139–50.
- 174 Vaegter HB, Handberg G, Graven-Nielsen T. Hypoalgesia after exercise and the cold pressor test is reduced in chronic musculoskeletal pain patients with high pain sensitivity. *Clin J Pain*. 2016;32:58–69.

- 175 Vaegter HB, Lyng KD, Yttereng FW, et al. Exercise-induced hypoalgesia after isometric wall squat exercise: a test-retest reliability study. *Pain Medicine*. 2019;20:129–37.
- 176 King CD, Sibille KT, Goodin BR, *et al.* Experimental pain sensitivity differs as a function of clinical pain severity in symptomatic knee osteoarthritis. *Osteoarthritis Cartilage*. 2013;21:1243–52. doi: 10.1016/j.joca.2013.05.015
- 177 Eisenberg E, Waisbrod H, Gerbershagen HU. Long-term peripheral nerve stimulation for painful nerve injuries. *Clin J Pain*. 2004;20:143–6.
- 178 Korakakis V, Whiteley R, Epameinontidis K. Blood flow restriction-induced analgesia in patients with anterior knee pain. *J Sci Med Sport*. 2017;20:e100.
- 179 Giles L, Webster KE, McClelland J, *et al.* Quadriceps strengthening with and without blood flow restriction in the treatment of patellofemoral pain: a double-blind randomised trial. *Br J Sports Med.* 2017;bjsports-2016-096329. doi: 10.1136/bjsports-2016-096329
- 180 Ferraz RB, Gualano B, Rodrigues R, et al. Benefits of Resistance Training with Blood Flow Restriction in Knee Osteoarthritis. *Med Sci Sports Exerc.* 2018;50:897– 905. doi: 10.1249/MSS.00000000001530
- 181 Hughes L, Patterson SD, Haddad F, et al. Examination of the comfort and pain experienced with blood flow restriction training during post-surgery rehabilitation of anterior cruciate ligament reconstruction patients: A UK National Health Service trial. *Physical Therapy in Sport*. 2019;39:90–8. doi: 10.1016/j.ptsp.2019.06.014
- 182 Korakakis V, Whiteley R, Giakas G. Physical Therapy in Sport Low load resistance training with blood fl ow restriction decreases anterior knee pain more than resistance training alone . A pilot randomised controlled trial. *Physical Therapy in Sport*. 2018;34:121–8. doi: 10.1016/j.ptsp.2018.09.007
- 183 Korakakis V, Whiteley R, Epameinontidis K. Physical Therapy in Sport Blood Flow Restriction induces hypoalgesia in recreationally active adult male anterior knee pain patients allowing therapeutic exercise loading. *Physical Therapy in Sport*. 2018;32:235–43. doi: 10.1016/j.ptsp.2018.05.021
- 184 Hughes L, Patterson SD. Low intensity blood flow restriction exercise: Rationale for a hypoalgesia effect. *Med Hypotheses*. 2019;132:109370.
- 185 Ozaki H, Natsume T, Abe T, et al. Effect Of Neuromuscular Electrical Stimulation With Blood Flow Restriction On Muscle Size And Strength. Med Sci Sports Exerc. 2015;47:536. doi: 10.1249/01.mss.0000478163.96038.fb
- 186 Head P, Waldron M, Theis N, *et al.* Acute Neuromuscular Electrical Stimulation (NMES) with Blood Flow Restriction: The Effect of Restriction Pressures. *J Sport Rehabil.* 2021;30:375–83. doi: 10.1123/JSR.2019-0505
- 187 Takarada Y, Takazawa H, Sato Y, *et al.* Effects of resistance exercise combined with moderate vascular occlusion on muscular function in humans Effects of resistance exercise combined with moderate vascular occlusion on muscular function in humans. *J Appl Physiol.* 2000;88:2097–106.
- 188 Kubota A, Sakuraba K, Sawaki K, et al. Prevention of disuse muscular weakness by restriction of blood flow. *Med Sci Sports Exerc*. 2008;40:529–34. doi: 10.1249/MSS.0b013e31815ddac6

- 189 Kubota A, Sakuraba K, Koh S, *et al.* Blood flow restriction by low compressive force prevents disuse muscular weakness. *J Sci Med Sport.* 2011;14:95–9. doi: 10.1016/j.jsams.2010.08.007
- 190 Iversen E, Røstad V, Larmo A. Intermittent blood flow restriction does not reduce atrophy following anterior cruciate ligament reconstruction. J Sport Health Sci. 2016;5:115–8. doi: 10.1016/j.jshs.2014.12.005
- 191 Dirks ML, Wall BT, Snijders T, et al. Neuromuscular electrical stimulation prevents muscle disuse atrophy during leg immobilization in humans. Acta Physiologica. 2014;210:628–41. doi: 10.1111/apha.12200
- 192 Gibson JNA, Smith K, Rennie MJ. Prevention of disuse muscle atrophy by means of electrical stimulation: maintenance of protein synthesis. *The Lancet*. 1988;332:767–70.
- 193 De Oliveira Melo M, Aragão FA, Vaz MA. Neuromuscular electrical stimulation for muscle strengthening in elderly with knee osteoarthritis A systematic review. *Complement Ther Clin Pract.* 2013;19:27–31. doi: 10.1016/j.ctcp.2012.09.002
- 194 Aboyans V, Criqui MH, Abraham P, *et al.* Measurement and interpretation of the ankle-brachial index: a scientific statement from the American Heart Association. *Circulation.* 2012;126:2890–909.
- 195 Hopkins WG. Measures of reliability in sports medicine and science. *Sports Medicine*. 2000;30:1–15. doi: 10.2165/00007256-200030010-00001
- 196 Currell K, Jeukendrup AE. Validity, Reliability and Sensitivity of Measures of Sporting Performance. 2008.
- 197 Lake DA. Neuromuscular electrical stimulation. Sports medicine. 1992;13:320–36.
- 198 Glaviano NR, Saliba S. Can the Use of Neuromuscular Electrical Stimulation Be Improved to Optimize Quadriceps Strengthening? *Sports Health: A Multidisciplinary Approach.* 2016;8:79–85. doi: 10.1177/1941738115618174
- 199 Breno J, Ribeiro DA, Rodrigues R, *et al.* Physical Therapy in Sport Inter-machine reliability of the Biodex and Cybex isokinetic dynamometers for knee fl exor / extensor isometric , concentric and eccentric tests. *Physical Therapy in Sport*. 2015;16:59–65. doi: 10.1016/j.ptsp.2014.04.004
- 200 Cavalcante JGT, Marqueti R de C, Geremia JM, *et al.* The Effect of Quadriceps Muscle Length on Maximum Neuromuscular Electrical Stimulation Evoked Contraction, Muscle Architecture, and Tendon-Aponeurosis Stiffness. *Front Physiol.* 2021;12. doi: 10.3389/fphys.2021.633589
- 201 Maayah MF, Al-Jarrah MD, Zahrani SS el, et al. Test-retest strength reliability of the Electronic Push/Pull Dynamometer (EPPD) in the measurement of the quadriceps and hamstring muscles on a new chair. Open Journal of Internal Medicine. 2012;2:123–8. doi: 10.4236/ojim.2012.22022
- 202 Head P, Austen B, Browne D, *et al.* Effect of practical blood flow restriction training during bodyweight exercise on muscular strength, hypertrophy and function in adults: A randomised controlled trial. *Int J Ther Rehabil.* 2015;22:263–71.
- 203 Loenneke JP, Fahs CA, Rossow LM, *et al.* The anabolic benefits of venous blood flow restriction training may be induced by muscle cell swelling. *Med Hypotheses*. 2012;78:151–4. doi: 10.1016/j.mehy.2011.10.014

- Lauber B, König D, Gollhofer A, *et al.* Isometric blood flow restriction exercise:
 acute physiological and neuromuscular responses. *BMC Sports Sci Med Rehabil.* 2021;13:1–9. doi: 10.1186/s13102-021-00239-7
- 205 Reeves ND, Maganaris CN, Narici M v. Ultrasonographic assessment of human skeletal muscle size. *Eur J Appl Physiol*. 2004;91:116–8.
- 206 Lixandrão ME, Ugrinowitsch C, Bottaro M, et al. Vastus lateralis muscle crosssectional area ultrasonography validity for image fitting in humans. The Journal of Strength & Conditioning Research. 2014;28:3293–7.
- 207 Timmins RG, Shield AJ, Williams MD, *et al.* Architectural adaptations of muscle to training and injury: A narrative review outlining the contributions by fascicle length, pennation angle and muscle thickness. Br J Sports Med. 2016;50:1467–72.
- 208 Hughes L, Rosenblatt B, Haddad F, et al. Comparing the Effectiveness of Blood Flow Restriction and Traditional Heavy Load Resistance Training in the Post-Surgery Rehabilitation of Anterior Cruciate Ligament Reconstruction Patients: A UK National Health Service Randomised Controlled Trial. Sports Medicine. 2019;49:1787–805. doi: 10.1007/s40279-019-01137-2
- 209 Muntner P, Shimbo D, Carey RM, et al. Measurement of blood pressure in humans: A scientific statement from the american heart association. *Hypertension*. 2019;73:E35–66. doi: 10.1161/HYP.000000000000087
- 210 Pickering TG. Blood pressure measurement and detection of hypertension. *The lancet.* 1994;344:31–5.
- 211 Seravalle G, Grassi G. Heart rate as cardiovascular risk factor. Postgrad Med. 2020;132:358–67.
- 212 Taylor K, Chapman D, Cronin J, *et al.* Fatigue monitoring in high performance sport: a survey of current trends. *J Aust Strength Cond.* 2012;20:12–23.
- 213 Fitzpatrick JF, Hicks KM, Russell M, et al. The Reliability of Potential Fatigue-Monitoring Measures in Elite Youth Soccer Players. J Strength Cond Res. 2021;35:3448–52. doi: 10.1519/JSC.00000000003317
- 214 Koo TK, Li MY. A Guideline of Selecting and Reporting Intraclass Correlation Coefficients for Reliability Research. J Chiropr Med. 2016;15:155–63. doi: 10.1016/j.jcm.2016.02.012
- 215 Borg GA. Psychophysical bases of perceived exertion. *Med sci sports exerc*. 1982;14:377–81.
- 216 Lamb KL, Eston RG, Corns D. Reliability of ratings of perceived exertion during progressive treadmill exercise.
- 217 Rossow LM, Fahs CA, Loenneke JP, et al. Cardiovascular and perceptual responses to blood-flow-restricted resistance exercise with differing restrictive cuffs. *Clin Physiol Funct Imaging*. 2012;32:331–7. doi: 10.1111/j.1475-097X.2012.01131.x
- 218 Ferreira-Valente MA, Pais-Ribeiro JL, Jensen MP. Validity of four pain intensity rating scales. *Pain*. 2011;152:2399–404.
- 219 Hawker GA, Mian S, Kendzerska T, *et al.* Measures of adult pain: Visual analog scale for pain (vas pain), numeric rating scale for pain (nrs pain), mcgill pain questionnaire (mpq), short-form mcgill pain questionnaire (sf-mpq), chronic pain

grade scale (cpgs), short form-36 bodily pain scale (sf. *Arthritis Care Res (Hoboken)*. 2011;63.

- 220 Poole DC, Burnley M, Vanhatalo A, et al. Critical power: An important fatigue threshold in exercise physiology. Med Sci Sports Exerc. 2016;48:2320–34. doi: 10.1249/MSS.00000000000939
- 221 Robertson GAJ, Wood AM. Return to Sport After Tibial Shaft Fractures: A Systematic Review. Sports Health. 2016;8:324–30. doi: 10.1177/1941738115601425
- 222 Williams GN, Snyder-Mackler L, Barrance PJ, *et al.* Quadriceps femoris muscle morphology and function after ACL injury: a differential response in copers versus non-copers. *J Biomech*. 2005;38:685–93.
- 223 Jones AM, Vanhatalo A, Burnley M, et al. Critical power: implications for determination of VO2max and exercise tolerance. *Med Sci Sports Exerc.* 2010;42:1876–90.
- Burnley M. Estimation of critical torque using intermittent isometric maximal voluntary contractions of the quadriceps in humans. *J Appl Physiol.* 2009;106:975–83.
- 225 de Menezes Bassan N, Denadai BS, de Lima LCR, *et al.* Effects of resistance training on impulse above end-test torque and muscle fatigue. *Exp Physiol.* 2019;104:1115–25.
- 226 Pethick J, Winter SL, Burnley M. Fatigue reduces the complexity of knee extensor torque fluctuations during maximal and submaximal intermittent isometric contractions in man. *Journal of Physiology*. 2015;593:2085–96. doi: 10.1113/jphysiol.2015.284380
- 227 Herbert RD, Gandevia SC. Twitch Interpolation in Human Muscles : Mechanisms and Implications for Measurement of Voluntary Activation Twitch Interpolation in Human Muscles : Mechanisms and Implications for Measurement of Voluntary Activation. *J Physiol.* 1999;82:2271–83. doi: 10.1152/jn.1999.82.5.2271
- 228 Harkey MS, Gribble PA, Pietrosimone BG. Disinhibitory interventions and voluntary quadriceps activation: A systematic review. J Athl Train. 2014;49:411–21.
- 229 Beltman JGM, Sargeant AJ, Van Mechelen W, *et al.* Voluntary activation level and muscle fiber recruitment of human quadriceps during lengthening contractions. *J Appl Physiol.* 2004;97:619–26.
- 230 Broxterman RM, Craig JC, Smith JR, *et al.* Influence of blood flow occlusion on the development of peripheral and central fatigue during small muscle mass handgrip exercise. *J Physiol.* 2015;593:4043–54.
- 231 Place N, Maffiuletti NA, Martin A, et al. Assessment of the reliability of central and peripheral fatigue after sustained maximal voluntary contraction of the quadriceps muscle. *Muscle Nerve*. 2007;35:486–95. doi: 10.1002/mus.20714
- 232 Allen GM, Gandevia SC, Mckenzie DK. RELIABILITY OF MEASUREMENTS OF MUSCLE STRENGTH AND VOLUNTARY ACTIVATION USING TWITCH INTERPOLATION.
- 233 Ferrari M, Muthalib M, Quaresima V. The use of near-infrared spectroscopy in understanding skeletal muscle physiology: Recent developments. *Philosophical*

Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences. 2011;369:4577–90. doi: 10.1098/rsta.2011.0230

- 234 Jeffries O, Waldron M, Pattison JR, *et al.* Enhanced local skeletal muscle oxidative capacity and microvascular blood flow following 7-day ischemic preconditioning in healthy humans. *Front Physiol.* 2018;9:1–12. doi: 10.3389/fphys.2018.00463
- 235 Ryan TE, Brophy P, Lin C-T, et al. Assessment of in vivo skeletal muscle mitochondrial respiratory capacity in humans by near-infrared spectroscopy: a comparison with in situ measurements. J Physiol. 2014;592:3231–41. doi: 10.1113/jphysiol.2014.274456
- 236 Kennedy MD, Haykowsky MJ, Boliek CA, *et al.* Regional muscle oxygenation differences in vastus lateralis during different modes of incremental exercise. *Dynamic Medicine*. 2006;5:1–8.
- 237 Mizuno M, Tokizawa K, Iwakawa T, *et al.* Inflection points of cardiovascular responses and oxygenation are correlated in the distal but not the proximal portions of muscle during incremental exercise. *J Appl Physiol.* 2004;97:867–73.
- 238 Lucero AA, Addae G, Lawrence W, et al. Reliability of muscle blood flow and oxygen consumption response from exercise using near-infrared spectroscopy. Exp Physiol. 2018;103:90–100. doi: 10.1113/EP086537
- 239 Ferrari M, Mottola L, Quaresima V. Principles, techniques, and limitations of near infrared spectroscopy. *Canadian journal of applied physiology*. 2004;29:463–87.
- 240 Carlier PG, Bertoldi D, Baligand C, *et al.* Muscle blood flow and oxygenation measured by NMR imaging and spectroscopy. *NMR in Biomedicine: An International Journal Devoted to the Development and Application of Magnetic Resonance In vivo.* 2006;19:954–67.
- 241 van Beekvelt MCP, van Engelen BGM, Wevers RA, et al. In vivo quantitative nearinfrared spectroscopy in skeletal muscle during incremental isometric handgrip exercise. Clin Physiol Funct Imaging. 2002;22:210–7. doi: 10.1046/j.1475-097X.2002.00420.x
- 242 Hamaoka T, Katsumura T, Murase N, *et al.* Muscle oxygen consumption at onset of exercise by near infrared spectroscopy in humans. Oxygen Transport to Tissue XXIV. Springer 2003:475–83.
- 243 Ferrari M, Muthalib M, Quaresima V. The use of near-infrared spectroscopy in understanding skeletal muscle physiology: Recent developments. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences.* 2011;369:4577–90. doi: 10.1098/rsta.2011.0230
- Ganesan G, Cotter JA, Reuland W, et al. Effect of blood flow restriction on tissue oxygenation during knee extension. Med Sci Sports Exerc. 2015;47. doi: 10.1249/MSS.00000000000393
- Biddulph B, Morris JG, Lewis M, *et al.* Reliability of Near-Infrared Spectroscopy with and without Compression Tights during Exercise and Recovery Activities. *Sports*. 2023;11. doi: 10.3390/sports11020023
- Melia M, Geissler B, König J, *et al.* Pressure pain thresholds: Subject factors and the meaning of peak pressures. *European Journal of Pain (United Kingdom)*.
 2019;23:167–82. doi: 10.1002/ejp.1298

- 247 Mailloux C, Beaulieu L-D, Wideman TH, *et al.* Within-session test-retest reliability of pressure pain threshold and mechanical temporal summation in healthy subjects. *PLoS One.* 2021;16:e0245278.
- Hollander DB, Reeves G V, Clavier JD, et al. Partial Occlusion During Resistance Exercise Alters Effort Sense and Pain. J Strength Cond Res. 2010;24:235–43. doi: 10.1519/JSC.0b013e3181c7badf
- 249 Tilley P, Bisset L. The Reliability and Validity of Using Ice to Measure Cold Pain Threshold. *Biomed Res Int*. 2017;2017:6–11. doi: 10.1155/2017/7640649
- 250 Wright A, Benson HAE, Will R, et al. Cold pain threshold identifies a subgroup of individuals with knee osteoarthritis that present with multimodality hyperalgesia and elevated pain levels. *Clinical Journal of Pain*. 2017;33:793–803. doi: 10.1097/AJP.00000000000458
- 251 Goodin BR, Bulls HW, Herbert MS, et al. Temporal summation of pain as a prospective predictor of clinical pain severity in adults aged 45 years and older with knee osteoarthritis: Ethnic differences. Psychosom Med. 2014;76:302–10. doi: 10.1097/PSY.000000000000058
- 252 Aspinall SL, Jacques A, Leboeuf-yde C, *et al.* Pressure pain threshold and temporal summation in adults with episodic and persistent low back pain trajectories : a secondary analysis at baseline and after lumbar manipulation or sham. 2020;5:1–12.
- 253 Analog V, Vas P, Rating N, et al. Measures of Adult Pain. 2011;63:240–52. doi: 10.1002/acr.20543
- 254 Petrella M, Selistre LFA, Serrão PRMS, et al. Kinetics, kinematics, and knee muscle activation during sit to stand transition in unilateral and bilateral knee osteoarthritis. Gait Posture. 2021;86:38–44. doi: 10.1016/j.gaitpost.2021.02.023
- 255 Ghahramani M, Stirling D, Naghdy F. The sit to stand to sit postural transition variability in the five time sit to stand test in older people with different fall histories. *Gait Posture*. 2020;81:191–6. doi: 10.1016/j.gaitpost.2020.07.073
- 256 Bohannon RW, Bubela DJ, Magasi SR, et al. Sit-to-stand test: Performance and determinants across the age-span. Isokinet Exerc Sci. 2010;18:235–40. doi: 10.3233/IES-2010-0389
- 257 Bohannon RW. Measuring knee extensor muscle strength. *Am J Phys Med Rehabil.* 2001;80:13–8.
- Bohannon RW, Bubela DJ, Magasi SR, et al. Sit-to-stand test: Performance and determinants across the age-span. Isokinet Exerc Sci. 2010;18:235–40. doi: 10.3233/IES-2010-0389
- 259 Jang S, Lee K, Ju JH. Recent updates of diagnosis, pathophysiology, and treatment on osteoarthritis of the knee. Int J Mol Sci. 2021;22:1–15.
- 260 Bremander AB, Dahl LL, Roos EM. Validity and reliability of functional performance tests in meniscectomized patients with or without knee osteoarthritis. *Scand J Med Sci Sports*. 2007;17:120–7. doi: 10.1111/j.1600-0838.2006.00544.x
- 261 Bremander AB, Dahl LL, Roos EM. Validity and reliability of functional performance tests in meniscectomized patients with or without knee osteoarthritis. *Scand J Med Sci Sports*. 2007;17:120–7. doi: 10.1111/j.1600-0838.2006.00544.x

- 262 Roos EM, Toksvig-Larsen S. Health and Quality of Life Outcomes Knee injury and Osteoarthritis Outcome Score (KOOS)-validation and comparison to the WOMAC in total knee replacement. 2003.
- 263 Collins NJ, Prinsen CAC, Christensen R, et al. Knee Injury and Osteoarthritis Outcome Score (KOOS): systematic review and meta-analysis of measurement properties. Osteoarthritis Cartilage. 2016;24:1317–29.
- Boone T, Board R, Astorino T, *et al.* Journal of Exercise Physiologyonline Volume
 19 Number 3 Editor-in-Chief JEPonline Effects of Electrostimulation with Blood
 Flow Restriction on Muscle Thickness and Strength of the Soleus. 2016.
- 265 Fisher JP, Blossom D, Steele J. A comparison of volume-equated knee extensions to failure, or not to failure, upon rating of perceived exertion and strength adaptations. 2015.
- 266 Kim D, Loenneke JP, Ye X, *et al.* Low-load resistance training with low relative pressure produces muscular changes similar to high-load resistance training. *Muscle Nerve.* 2017;56.
- 267 Fatela P, Reis JF, Mendonca G V., *et al.* Acute effects of exercise under different levels of blood-flow restriction on muscle activation and fatigue. *Eur J Appl Physiol.* 2016;116:985–95. doi: 10.1007/s00421-016-3359-1
- 268 Cook SB, Murphy BG, Labarbera KE. Neuromuscular function after a bout of lowload blood flow-restricted exercise. *Med Sci Sports Exerc.* 2013;45:67–74. doi: 10.1249/MSS.0b013e31826c6fa8
- 269 Pierce JR, Clark BC, Ploutz-Snyder LL, *et al.* Growth hormone and muscle function responses to skeletal muscle ischemia. *J Appl Physiol.* 2005;101:1588–95. doi: 10.1152/japplphysiol.00585.2006
- 270 Cook SB, Clark BC, Ploutz-Snyder LL. Effects of exercise load and blood-flow restriction on skeletal muscle function. *Med Sci Sports Exerc.* 2007;39:1708–13. doi: 10.1249/mss.0b013e31812383d6
- Karabulut M, Mccarron J, Abe T, *et al.* The effects of different initial restrictive pressures used to reduce blood flow and thigh composition on tissue oxygenation of the quadriceps. *J Sports Sci.* 2011;29:951–8. doi: 10.1080/02640414.2011.572992
- 272 Cook SB, Kanaley JA, Ploutz-Snyder LL. Neuromuscular function following muscular unloading and blood flow restricted exercise. *Eur J Appl Physiol.* 2014;114:1357–65. doi: 10.1007/s00421-014-2864-3
- 273 Suga T, Okita K, Morita N, *et al.* Intramuscular metabolism during low-intensity resistance exercise with blood flow restriction. 2009;1119–24. doi: 10.1152/japplphysiol.90368.2008.
- 274 Sugaya M, Yasuda T, Suga T, *et al.* Change in intramuscular inorganic phosphate during multiple sets of blood flow-restricted low-intensity exercise. *Clin Physiol Funct Imaging.* 2011;31:411–3.
- 275 Suga T, Okita K, Takada S, *et al.* Effect of multiple set on intramuscular metabolic stress during low-intensity resistance exercise with blood flow restriction. *Eur J Appl Physiol.* 2012;112:3915–20. doi: 10.1007/s00421-012-2377-x
- 276 Amann M, Sidhu SK, Weavil JC, *et al.* Autonomic responses to exercise: group III/IV muscle afferents and fatigue. *Autonomic Neuroscience*. 2015;188:19–23.

- 277 Amann M, Calbet JAL. Convective oxygen transport and fatigue. *J Appl Physiol.* 2008;104:861–70.
- 278 Krustrup P, Söderlund K, Relu MU, et al. Heterogeneous recruitment of quadriceps muscle portions and fibre types during moderate intensity knee-extensor exercise: effect of thigh occlusion. Scand J Med Sci Sports. 2009;19:576–84.
- 279 Allen DG, Lamb GD, Westerblad H. Skeletal muscle fatigue: cellular mechanisms. *Physiol Rev.* 2008;88:287–332.
- 280 Wernbom M, Järrebring R, Andreasson M a, et al. Acute effects of blood flow restriction on muscle activity and endurance during fatiguing dynamic knee extensions at low load. Journal of strength and conditioning research / National Strength & Conditioning Association. 2009;23:2389–95. doi: 10.1519/JSC.0b013e3181bc1c2a
- 281 Loenneke JP, Kim D, Fahs CA, et al. The influence of exercise load with and without different levels of blood flow restriction on acute changes in muscle thickness and lactate. Clin Physiol Funct Imaging. 2017;37:734–40. doi: 10.1111/cpf.12367
- 282 Counts BR, Dankel SJ, Barnett BE, *et al.* Influence of relative blood flow restriction pressure on muscle activation and muscle adaptation. *Muscle Nerve.* 2016;53:438– 45. doi: 10.1002/mus.24756
- 283 Yasuda T, Loenneke JP, Thiebaud RS, et al. Effects of Blood Flow Restricted Low-Intensity Concentric or Eccentric Training on Muscle Size and Strength. PLoS One. 2012;7:1–7. doi: 10.1371/journal.pone.0052843
- 284 Schoenfeld BJ. Potential mechanisms for a role of metabolic stress in hypertrophic adaptations to resistance training. *Sports medicine*. 2013;43:179–94.
- 285 Jones MD, Taylor JL, Barry BK. Occlusion of blood flow attenuates exerciseinduced hypoalgesia in the occluded limb of healthy adults. *J Appl Physiol.* 2017;122:1284–91.
- 286 Teixeira EL, Barroso R, Silva-Batista C, *et al.* Blood flow restriction increases metabolic stress but decreases muscle activation during high-load resistance exercise. *Muscle Nerve.* 2018;57:107–11. doi: 10.1002/mus.25616
- 287 Wernbom M, Augustsson J, Thomeé R. Effects of Vascular Occlusion on Muscular Endurance in Dynamic Knee Extension Exercise At Different Submaximal Loads. J Strength Cond Res. 2006;20:372–7. doi: 10.1519/R-16884.1
- 288 Kang JH, Hyong IH. The influence of neuromuscular electrical stimulation on the heart rate variability in healthy subjects. *J Phys Ther Sci.* 2014;26:633–5.
- 289 Lee SY, Im SH, Kim BR, et al. The effects of neuromuscular electrical stimulation on cardiopulmonary function in healthy adults. Ann Rehabil Med. 2012;36:849–56.
- 290 Iida H, Kurano M, Takano H, et al. Hemodynamic and neurohumoral responses to the restriction of femoral blood flow by KAATSU in healthy subjects. Eur J Appl Physiol. 2007;100:275–85.
- 291 Loenneke JP, Fahs CA, Thiebaud RS, et al. The acute hemodynamic effects of blood flow restriction in the absence of exercise. Clin Physiol Funct Imaging. 2013;33:79–82. doi: 10.1111/j.1475-097X.2012.01157.x

- 292 Fisher J, Steele J, Smith D. High-and low-load resistance training: interpretation and practical application of current research findings. *Sports Medicine*. 2017;47:393–400.
- 293 Abdalla LHP, Denadai BS, Bassan NM, *et al.* Exercise tolerance during muscle contractions below and above the critical torque in different muscle groups. *Applied Physiology, Nutrition, and Metabolism.* 2018;43:174–9.
- 294 Vanhatalo A, Black MI, DiMenna FJ, et al. The mechanistic bases of the powertime relationship: muscle metabolic responses and relationships to muscle fibre type. Journal of Physiology. 2016;594:4407–23. doi: 10.1113/JP271879
- 295 Byrd MT, Switalla JR, Eastman JE, et al. Contributions of body-composition characteristics to critical power and anaerobic work capacity. Int J Sports Physiol Perform. 2018;13:189–93.
- 296 Dekerle J, de Souza KM, de Lucas RD, *et al.* Exercise tolerance can be enhanced through a change in work rate within the severe intensity domain: work above critical power is not constant. *PLoS One*. 2015;10:e0138428.
- 297 Ringdal GI, Ringdal K, Juliebø V, *et al.* Using the Mini-Mental State Examination to screen for delirium in elderly patients with hip fracture. *Dement Geriatr Cogn Disord.* 2011;32:394–400.
- 298 Mitchell EA, Martin NRW, Bailey SJ, *et al.* Critical power is positively related to skeletal muscle capillarity and type i muscle fibers in endurance-trained individuals. *J Appl Physiol.* 2018;125:737–45. doi: 10.1152/japplphysiol.01126.2017
- 299 Vieira A, Gadelha AB, Ferreira-Junior JB, *et al.* Session rating of perceived exertion following resistance exercise with blood flow restriction. *Clin Physiol Funct Imaging*. 2015;35:323–7. doi: 10.1111/cpf.12128
- 300 Neto GR, Sousa MSC, Costa e Silva G V., *et al.* Acute resistance exercise with blood flow restriction effects on heart rate, double product, oxygen saturation and perceived exertion. *Clin Physiol Funct Imaging.* 2016;36:53–9. doi: 10.1111/cpf.12193
- 301 Loenneke JP, Balapur A, Thrower AD, *et al.* The perceptual responses to occluded exercise. *Int J Sports Med.* 2011;32:181–4. doi: 10.1055/s-0030-1268472
- 302 Hughes L, Grant I, Patterson SD. Aerobic exercise with blood flow restriction causes local and systemic hypoalgesia and increases circulating opioid and endocannabinoid levels. *J Appl Physiol.* 2021;131:1460–8. doi: 10.1152/japplphysiol.00543.2021
- 303 van Beekvelt MCP, Colier WNJM, Wevers RA, *et al.* Performance of near-infrared spectroscopy in measuring local O2 consumption and blood flow in skeletal muscle. *J Appl Physiol.* 2001;90:511–9.
- 304 Cayot TE, Lauver JD, Silette CR, *et al.* Effects of blood flow restriction duration on muscle activation and microvascular oxygenation during low-volume isometric exercise. *Clin Physiol Funct Imaging.* 2016;36:298–305.
- 305 Lauver JD, Cayot TE, Rotarius T, *et al.* The effect of eccentric exercise with blood flow restriction on neuromuscular activation, microvascular oxygenation, and the repeated bout effect. *Eur J Appl Physiol.* 2017;117:1005–15.

- 306 Takarada Y, Tsuruta T, Ishii N. Cooperative effects of exercise and occlusive stimuli on muscular function in low-intensity resistance exercise with moderate vascular occlusion. Jpn J Physiol. 2004;54:585–92. doi: 10.2170/jjphysiol.54.585
- 307 Bagley EE, Ingram SL. Endogenous opioid peptides in the descending pain modulatory circuit. Neuropharmacology. 2020;173.
- 308 Hirayama A, Saitoh Y, Kishima H, et al. Reduction of intractable deafferentation pain by navigation-guided repetitive transcranial magnetic stimulation of the primary motor cortex. Pain. 2006;122:22–7. doi: 10.1016/j.pain.2005.12.001
- 309 Enoka RM, Duchateau J. Muscle fatigue: what, why and how it influences muscle function. *J Physiol.* 2008;586:11–23.
- 310 Enoka RM, Duchateau J. Translating fatigue to human performance. *Med Sci Sports Exerc*. 2016;48:2228–38. doi: 10.1249/MSS.00000000000929
- 311 Boyas S, Guével A. Neuromuscular fatigue in healthy muscle: Underlying factors and adaptation mechanisms. Ann Phys Rehabil Med. 2011;54:88–108. doi: 10.1016/j.rehab.2011.01.001
- 312 Loenneke JP, Dankel SJ, Bell ZW, et al. Is muscle growth a mechanism for increasing strength? *Med Hypotheses*. 2019;125:51–6. doi: 10.1016/j.mehy.2019.02.030
- 313 Edwards MH, Gregson CL, Patel HP, *et al.* Muscle size, strength, and physical performance and their associations with bone structure in the Hertfordshire Cohort Study. *Journal of Bone and Mineral Research.* 2013;28:2295–304.
- 314 Visser M, Goodpaster BH, Kritchevsky SB, et al. Muscle Mass, Muscle Strength, and Muscle Fat Infiltration as Predictors of Incident Mobility Limitations in Well-Functioning Older Persons. 2005.
- 315 Salimin N. Muscle Architecture and Exercise Performance: A Mini Review. *Biomed J Sci Tech Res.* 2018;3. doi: 10.26717/bjstr.2018.03.000958
- 316 Hughes L, Rosenblatt B, Haddad F, et al. Comparing the Effectiveness of Blood Flow Restriction and Traditional Heavy Load Resistance Training in the Post-Surgery Rehabilitation of Anterior Cruciate Ligament Reconstruction Patients: A UK National Health Service Randomised Controlled Trial. *Sports Medicine*. 2019;49:1787–805. doi: 10.1007/s40279-019-01137-2
- 317 Sharifnezhad A, Marzilger R, Arampatzis A. Effects of load magnitude, muscle length and velocity during eccentric chronic loading on the longitudinal growth of the vastus lateralis muscle. *Journal of Experimental Biology*. 2014;217:2726–33. doi: 10.1242/jeb.100370
- 318 Bishop D, Jenkins DG. The influence of resistance training on the critical power function & time to fatigue at critical power. *Aust J Sci Med Sport*. 1996;28:101–5.
- 319 Sawyer BJ, Stokes DG, Womack CJ, *et al.* Strength training increases endurance time to exhaustion during high-intensity exercise despite no change in critical power. *The Journal of Strength & Conditioning Research*. 2014;28:601–9.
- 320 Muthalib M, Kerr G, Nosaka K, et al. Local muscle metabolic demand induced by neuromuscular electrical stimulation and voluntary contractions at different force levels: a NIRS study. Eur J Transl Myol. 2016;26:169–74. doi: 10.4081/ejtm.2016.6058

- 321 Tuesta M, Yáñez-Sepúlveda R, Verdugo-Marchese H, et al. Near-Infrared Spectroscopy Used to Assess Physiological Muscle Adaptations in Exercise Clinical Trials: A Systematic Review. Biology (Basel). 2022;11.
- 322 Hultman E, Greenhaff PL, Ren JM, et al. Energy metabolism and fatigue during intense muscle contraction. *Biochem Soc Trans.* 1991;19:347–53. doi: 10.1042/bst0190347
- 323 Russ DW, Elliott MA, Vandenborne K, *et al.* Metabolic costs of isometric force generation and maintenance of human skeletal muscle. *Am J Physiol Endocrinol Metab.* 2002;282. doi: 10.1152/ajpendo.00285.2001
- 324 Hultman E, Sjöholm H. Energy metabolism and contraction force of human skeletal muscle in situ during electrical stimulation. *J Physiol*. 1983;345:525–32.
- 325 Sillen MJH, Franssen FME, Gosker HR, *et al.* Metabolic and Structural Changes in Lower-Limb Skeletal Muscle Following Neuromuscular Electrical Stimulation: A Systematic Review. *PLoS One*. 2013;8. doi: 10.1371/journal.pone.0069391
- 326 Jubeau M, Zory R, Gondin J, *et al.* Effect of electrostimulation training–detraining on neuromuscular fatigue mechanisms. *Neurosci Lett.* 2007;424:41–6.
- 327 Raja SN, Carr DB, Cohen M, *et al.* The revised International Association for the Study of Pain definition of pain: concepts, challenges, and compromises. Pain. 2020;161:1976–82.
- 328 MacDonald DI, Wood JN, Emery EC. Molecular mechanisms of cold pain. Neurobiology of Pain. 2020;7.
- 329 Arntz A, Claassens L. The meaning of pain influences its experienced intensity. *Pain*. 2004;109:20–5.
- 330 Yam MF, Loh YC, Tan CS, *et al.* General pathways of pain sensation and the major neurotransmitters involved in pain regulation. Int J Mol Sci. 2018;19.
- 331 Karanasios S, Lignos I, Kouvaras K, *et al.* Low-Intensity Blood Flow Restriction Exercises Modulate Pain Sensitivity in Healthy Adults: A Systematic Review. Healthcare (Switzerland). 2023;11.
- 332 Koltyn KF, Brellenthin AG, Cook DB, *et al.* Mechanisms of exercise-induced hypoalgesia. *J Pain.* 2014;15:1294–304.
- 333 Papavasileiou A, Xenofondos A, Baudry S, et al. Protocols Targeting Afferent Pathways via Neuromuscular Electrical Stimulation for the Plantar Flexors: A Systematic Review. Sensors. 2023;23.
- 334 Cheng JC, Erpelding N, Kucyi A, *et al.* Individual differences in temporal summation of pain reflect pronociceptive and antinociceptive brain structure and function. *Journal of Neuroscience.* 2015;35:9689–700. doi: 10.1523/JNEUROSCI.5039-14.2015
- 335 Norbury R, Dickens L, Grant I, et al. Remote ischaemic preconditioning increase tolerance to experimentally induced cold pain. Eur J Sport Sci. 2023;23:2435–42. doi: 10.1080/17461391.2023.2241831
- 336 Hughes XL, Stephen X, Patterson D. The effect of blood flow restriction exercise on exercise-induced hypoalgesia and endogenous opioid and endocannabinoid mechanisms of pain modulation. *J Appl Physiol*. 2020;128:914–24. doi: 10.1152/japplphysiol.00768.2019.-This

- 337 Rudolph KS, Cloutier M, Stackhouse S. Pain inhibition—the unintended benefit of electrically elicited muscle strengthening contractions. *BMC Musculoskelet Disord*. 2023;24. doi: 10.1186/s12891-023-06243-x
- 338 Fisher JP, Farrow J, Steele J. Acute fatigue, and perceptual responses to resistance exercise. *Muscle Nerve*. 2017;56:E141–6. doi: 10.1002/mus.25645
- Bement MKH, Dicapo J, Rasiarmos R, *et al.* Dose response of isometric contractions on pain perception in healthy adults. *Med Sci Sports Exerc.* 2008;40:1880–9. doi: 10.1249/MSS.0b013e31817eeecc
- 340 Misra G, Paris TA, Archer DB, et al. Dose-response effect of isometric force production on the perception of pain. PLoS One. 2014;9. doi: 10.1371/journal.pone.0088105
- 341 Goodin BR, Bulls HW, Herbert MS, et al. Temporal summation of pain as a prospective predictor of clinical pain severity in adults aged 45 years and older with knee osteoarthritis: Ethnic differences. Psychosom Med. 2014;76:302–10. doi: 10.1097/PSY.000000000000058
- 342 Slysz JT, Burr JF. Ischemic Preconditioning: Modulating Pain Sensitivity and Exercise Performance. *Front Physiol.* 2021;12. doi: 10.3389/fphys.2021.696488
- 343 Wright A, Benson HAE, Will R, et al. Cold pain threshold identifies a subgroup of individuals with knee osteoarthritis that present with multimodality hyperalgesia and elevated pain levels. Clinical Journal of Pain. 2017;33:793–803. doi: 10.1097/AJP.00000000000458
- 344 Ploutz-Snyder LL, Manini T, Ploutz-Snyder RJ, et al. Functionally relevant thresholds of quadriceps femoris strength. J Gerontol A Biol Sci Med Sci. 2002;57:B144–52.
- 345 Zeng C, li H, Yang T, *et al.* Electrical stimulation for pain relief in knee osteoarthritis: Systematic review and network meta-analysis. *Osteoarthritis Cartilage*. 2015;23:189–202. doi: 10.1016/j.joca.2014.11.014
- 346 Grantham B, Korakakis V, O'Sullivan K. Does blood flow restriction training enhance clinical outcomes in knee osteoarthritis: A systematic review and metaanalysis. Physical Therapy in Sport. 2021;49:37–49.
- 347 De Oliveira Melo M, Aragão FA, Vaz MA. Neuromuscular electrical stimulation for muscle strengthening in elderly with knee osteoarthritis - A systematic review. Complement Ther Clin Pract. 2013;19:27–31.
- 348 Tomschi F, Lieverkus D, Hilberg T. Exercise-induced hypoalgesia (EIH) in response to different exercise intensities. *Eur J Appl Physiol.* 2022;122:2213–22. doi: 10.1007/s00421-022-04997-1
- Boerio D, Jubeau M, Zory R, *et al.* Central and peripheral fatigue after electrostimulation-induced resistance exercise. *Med Sci Sports Exerc.* 2005;37:973–8. doi: 10.1249/01.mss.0000166579.81052.9c
- 350 Osteoarthritis in over 16s: diagnosis and management NICE guideline. 2022.
- 351 Hassan BS, Doherty SA, Mockett S, *et al.* Effect of pain reduction on postural sway, proprioception, and quadriceps strength in subjects with knee osteoarthritis. *Ann Rheum Dis.* 2002;61:422–8. doi: 10.1136/ard.61.5.422

- 352 Evans C, Vance S, Brown M. Short-term resistance training with blood flow restriction enhances microvascular filtration capacity of human calf muscles. *J Sports Sci.* 2010;28:999–1007. doi: 10.1080/02640414.2010.485647
- 353 Nielsen JL, Frandsen U, Jensen KY, et al. Skeletal Muscle Microvascular Changes in Response to Short-Term Blood Flow Restricted Training—Exercise-Induced Adaptations and Signs of Perivascular Stress. Front Physiol. 2020;11. doi: 10.3389/fphys.2020.00556
- 354 Rosenberry R, Nelson MD. REACTIVE HYPEREMIA: A REVIEW OF METHODS, MECHANISMS, AND CONSIDERATIONS 1 2.
- 355 Goodin BR, Bulls HW, Herbert MS, *et al.* Temporal summation of pain as a prospective predictor of clinical pain severity in adults aged 45 years and older with knee osteoarthritis: Ethnic differences. *Psychosom Med.* 2014;76:302–10. doi: 10.1097/PSY.000000000000058
- 356 Graven-Nielsen T, Arendt-Nielsen L. Induction and assessment of muscle pain, referred pain, and muscular hyperalgesia. *Curr Pain Headache Rep.* 2003;7:443– 51.
- 357 Slysz JT, Burr JF. Ischemic Preconditioning: Modulating Pain Sensitivity and Exercise Performance. *Front Physiol.* 2021;12. doi: 10.3389/fphys.2021.696488
- 358 Melia M, Geissler B, König J, *et al.* Pressure pain thresholds: Subject factors and the meaning of peak pressures. *European Journal of Pain (United Kingdom)*. 2019;23:167–82. doi: 10.1002/ejp.1298
- 359 Wright A, Benson HAE, Will R, et al. Cold pain threshold identifies a subgroup of individuals with knee osteoarthritis that present with multimodality hyperalgesia and elevated pain levels. Clinical Journal of Pain. 2017;33:793–803. doi: 10.1097/AJP.00000000000458
- 360 Bickel CS, Yarar-Fisher C, Mahoney ET, et al. Neuromuscular Electrical Stimulation–Induced Resistance Training After SCI: A Review of the Dudley Protocol. Top Spinal Cord Inj Rehabil. 2015;21:294–302. doi: 10.1310/sci2104-294
- 361 Freitas de M. Role of metabolic stress for enhancing muscle adaptations: Practical applications 55 Targeted temperature management in neurological intensive care unit Basic Study 68 Nutech functional score: A novel scoring system to assess spinal cord injury patients. World Journal of Methodology World J Methodol. 2017;7:33–72.
- 362 Yoshikawa M, Morifuji T, Matsumoto T, *et al.* Effects of combined treatment with blood flow restriction and low-current electrical stimulation on muscle hypertrophy in rats. *J Appl Physiol.* 2019;127:1288–96. doi: 10.1152/japplphysiol.00070.2019
- 363 Toshiharu Natsume HORKHK& HN. Effects of training intensity in electromyostimulation on human skeletal muscle. *European Journal of Applied Physiology* . 2018;118:1339–47.



25 May 2017

SMEC_2016-17_104

Paul Head (SHAS): 'The acute effect of neuromuscular electrical stimulation combined with varying degrees of blood flow restriction on physiological and haemodynamic variables'.

Dear Paul

University Ethics Sub-Committee

Thank you for submitting your ethics application for consideration.

I can confirm that your application has been considered by the Ethics Sub-Committee and that ethical approval is granted.

Yours sincerely

lowor Presso

Prof Conor Gissane Chair of the Ethics Sub-Committee

Cc Stephen Patterson, Mark Waldron, Nicola Theis



11 September 2018

SMEC_2017-18_143

Paul Head (SHAS): 'Effect of neuromuscular electrical stimulation combined with varying degrees of blood flow restriction on muscular and cardiovascular function'.

Dear Paul

University Ethics Sub-Committee

Thank you for re-submitting your ethics application for consideration.

I can confirm that all required amendments have been made and that you therefore have ethical approval to undertake your research.

Yours sincerely



Approval Sheet

Name of proposer(s)	Paul Head
Name of supervisor	Dr Stephen Patterson, Dr Mark Waldron & Dr Nicola Theis
Programme of study	PhD
Title of project	Effect of neuromuscular electrical stimulation combined with varying degrees of blood flow restriction on pressure pain

Supervisors, please complete section 1. If approved at level 1, please forward a copy of this Approval Sheet to the School Ethics Representative for their records.

 SECTION 1: To be completed by supervisor.

 Approved at Level 1.

 Refer to School Ethics Representative for consideration at Level 2 or Level 3.

 Signature of Supervisor (for student research projects):

SECTION 2: To be completed by School Ethics Representative



10 November 2022

SMU_ETHICS_2021-22-329

Paul Head (SAHPS): 'Effect of neuromuscular electrical stimulation combined with varying degrees of blood flow restriction on pressure and cold pain with knee pathology'

Dear Paul,

University Ethics Sub-Committee

Thank you for re-submitting your ethics application for consideration.

I can confirm that all required amendments have been made and that you therefore have ethical approval to undertake your research.

Yours sincerely

Rating	Perceived Exertion		
6	No exertion		
7	Extremely light		
8			
9	Very light		
10			
11	Light		
12			
13	Somewhat hard		
14			
15	Hard		
16			
17	Very hard		
18			
19	Extremely hard		
20	Maximal exertion		

Appendix 5.2 Numeric pain rating scale



0-10 NUMERIC PAIN RATING SCALE

Appendix 5.3 Physical activity and readiness questionnaire

CONFIDENTIAL MEDICAL HISTORY / PHYSICAL ACTIVITY READINESS QUESTIONNAIRE (PAR-Q) FORM

This screening form must be used in conjunction with an agreed Consent Form.

Full Name: Height (cm):				Date of Bi Weight (k	rth:		
Have you ever su	ffered from any	of the followi Yes No	ng medica Detai	al conditions? I	f yes pl	lease give details:	
Heart Disease or	attack						
High or low blood	pressure						
Stroke							
Cancer							
Diabetes							
Astrima High cholecterol		8 8					
Follensy		8 8					
Allergies							
Other, please give	e details	ō ō					
Do you suffer from	m any blood bor	ne diseases?	lf ye	s please give (details;		
-			-				
Please give detai	Is of any medica	ation you are	currently	taking or have	taken	regularly within the last y	year:
Please give detai capacity to exerci	Is of any muscu ise or caused yo	loskeletal in u to take tim	juries yo e off work	u have had in or seek medic	the pas al advid	t 6 months which have ce:	affected your
Other Important During a typical w	Information week approximat	ely how man	y hours w	ould you spen	d exerc	ising?	
If you smoke plea	ase indicate how	v many per d	ay:				_
If you drink alcoh	ol please indica	te how many	units ner	week			_
ii you ullik alcon	ior please indica	te now many	unite per	WOOK.			
Are you currently	taking any supp	plements or	medicatio	on? Please giv	/e detai	ls:	
Is there any rea	son not promp	ted above t	hat would	l prevent you	from p	articipating within the	relevant activity?
By signing th circumstance	is document as that would	l agree to in prevent me	nform the from pa	e relevant in articipating i	dividu n spec	al(s) of any change(s ific activities.	s) to my
Signature (Pa	articipant):			Da	te:		
Signature (Te	est Coordinate	or*):		Da	te:		

Appendix 5.4 Wellness questionnaire

	5	4	3	2	1	Record Score
FATIGUE	Very fresh	Fresh	Normal	More tired than normal	Always tired	
SLEEP QUALITY	Very restful	Good	Difficulty falling asleep	Restless sleep	Insomnia	
GENERAL MUSCLE SORENESS	Feeling great	Feeling good	Normal	Increase in soreness/tightness	Very sore	
STRESS LEVELS	Very relaxed	Relaxed	Normal	Feeling stressed	Highly stressed	
MOOD	Very positive mood	A generally good mood	Less interested in others &/or activities than usual	Snappiness at team- mates, family and co-workers	Highly annoyed/ irritable/down	

Appendix 5.4 Knee injury and osteoarthritis outcome score questionnaire

KOOS KNEE SURVEY

Today's date: ____/___ Date of birth: ____/___

Name:

INSTRUCTIONS: This survey asks for your view about your knee. This information will help us keep track of how you feel about your knee and how well you are able to perform your usual activities.

Answer every question by ticking the appropriate box, only <u>one</u> box for each question. If you are unsure about how to answer a question, please give the best answer you can.

Symptoms

These questions should be answered thinking of your knee symptoms during the **last week**.

S1. Do you have	e swelling in you	r knee?		
Never	Rarely	Sometimes	Often	Always

S2. Do you feel grinding, hear clicking or any other type of noise when your knee moves?

Never	Rarely	Sometimes	Often	Always
S3. Does your k	nee catch or han	g up when moving?	,	
Never	Rarely	Sometimes	Often	Always
S4. Can you stra	ughten your knee	e fully?		
Always	Often	Sometimes	Rarely	Never
S5. Can you ber	nd your knee full	y?		
Always	Often	Sometimes	Rarely	Never

Stiffness

The following questions concern the amount of joint stiffness you have experienced during the **last week** in your knee. Stiffness is a sensation of restriction or slowness in the ease with which you move your knee joint.

S6. How severe is your knee joint stiffness after first wakening in the morning? None Mild Moderate Severe Extreme S7. How severe is your knee stiffness after sitting, lying or resting later in the day? None Mild Moderate Severe Extreme